No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 4th Revised Sheet 1 Replacing 3rd Revised Sheet 1

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE

The Telephone Company will provide Carrier Common Line Access Service (Carrier Common Line Access) to ICs in conjunction with line side and trunk side Switched Access Service.

3.1 General Description

Carrier Common Line Access provides for the use of Telephone Company common lines by ICs for access to End Users to furnish IC intrastate telecommunications service.

Carrier Common Line Access is provided where the IC obtains Telephone Company line side or trunk side Switched Access Service under this Tariff.

Premium Access is Switched Access Service other than the DNAL BSA provided to IC's under this Tariff which furnish intrastate MTS/WATS.

Nonpremium Access is Switched Access Service other than the DNAL BSA provided to IC's under this Tariff that do not furnish intrastate MTS/WATS. FGA, FGB, BSA-A and BSA-B are not offered for use in terminating FGC or BSA-C originating communications.

3.2 Limitations

- A. A telephone number is not provided with Carrier Common Line Access.
- B. Detail billing is not provided for Carrier Common Line Access.
- C. Directory listings are not included in the rates and charges for Carrier Common Line Access.
- D. Intercept arrangements are not included in the rates and charges for Carrier Common Line Access.
- E. All trunk side connections provided in the same combined access group or BSA will be limited to the same features and operating characteristics.
- F. All line side connections provided in the same combined access group or BSA will be limited to the same features and operating characteristics.
- G. WATS Access Line Service minutes switched at Telephone Company WATS Serving Offices (i.e., originating minutes for outward WATS and WATS-type services and terminating minutes for inward WATS and WATS-type services) shall not be assessed Carrier Common Line Access per minute charges.
- H. Minutes of use associated with the closed end of WATS/800 service are not subject to Carrier Common Line charges. However, when common lines are utilized for the termination of 800 NPAS, Carrier Common Line charges are_applicable.



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff

Section 3

3rd REGEIVED 1 Replacing 2nd Revised Sheet 1

ACCESS SERVICES

MAR 291993

MISSOURI

CARRIER COMMON LINE ACCESS SERVICE

The Telephone Company will provide Carrier Common Line Rublic Service Commission (Carrier Common Line Access) to ICs in conjunction with line side and trunk (AT) side Switched Access Service. (AT)

3.1 General Description

Carrier Common Line Access provides for the use of Telephone Company common lines by ICs for access to End Users to furnish IC intrastate telecommunications service.

Carrier Common Line Access is provided where the IC obtains Telephone Company line side or trunk side Switched Access Service under this Tariff.

- Premium Access is Switched Access Service other than the DNAL BSA provided (AT) to IC's under this Tariff which furnish intrastate MTS/WATS.
- Nonpremium Access is Switched Access Service other than the DNAL BSA (AT) provided to IC's under this Tariff that do not furnish intrastate MTS/WATS. FGA, FGB, BSA-A and BSA-B are not offered for use in terminating (AT) (AT) FGC or BSA-C originating communications. CANCELLED

(AT)

- 3.2 Limitations
- A. A telephone number is not provided with Carrier Common Line Accessive 1 1993 \pm
- B. Detail billing is not provided for Carrier Common Line Access. BY $\frac{\mu R.s.}{\mu R.s.}$ Public Service Commissic
- Directory listings are not included in the rates and charges for GalSGOURI C. Common Line Access.
- D. Intercept arrangements are not included in the rates and charges for Carrier Common Line Access.
- E. All trunk side connections provided in the same combined access group or (AT) (AT) BSA will be limited to the same features and operating characteristics.
- F. All line side connections provided in the same combined access group or BSA (AT) will be limited to the same features and operating characteristics.
 - G. WATS Access Line Service minutes switched at Telephone Company WATS Serving Offices (i.e., originating minutes for outward WATS and WATS-type services and terminating minutes for inward WATS and WATS-type services) shall not be assessed Carrier Common Line Access per minute charges.
 - H. Minutes of use associated with the closed end of WATS/800 service are not subject to Carrier Common Line charges. However, when common lines are utilized for the termination of 800 Access Service, Carrier Common Line charges are applicable PH PN

	Junges ul		
Issued:	MAR 2 6	Effective: 199By R.D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri	92 - 304
			O. PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 2nd Revised Sheet 1 Replacing 1st Revised Sheet 1 RECEIVED

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE

OCT 1 3 1987

Public Service Commission

APR 11 1993

The Telephone Company will provide Carrier Common Line Access Service(Carrier Common Line Access) to ICs.MISSOURI

3.1 General Description

Carrier Common Line Access provides for the use of Telephone Company common lines by ICs for access to End Users to furnish IC intrastate telecommunications service.

Carrier Common Line Access is provided where the IC obtains Telephone Company Switched Access Service under this Tariff.

Premium Access is Switched Access Service provided to IC's under this Tariff which furnish intrastate MTS/WATS.

Nonpremium Access is Switched Access Service provided to IC's under this Tariff that do not furnish intrastate MTS/WATS. Feature Group A and B access are not offered for use in terminating Feature Group C originating communications.

- 3.2 Limitations
 - A. A telephone number is not provided with Carrier Common Line Access R
 - B. Detail billing is not provided for Carrier Common Line Public Service Commission
 - C. Directory listings are not included in the rates and charges for MISSOURI Carrier Common Line Access.
 - D. Intercept arrangements are not included in the rates and charges for Carrier Common Line Access.
 - E. All trunk side connections provided in the same combined access group will be limited to the same features and operating characteristics.
 - F. All line side connections provided in the same combined access group will be limited to the same features and operating characteristics.
 - G. WATS Access Line Service minutes switched at Telephone Company WATS Serving Offices (i.e., originating minutes for outward WATS and WATStype services and terminating minutes for inward WATS and WATS-type services) shall not be assessed Carrier Common Line Access per minute charges.
- (AT) H. Minutes of use associated with the closed end of WATS/800 service are not subject to Carrier Common Line charges. However, when common lines are utilized for the termination of 800 Access Service, Carrier (AT)
 (AT) Common Line charges are applicable.

Issued: OCT 1 4 1987 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

ï

ļ

1 ļ i

	tari exce	supplement to this ff will be issued opt for the purpose canceling this tariff.			cess Services Tariff Section 3 lst Revised Sheet 1 ing-Original_Sheet_1
-		ACC	ESS SERVICES		KEGEIVED
	3.	CARRIER COMMON LINE ACCESS SERV	ICE	1. 	JUN 27 1980
		The Telephone Company will prov (Carrier Common Line Access) to		1	ccess Service MISSUJR!
	3.1	General Description		1700	lic Service Commission ;
		Carrier Common Line Access pr common lines by ICs for acces telecommunications service.			
		Carrier Common Line Access is Company Switched Access Servi			tains Telephone
(CT)		Premium Access is Switched Ac Tariff which furnish intrasta		ided to	IC's under this
(CT) (AT)		Nonpremium Access is Switched this Tariff that do not furni and B access are not offered originating communications.	sh intrastate MTS for use in termin	/WATS. ating Fe	Feature Group
	3.2	2 Limitations			nit LAR Strission
	Α.	originating communications. 2 Limitations . A telephone number is not pro . Detail billing is not provide	ovided with Carrie	r Common	Line Access
	В.	. Detail billing is not provide	ed for Carrier Com	mon Line	Accessublic Miss
	C.	originating communications. 2 Limitations 3 A telephone number is not pro 4 Detail billing is not provide 5 Directory listings are not in 6 Carrier Common Line Access.	ncluded in the rat	es and c	harges for
	D.	. Intercept arrangements are no Carrier Common Line Access.	ot included in the	rates a	nd charges for
(CT)	Ε.	. All trunk side connections pr will be limited to the same f			
(AT)	F.	. All line side connections pro will be limited to the same f			
	G.	 All line side connections probe of the same type (i.e., eibe FX/ONAL-type). 	ovided to a custom ither all must be	er in a MTS/WATS	hunt group must -type or all must
	·		-	•	JUL 1 1986
	Issi	^{ued:} JUN 2 7 1986	Effective:	JUL	8 6 - 8 4 A: 1986: envice Commission
		By R. D. BARRON, Press Southwestern Bell			ay " A namet - Tankar a Annakar - Annakar a Annakar - An

No supplement to this tariff will be issued except for the purpose of canceling this tariff.



ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE

The Telephone Company will provide Carrier Common Line Access Service (Carrier Common Line Access) to IC's.

3.1 General Description

Carrier Common Line Access provides for the use of Telephone Company common lines by IC's for access to End Users to furnish IC intrastate telecommunications service.

Carrier Common Line Access is provided where the IC obtains Telephone Company Switched Access Service under this Tariff.

Premium Access is Switched Access Service provided to intrastate carriers under this Tariff which furnish intrastate MTS/WATS.

Nonpremium Access is Switched Access Service provided to intrastate carriers under this Tariff that do not furnish intrastate MTS/WATS.

3.2 Limitations

- A. A telephone number is not provided with Carrier Common Line Access.
- B. Detail billing is not provided for Carrier Common Line Access.
- C. Directory listings are not included in the rates and charges for Carrier Common Line Access.
- D. Intercept arrangements are not included in the rates and charges for Carrier Common Line Access.
- E. Switched Access Service access minutes which originate from or terminate to a Dedicated Access Line (DAL) Service will be excluded from the Carrier Common Line Access Line access minutes.
- 3.3 Undertaking of the Telephone Company
- A. Where the IC is provided with Switched Access Service under other sections of this Tartif. When relephone Company will provide the use of Telephone Company common lines by an IC for access to End Users at rates and charges as specified in Paragraph 3.8, following.

	BY LALA S. #		U/11 - 1 1934 8 3 - 2 5 3
•	PUBLIC SERVICE COMMISSION		Public Selling Commission
	· · · · · · · · · · · · · · · · · · ·	=====	

Issued: **DEC** 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 3rd Revised Sheet 2 Replacing 2nd Revised Sheet 2

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- (AT) 3.2 Limitations-(Continued)
- (AT) I. Carrier Common Line Service is not available with the DNAL BSE.
 - 3.3 Undertaking of the Telephone Company
 - A. Where the IC is provided with line side or trunk side Switched Access Service under other sections of this Tariff, the Telephone Company will provide the use of Telephone Company common lines by an IC for access to End Users at rates and charges as specified in Paragraph 3.8, following.
 - B. The Switched Access Service provided by the Telephone Company includes the Switched Access Service provided for both interstate and intrastate communications and the Carrier Common Line Access rates and charges as set forth in Paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, following.
 - C. When access to the local exchange is required to provide an MTS/WATS-type service using a resold Private Line Service, Switched Access Service Rates and Regulations, as set forth in Section 6, following, will apply. Carrier Common Line Access rates and charges as set forth in paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, F., following.
 - D. When the IC is provided Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access as set forth in Section 6, following, the Telephone Company will collect sent-paid monies from pay telephone stations and will remit monies to the IC as set forth in Paragraph 3.6, following. The Telephone Company will provide the message call detail format and bill periods used to determine the monies upon request from the IC.

(AT)

Issued: March 26, 1993



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(MT) (FC)

(RT)

Access Services Tariff Section 3 2nd Revised Sheet 2 Replacing 1st Revised Sheet 2

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.3 Undertaking of the Telephone Company

SEP 2 5 1989

RECEIVED

- MISSOUSI
- A. Where the IC is provided with Switched Access Service undersother (sections Of of this Tariff, the Telephone Company will provide the use of Telephone Company common lines by an IC for access to End Users at rates and charges as specified in Paragraph 3.8, following.
- B. The Switched Access Service provided by the Telephone Company includes the Switched Access Service provided for both interstate and intrastate communications and the Carrier Common Line Access rates and charges as set forth in Paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, following.
 - C. When access to the local exchange is required to provide an MTS/WATS-type service using a resold Private Line Service, Switched Access Service Rates and Regulations, as set forth in Section 6, following, will apply. Carrier Common Line Access rates and charges as set forth in paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, F., following.
- (FC) D. When the IC is provided Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access as set forth in Section 6, following, the Telephone Company will collect sent-paid monies from pay telephone stations and will remit monies to the IC as set forth in Paragraph 3.6, following. The Telephone Company will provide the message call detail format and bill periods used to determine the monies upon request from the IC.

								NCELLE	
•						Pu	BY 🗟 blic Sei	R 11 1993 ² <i>R</i> . <u>5</u> vice Com ISSOUR	<u>`</u> 2
(RT)								FILE	D
	Issued:	JEP 2 5 1989		Bffective:	DCT	1	1989	OCT 1	1989
		By R.D. Sou	denses cern	President-Misso Bell Telephone Louis, Missouri	· · ··································	ivis any	rion Fublic	Service (14 Commission

Access Services Tariff

Replacing-Original-Sheet 2

JUN 27 1986

間にの

1st Revised Sheet 2

Section 3

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.3 Undertaking of the Telephone Company

- (MT) A. Where the IC is provided with Switched Access Service under other sections of this Tariff, the Telephone Company will provide the use of Telephone Company common lines by an IC for access to End Users at rates and charges as specified in Paragraph 3.8, following.
- (CP)Β. Where the customer is reselling intrastate MTS/WATS, MTS-type and/or WATS-type service(s), the customer may, at the option of the customer, employ oridinary local business exchange service at generally applicable local business exchange rates under the Telephone Company General and/or Local Exchange Tariffs. Switched Access or Carrier Common Line Access charges under this Tariff will not apply for such access except as set forth in Section 6.7, and Paragraph 3.7, D., following, provided the underlying carrier that provides the MTS/WATS and/or MTS/WATS-type service(s) which are resold pays Switched Access Services charges including Carrier Common Line Access charges for the services which are resold. Such Switched Access Service and Carrier Common Line Access charges paid shall include charges for origination and termination of calls on both ends of the services which are resold. If the customer wishes to obtain special arrangements such as trunk side service connections for such resale, the customer may, at the option of the customer, obtain Switched Access Service under this Tariff as set forth in Section 6, following. Carrier Common Line Access charges will apply for such access as set forth in Paragraph 3.7, D., following.
- (CT) C. When access to the local exchange is required to provide an MTS/WATStype service using a resold Private Line Service, Switched Access Service Rates and Regulations, as set forth in Section 6, following, will apply. Carrier Common Line Access rates and charges as set forth in paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, F., following.
- (CT) D. Where a multiline hunt group or trunk group arrangement is made up of (1) Feature Group A, B or D Switched Access Service and (2) local exchange business lines or other access arrangements associated with MTS/WATS, MTS-type or WATS-type services as provided in Paragraph 3.3, B., preceding, Carrier Common Line Access Charges for such combined access arrangements are calculated in accordance with the regulations in Paragraph 3.7, D., following CANCEL

(MT) MH 1 1986 86-84 Commission Fublic Service Commission Issued: JUN 27 1986 1 1986 JUL By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

Access Services Tariff DEUEISection 3 Original Sheet 2 DEC 20 TEC3 LIISSOURI Public Service Commission

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.3 Undertaking of the Telephone Company-(Continued)

- Where the IC is solely reselling MTS/WATS or MTS-type services, the в. Telephone Company, when the IC orders line side switched access, will not provide Switched Access Service under this Tariff, but will provide switched access such as local business exchange service under the Telephone Company's General and/or Local Exchange Service Tariffs for the locations involved. Carrier Common Line Access Charges will not apply for such access.
- Where the IC is (1) reselling Private Line Service of another IC to provide С. MTS/WATS-type service or reselling WATS-type service or reselling WATStype services of an IC and (2) the Telephone Company provides a Special Access Service, other than Dedicated Access Line Service, for the resold service terminated at the IC terminal location, the Telephone Company will, when the IC offering the resale service orders any switched access, provide Special Access Service as set forth in Section 7, Paragraph 7.4.2, B., following, Switched Access Service as set forth in Section 6, following. Carrier Common Line Access rates and charges as set forth in Paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7., G., following.
- D. When the IC plans to use Switched Access Service (line side and/or trunk side) provided as set forth in Paragraph 6.7.3, following, in association with the resale of an IC service that is provided through the use of Dedicated Access Line Service provided under this Tariff (i.e., resale of WATS), the Carrier Common Line Access rates and charges as set forth in Paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, D., following.
- Ε. The Switched Access Service provided by the Telephone Company includes the Switched Access Service provided for both interstate and intrastate communications and the Carrier Common Line Access rates and charges as set forth in Paragraph 3.8, following, apply in accordance with the regulations as set forth in Paragraph 3.7, F., following.
- F. When the IC is provided Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional_Features for sent-paid pay telephone. access as set forth in Section of following, the Telephone Company will collect sent-paid montes from pay telephone stations and will remit monies to the IC as set forth in Paragraph 3.6, following. The Telephone, Company will provide the message call detaighformat and bill periods used to book determine the monies upon request from the IC. 83-253 Pholip Service Comprission

Issued: DEC 2 9 1983 OF MISSOUN JAN 0 1 1984 Effective:

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

<u>.S#2</u> SERVICE COMMISSION

BYLAR

PUBIIC

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 3rd Revised Sheet 3 Replacing 2nd Revised Sheet 3

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.4 Obligations of the IC
 - A. The Switched Access Service associated with Carrier Common Line access shall be ordered by the IC under other sections of this Tariff.
 - B. The IC facilities at the IC terminal location shall provide the necessary on-hook and off-hook supervision.
- C. Unless the IC reports interstate use as set forth in Paragraph 3.4, D., following, or FGA, FGB, FGD, BSA-A, BSA-B or BSA-D, Switched Access Service as set forth in Paragraph 3.4, F., following, all Switched Access Service provided under this Tariff ordered by the IC will be subject to Carrier Common Line Access charges.
 - D. When the IC reports interstate and intrastate use of Switched Access Service, the associated Carrier Common Line Access used by the IC for intrastate will be determined as set forth in Paragraph 3.7, following.
- (AT) E. Where FGC or BSA-C end office switching is provided without Telephone Company recording and the IC records minutes of use which will be used to determine Carrier Common Line Access
 (AT) Charges (i.e., FGC or BSA-C operator and TSPS calls such as pay telephone sent-paid, operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls), the IC shall furnish such minutes of use detail to the Telephone Company in a timely manner. If the IC does not furnish the data to the Telephone Company, the IC shall identify all Switched Access Services which could carry such calls in order for the Telephone Company to accumulate the minutes of use through the use of special Telephone Company measuring and recording equipment.

(AT)

(AT)

Issued: March 26, 1993



No supplement to this tariff vill be issued except for the purpose of canceling this tariff.

3.

(HT)

Access Services Tariff Section 3 2nd Revised Sheet 3 Replacing 1st Revised Sheet 3 RECEIVED

ACCESS SERVICES

SEP 2 5 1989

3.4 Obligations of the IC

CARRIER COMMON LINE ACCESS SERVICE-(Continued)

MISSOURI

Public Service Commission

- Α. The Switched Access Service associated with Carrier Common Line access shall be ordered by the IC under other sections of this Tariff.
- The IC facilities at the IC terminal location shall provide the necessary **B**. on-hook and off-hook supervision.
- С. Unless the IC reports interstate use as set forth in Paragraph 3.4, D., following, or Feature Group A, B or D, Switched Access Service as set forth in Paragraph 3.4, F., following, all Switched Access Service provided under this Tariff ordered by the IC will be subject to Carrier Common Line Access charges.
- When the IC reports interstate and intrastate use of Switched Access D. Service, the associated Carrier Common Line Access used by the IC for intrastate will be determined as set forth in Paragraph 3.7, following.
- E. Where Feature Group C end office switching is provided without Telephone Company recording and the IC records minutes of use which will be used to determine Carrier Common Line Access Charges (i.e., Feature Group C operator and TSPS calls such as pay telephone sent-paid, operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls), the IC shall furnish such minutes of use detail to the Telephone Company in a timely manner. If the IC does not furnish the data to the Telephone Company, the IC shall identify all Switched Access Services which could carry such calls in order for the Telephone Company to accumulate the minutes of use through the use of special Telephone Company measuring and recording equipment.

CANCELLED

APR 11 1993 #3 BY 3rd R.S **Public Service Commission** MISSOURI

FILED

007 1 1989

Issued:

SEP 2 5 1989

Bffective: OCT 1 1989 89-14 By R.D. BARRON, President-Missouri Division ublic Service Commissio Southwestern Bell Telephone Company St. Louis, Missouri

1

-

ī

I

		pplement to this f will be issued	Access Services Tariff Section 3
		t for the purpose	lst Revised Sheet 3
		nceling this tariff.	Replacing_Original_Sheet_3
		ACCESS SERVI	CES REGEIVED
	3. C	ARRIER COMMON LINE ACCESS SERVICE-(Cont	inued)
	3.3	Undertaking of the Telephone Company-(Continued) JUN 27 1986
(MT)	E.	The Switched Access Service provided b the Switched Access Service provided f communications and the Carrier Common set forth in Paragraph 3.8, following, regulations as set forth in Paragraph	or both interstate and intrastate Line Access rates and charges lassion apply in accordance with the
	_		
(MT)	F.	When the IC is provided Operator Trunk or Operator Trunk-Full Feature Optiona access as set forth in Section 6, foll collect sent-paid monies from pay tele to the IC as set forth in Paragraph 3. will provide the message call detail f determine the monies upon request from	1 Features for sent-paid pay telephone owing, the Telephone Company will phone stations and Cill remit monies 6, following. The Telephone Company ormat and bill periods used to 35 the IC.
	3.4	Obligations of the IC	Public Schuce Commission
	A.	The Switched Access Service associated be ordered by the IC under other secti	with Carrier Common Line Access shall
•	B.	The IC facilities at the IC terminal 1 on-hook and off-hook supervision.	ocation shall provide the necessary
	C.	Unless the IC reports interstate use a following, or Feature Group A, B or D, forth in Paragraph 3.4, F., following, provided under this Tariff ordered by Carrier Common Line Access charges.	Switched Access Service as set all Switched Access Service
(CT)	D.	When the IC reports interstate and int	rastate use of Switched Access Service,
(CT)		the associated Carrier Common Line Acc will be determined as set forth in Par	ess used by the IC for intrastate
Û Î	Ε.	the IC shall furnish such minutes of u a timely manner. If the IC does not be pany, the IC shall identify all Switch such calls in order for the Telephone use through the use of special Telephone equipment.	hinutes of use which will be used to charges (i.e., Feature Group C elephone sent-paid, operator-DDD, third-number and/or other like calls), use detail to the Telephone Company in furnish the data to the Telephone Com- ed Access Services which could carry Company to accumulate the minutes of
-	Issue		fective: JUL 1 1986
		By R. D. BARRON, Presic Southwestern Bell St. Louis	Telephone Company

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

DEC 20 1003 MISSOUKI Public Service Commission

Access Services Tariff

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.4 Obligations of the IC
- A. The Switched Access Service associated with Carrier Common Line Access shall be ordered by the IC under other sections of this Tariff.
- B. The IC facilities at the IC terminal location shall provide the necessary on-hook and off-hook supervision.
- C. Unless the IC reports interstate use as set forth in Paragraph 3.4, D., following, trunk side Switched Access Service for use in association with the resale of IC services as set forth in Paragraph 3.4, F., following, line side Switched Access Service as set forth in Paragraph 6.7.3, following or exchange use as set forth in Paragraph 3.4, G., following, all Switched Access Service provided under this Tariff ordered by the IC will be subject to Carrier Common Line Access charges.
- D. When the IC reports interstate use of Switched Access Service, the associated Carrier Common Line Access used by the IC for both interstate and intrastate will be apportioned as set forth in Paragraph 3.7, F., following.
- E. Where Feature Group C end office switching is provided without Telephone Company recording and the IC records minutes of use which will be used to determine Carrier Common Line Access Charges (i.e., Feature Group C operator and TSPS calls such as pay telephone sent-paid, operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls), the IC shall furnish such minutes of use detail to the Telephone Company in a timely manner. If the IC does not furnish the data to the Telephone Company, the IC shall identify all Switched Access Services which could carry such calls in order for the Telephone Company to accumulate the minutes of use through the use of special Telephone Company measuring and recording equipment.

GANGELLED FILED JUL 1 1986 JAN - 1 1931 BYIST PUBLIC SERVICE COMMISSION OF MISSOUN

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.4 Obligations of the IC-(Continued)
 - F. Where Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access is provided to the IC and the IC wishes to receive the monies it is due for the monies collected by the Telephone Company from coin pay telephone stations, the IC shall furnish to the Telephone Company at a location specified by the Telephone Company, The IC message call detail for the IC sent-paid (coin) pay telephone calls in accordance with the Telephone Company collection schedule. The IC message call detail furnished shall be in a standard format establishment by the Telephone Company. If no IC message call detail is received from the IC for each bill period established by the Telephone Company, the Telephone Company will assume there were no IC sent-paid (coin) pay telephone calls for the period. In addition, the IC shall furnish a schedule of its charges for sent-paid (coin) calls to the Telephone Company at a location and date as specified by the Telephone Company. Any change in the IC's schedule of charges shall be furnished to the Telephone Company one day after the change becomes effective.
- G. Customers ordering 800 NPAS must provide to the Telephone Company a percent Intrastate-IntraLATA Carrier Common Line report as specified in Section 6, Paragraph 6.9.1, following, for proper rate application of Carrier Common Line charges, when appropriate.

Issued: March 22, 1993



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 4t**Precised Section** 4 Replacing 3rd Revised Sheet 4

MAR 29 1993

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

MISSOURI Public Service Commission

- 3.4 Obligations of the IC-(Continued)
- Where Operator Trunk-Coin or Combined Coin and Non-Coin or Operator F. Trunk-Full Feature Optional Features for sent-paid pay telephone access is provided to the IC and the IC wishes to receive the monies it is due for the monies collected by the Telephone Company from coin pay telephone stations, the IC shall furnish to the Telephone Company at a location specified by the Telephone Company, The IC message call detail for the IC sent-paid (coin) pay telephone calls in accordance with the Telephone Company collection schedule. The IC message call detail furnished shall be in a standard format establishment by the Telephone Company. If no IC message call detail is received from the IC for each bill period established by the Telephone Company, the Telephone Company will assume there were no IC sent-paid (coin) pay telephone calls for the period. In addition, the IC shall furnish a schedule of its charges for sent-paid (coin) calls to the Telephone Company at a location and date as specified by the Telephone Company. Any change in the IC's schedule of charges shall be furnished to the Telephone Company one day after the change becomes effective.
- G. Customers ordering non-joint provided 800 Access Service must provide to the Telephone Company a percent Intrastate-IntraLATA Carrier Common Line report as specified in Section 6, Paragraph 6.9.1, following, for proper rate application of Carrier Common Line charges.

CANCELLED

MAY 1 1993 BY 5 R .S. #4 Public Service Commission MISSOURI

FILED

APR 1 1 1993 92 - 304 MO. PUBLIC SERVICE COMM.

Issued:

(FC)

: MAR 2 6 1993

Effective:

APR 1 1 1993

By R.D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 3 3rd Revised Sheet 4 Replacing 2nd Revised Sheet 4

No supplement to this tariff vill be issued except for the purpose of canceling this tariff.

(RT)

(RT)

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.4 Obligations of the IC-(Continued)

RECEIVED

SEP 2 5 1989

MISSOURI Public Service Commission

CANCELLED

APR 11 1993 BY <u>+ th R.S. * +</u> Public Service Commission MISSOURI

(FC) F. Where Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access is provided to the IC and the IC vishes to receive the monies it is due for the monies collected by the Telephone Company from coin pay telephone stations, the IC shall furnish to the Telephone Company at a location specified by the Telephone Company, The IC message call detail for the IC sent-paid (coin) pay telephone calls in accordance with the Telephone Company collection schedule. The IC message call detail furnished shall be in a standard format establishment by the Telephone Company. If no IC message call detail is received (RT) from the IC for each bill period established by the Telephone Company, the (RT) Telephone Company will assume there were no IC sent-paid (coin) pay telephone calls for the period. In addition, the IC shall furnish a schedule of its charges for sent-paid (coin) calls to the Telephone Company at a location and date as specified by the Telephone Company. Any change in the IC's schedule of charges shall be furnished to the Telephone Company one day after the change becomes effective.

(FC) G. Customers ordering non-joint provided 800 Access Service must provide to the Telephone Company a percent Intrastate-IntraLATA Carrier Common Line report as specified in Section 6, Paragraph 6.6.1 following, for proper rate application of Carrier Common Line charges.

FILED

Issued: SEP 2 5 1989 By R.D. BARRON, President-Missouri Division^OUblic Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3

- 2nd Revised Sheet 4
- **Replacing 1st Revised Sheet 4**

DEC 2

1988

ACCESS SERVICES CANCELLED RECEIVED

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued) $_{0CT 1}$ 1989
 - 3.4 Obligations of the IC-(Continued)
 - By Br R S. #4 MISSOUNI
 By Br R S. #4 MIS charges in accordance with the regulations as set forth in the Paragraph 3.7, D., following, if the customer or the provider of the WATS service furnishes documentation of the WATS usage and/or the customer furnishes documentation of the WATS-type service. The documentation supplied by the customer shall be supplied each month as set forth in Section 6, Paragraph 6.7.3, A., following.
 - G. When the customer orders Switched Access Service as set forth in Paragraph F., preceding, the Telephone Company may request when resold MTS/WATS is involved, a certified copy of the customer's MTS/WATS usage billing from either the customer or the provider of the MTS/WATS service. When resold MTS/WATS-type service is involved, a certified copy of customer's MTS/VATS-type usage billing will be requested from the customer. The requests for this billing will relate back no more than 12 months prior to the current billing period.
 - Where Operator Trunk-Coin or Combined Coin and Non-Coin or Operator H. Trunk-Full Peature Optional Peatures for sent-paid pay telephone access is provided to the IC and the IC wishes to receive the monies it is due for the monies collected by the Telephone Company from coin pay telephone stations, the IC shall furnish to the Telephone Company at a location specified by the Telephone Company, The IC message call detail for the IC sent-paid (coin) pay telephone calls in accordance with the Telephone Company collection schedule. The IC message call detail furnished shall be in a standard format establishment by the Telephone Company as set forth in Section 8, Paragraph 8.2.1, B.1.c., following. If no IC message call detail is received from the IC for each bill period established by the Telephone Company, the Telephone Company vill assume there were no IC sent-paid (coin) pay telephone calls for the period. In addition, the IC shall furnish a schedule of its charges for sent-paid (coin) calls to the Telephone Company at a location and date as specified by the Telephone Company. Any change in the IC's schedule of charges shall be furnished to the Telephone Company one day after the change becomes effective.
- (AT) I. Customers ordering non-joint provided 800 Access Service must provide to the Telephone Company a percent Intrastate-IntraLATA Carrier Common Line report as specified in Section 6 Paragraph 6.6.1 following, for proper rate (AT) application of Carrier Common Line charges.

FILED

- 86

89

Issued: DEC 2 1988

Effective: DEC 1 4 1988 By R.D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

· 1

• -

T

ĺ

i

:

ļ

ļ

Т

1 . 1

1

	tarif excep	upplement to this ff will be issued ot for the purpose anceling this tariff.		Access Services Tariff Section 3 lst Revised Sheet 4 Replacing Original Sheet 4
			ACCESS SERVICES	REGEIVED
	3. 0	CARRIER COMMON LINE ACCESS	SERVICE-(Continued) .	
	3.4	Obligations of the IC-(Co	ntinued)	JUN 2 7 1980
(CP)	F.	When the customer orders preceding, the customer w charges in accordance wit 3.7, D., following, if th service furnishes documen customer furnishes docume documentation supplied by as set forth in Paragraph	ill be charged the Car h the regulations as s e customer or the prov tation of the WATS usa ntation of the WATS-ty the customer shall be	rier Common, Line Access et forth in Paragraph ISSION ider of the WATS ge and/or the pe service. The supplied each month
(RT)				
(AT)	G.	from either the customer When resold MTS/WATS-type	Telephone Company may fied copy of the custor or the provider of the service is involved, usage billing will be ling will relate back	request when resold MTS/ mer's MTS/WATS usage billing MTS/WATS service. a certified copy of requested from the reuseomer.
(RT)				THE MAR S. F. Signission
(FC) (MT) (MT)	Η.	it is due for the monies coin pay telephone statio	collected by the Telepins, the IC shall furni cified by the Telephon nt-paid (coin) pay tel hone Company collectio shed shall be in a sta pany as set forth in S If no IC message cal period established by l assume there were no he period. In additio for sent-paid (coin) date as specified by	sh to the Telephone e Company, the IC message ephone calls in n schedule. The IC ndard format establish- ection 8, Paragraph l detail is received the Telephone Company, IC sent-paid (coin) n, the IC shall furnish calls to the Telephone the Telephone Company
	Issue	ed: JUN 2 7 1986	Effective:	JUL 1986 Service Commission
		Southwes	ON, President-Missouri tern Bell Telephone Co t. Louis, Missouri	Division mpany

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Issued:

DEC 2 9 1983

ACCESS SERVICES

CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.4 Obligations of the IC-(Continued)

- When the IC uses a line side or trunk side Switched Access Service provided F. under this Tariff in association with the resale of an IC service that is provided through the use of Dedicated Access Line Service provided under this Tariff (i.e., resale of WATS), the IC will be charged the Carrier Common Line Access Charges in accordance with the regulations as set forth in Paragraph 3.7, D., following, if it reports the following data. The IC shall report the line side or trunk side Switched Access Services involved, the number of resold Dedicated Access Line Services with 2,500'access minutes or more per month, and the IC terminal locations involved as set forth in Section 6, Paragraph 6.7.3, A., B. and C., following.
- When the IC uses a Switched Access Service to furnish a service which is G. defined by the appropriate Regulatory Commission to be an exchange service, the IC may apply for credit to its monthly transitional charges in accordance with the regulations as set forth in Paragraph 3.7, E., following, by reporting the originating and terminating access minutes that both originate and terminate in the same local calling area. The local calling area is determined as set forth in Telephone Company General and/or Local Exchange Service Tariffs and is a geographical area associated with an exchange within which calls may be made without MTS charges. The IC shall report the access minutes for credit on a calendar-month basis, the name of the service and the Regulatory Commission which defines the service as an exchange service.
- When the IC notifies the Telephone Company that Switched Access Service is to H. be used as set forth in Paragraphs 3.4, D., F. and G., preceding, and Paragraph 6.7.3, following, the notification automatically authorizes the Telephone Company the right to audit the IC terminal locations to verify the use of facilities as the IC has reported in Paragraphs 3.4, D., F. or G., preceding, or Paragraph 6.7.3, following, and all of the records, workpapers and backup documentation for each report as set forth in Paragraphs 3.4, D., F. or G., preceding, or Paragraph 6.7.3, following, and contact and review the records of other entities involved to verify the data the IC reports is accurate. All of the records, workpapers and backup documentation for each report furnished to the Telephone Company as set forth in Paragraphs 3.4, D., F. or G., preceding, or Paragraph 6.7.3, following, shall be available for one year from the date of the report and shall be made available during normal business hours at an IC location in the involved-LATA upon__ reasonable request by the Telephone Company in order to permit a review by the Telephone Company auditor of our auditor under contract to the Telephone Company. If the records, workpapers and backup documentation are not provided or are insufficient or not in accordance with the provision of JULI 83 - 253 Duhlin Canitan 0 BY 17

PUBLIC SERVICE COMPLETION

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

OF MISSOURI

JAN 0 1 1984

1:0010

Access Services Tariff Section 3 BECOriginal Sheet 4 DEC 20 1203 **MISSOURI** Public Service Commission

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.5 Payment Arrangements

- A. The Telephone Company will bill the Carrier Common Line Access. The bill day (i.e., the billing date of the bill) in a month for each IC account will be established by the Telephone Company. Payment is due from the IC 31 days after the bill day date (payment date) or by the next bill date (i.e., same date in the following month as the bill date), whichever is the shortest interval, and is payable in immediately available funds. If such payment date is a Saturday, Sunday or holiday (i.e., New Year's Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day, the second Tuesday in November, and a day when Washington's Birthday, Memorial Day or Veteran's Day is legally observed), payment will be due from the IC as follows: If the payment date falls on a Saturday, Sunday or on a holiday which is observed on a Monday, the payment date shall be the first non-holiday day following such Saturday, Sunday or holiday.
- B. Further, if any portion of the Carrier Common Line Access payment is received by the Telephone Company after the payment date as set forth in Paragraph 3.5, A., preceding, or if any portion of the Carrier Common Line Access payment is received by the Telephone Company in funds which are not immediately available to the Telephone Company, then a late-payment penalty shall be due to the Telephone Company. The late-payment penalty shall be the portion of the Carrier Common Line Access payment not received by the payment date times a late factor. The late factor shall be the lesser of:
 - 1. The highest interest rate (in decimal value) which may be levied by law for commercial transactions, compounded daily for the number of days from the payment date to and including the date that the IC actually makes the payment to the Telephone Company, or
 - 2. 0.000590 per day, compounded daily for the number of days from the payment date to and including the date that the IC actually makes the payment to the Telephone Company.
- C. In the event a billing dispute concerning a month's Carrier Common Line Access billed to the IC by the Telephone Company is resolved in favor of the Telephone Company, any payments withheld pending settlement of the dispute shall be subject to the late-payment penalty set forth in Paragraph 3.5, B., preceding. If the customer disputes the bill on or before the payment date, and pays the undisputed amount on or before

Issued: January 8, 1988

Effective: February 8, 1988

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



(CR)

ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tarif:	f
Section 3	
lst Revised Sheet	
Replacing Original Sheet	5
REGEIVED	A

JUN 27 1990

PLOTE SCIERCE COMMISSION

1 1986

JUL

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.5 Payment Arrangements
 - A. The Telephone Company will bill the Carrier Common Line Accession The bill day (i.e., the billing date of the bill) in a month for each IC account of will be established by the Telephone Company. Payment is due from the IC-31 days after the bill day date (payment date) or by the next bill date (i.e., same date in the following month as the bill date), whichever is the shortest interval, and is payable in immediately available funds. If such payment date is a Saturday, Sunday or holiday (i.e., New Year's Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day, the second Tuesday in November, and a day when Washington's Birthday, Memorial Day or Veteran's Day is legally observed), payment will be due from the IC as follows: If the payment date falls on a Saturday, Sunday or on a holiday which is observed on a Monday, the payment date shall be the first non-holiday day following such Saturday, Sunday or holiday.
- (MT) B. Further, if any portion of the Carrier Common Line Access payment is received by the Telephone Company after the payment date as set forth in Paragraph 3.5, A., preceding, or if any portion of the Carrier Common Line Access payment is received by the Telephone Company in funds which are not immediately available to the Telephone Company, then a late-payment penalty shall be due to the Telephone Company. The late-payment penalty shall be the portion of the Carrier Common Line Access payment not received by the payment date times a late factor. The late factor shall be the lesser of:
 - 1. The highest interest rate (in decimal value) which may be levied by law for commercial transactions, compounded daily for the number of days from the payment date to and including the ALLED that the IC actually makes the payment to the Telephone Company, or
 - 2. 0.000657 per day, compounded daily for the number of days from days from the payment date to and including the date that the IC actual rvice Commission makes the payment to the Telephone Company. Public MISSOURI

(CP) C. In the event a billing dispute concerning a month's Carrier Common Line Access billed to the IC by the Telephone Company is resolved in favor of the Telephone Company, any payments withheld pending settlement of the U dispute shall be subject to the late-payment penalty set forth in Paragraph 3.5, B., preceding. If the customer disputes the bill on or 1986 before the payment date, and pays the undisputed amount on or before

Issued: JUN 27 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

P.S.C. MoNo. 3	P.	.s.c	. Mo	No		36
----------------	----	------	------	----	--	----

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CAI	RIER	COMMON	LINE	ACCESS	SERVICE-((Continued)
--------	------	--------	------	--------	-----------	-------------

3.4 Obligations of the IC-(Continued)

H. (Continued)

this paragraph and Paragraphs 3.4, D., F. and G., preceding, and Paragraph 6.7.3, following, the adjustments and rates as set forth in Paragraphs 3.7, D., E. and F., following, shall not apply until the deficiencies are corrected and new reports as set forth in Paragraphs 3.4, D., F. and G., preceding, and Paragraph 6.7.3, following are delivered to the Telephone Company.

I. Where Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access is provided to the IC and the IC wishes to receive the monies it is due for the monies collected by the Telephone Company from coin pay telephone stations, the IC shall furnish to the Telephone Company at a location specified by the Telephone Company, the IC message call detail for the IC sent-paid (coin) pay telephone calls in accordance with the Telephone Company collection schedule. The IC message call detail furnished shall be in a standard format establishment by the Telephone Company as set forth in Section 8, Paragraph 8.2.1, B.l.c., following. If no IC message call detail is received from the IC for each bill period established by the Telephone Company, the Telephone Company will assume there were no IC sent-paid (coin) pay telephone calls for the period. In addition, the IC shall furnish a schedule of its charges for sent-paid (coin) calls to the Telephone Company at a location and date as specified by the Telephone Company. Any change in the IC's schedule of charges shall be furnished to the Telephone Company one day after the change becomes effective.

3.5 Payment Arrangements

A. The Telephone Company will bill the Carrier Common Line Access. The bill day (i.e., the billing date of the bill) in a month for each IC account will be established by the Telephone Company. Payment is due from the IC 31 days after the bill day date (payment date) or by the next bill date (i.e., same date in the following month as the bill date), whichever is the shortest interval, and the bill bill date funds. If such payment date is all baturday, Sunday or holiday (i.e., New Year's Day, Thdependence Day, Labor Day, Thanksgiving Day, Christmas Day, the second Tuesday in November, and a day, when Wasobington's Birthday, Memorial Day_or Veteran's Day is legally observed), payment will be due from the IC as follows:

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

BY INK S. #

PUBLIC SERVICE COMMISSION

DEC 20 1983	
ENISSOURI Public Service Commission	1

Or E that Sheet 5

Section 3

Access Services Tariff

No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 1st Revised Sheet 6 Replacing Original Sheet 6

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.5 Payment Arrangements-(Continued)
 - C. (Continued)

the payment date, any late payment charge for the disputed amount will not start until 10 days after the payment date. If the billing dispute is resolved in favor of the customer, no late payment penalty will apply to the disputed amount. In addition, if the customer disputes the billed amount and pays the total amount (i.e., the nondisputed amount and the disputed amount) on or before the payment date and the billing dispute is resolved in the favor of the customer, the customer will receive a credit for a disputed amount penalty from the Telephone Company if the billing dispute is not resolved within 10 working days following the payment date or the date the customer furnishes to the Telephone Company documentation to support its claim plus 10 working days, whichever date is the later date. The disputed amount penalty shall be the disputed amount resolved in the customer's favor times a penalty factor found in Paragraph B., preceding.

Issued: June 27, 1986



ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Orfging Sheet () DEC 2 0 1283 MISSOURI Public Service Commission

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.5 Payment Arrangements-(Continued)

A. (Continued)

If the payment date falls on a Saturday, Sunday or on a holiday which is observed on a Monday, the payment date shall be the first non-holiday day following such Saturday, Sunday or holiday.

- B. Further, if any portion of the Carrier Common Line Access payment is received by the Telephone Company after the payment date as set forth in Paragraph 3.5, A., preceding, or if any portion of the Carrier Common Line Access payment is received by the Telephone Company in funds which are not immediately available to the Telephone Company, then a late-payment penalty shall be due to the Telephone Company. The latepayment penalty shall be the portion of the Carrier Common Line Access payment not received by the payment date times a late factor. The late factor shall be the lesser of:
 - 1. The highest interest rate (in decimal value) which may be levied by law for commercial transactions, compounded daily for the number of days from the payment date to and including the date that the IC actually makes the payment to the Telephone Company, or
 - 2. 0.000657 per day, compounded daily for the number of days from the payment date to and including the date that the IC actually makes the payment to the Telephone Company.
- C. In the event a billing dispute concerning a month's Carrier Common Line Access billed to the IC by the Telephone Company is resolved in favor of the Telephone Company, any payments withheld pending settlement of the dispute shall be subject to the late-payment penalty set forth in Paragraph 3.5, B., preceding. If the billing dispute is resolved in favor of the IC, no late payment penalty will apply to the disputed amount.



JUL 1 1986 FILEN BY IRL R.S. #6 PUBLIC SERVICE COMMISSION JAN - 1 (SC) OF MISSOURI 83-253 Ht.n_p. -

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 7 Replacing 1st Revised Sheet 7

ACCESS SERVICES

(RŢ)

(RT)

Issued: June 24, 1994

Effective: August 17, 1994

By M. H. SCHULTEIS, Executive Director-External Affairs Southwestern Bell Telephone St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 1st Revised Sheet 7 Replacing Original Sheet 7

REGEIVED

JUN 2 7 1985

BRA CONRICO COMMISSION

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- 3.6 Payment of Coin Sent-Paid Monies

The Telephone Company will collect the monies from coin pay telephone LLED stations and will determine and remit amounts due to an ICI which is provided Operator Trunk-Coin or Combined Coin and Non-Cointor Operator () Trunk-Full Feature Optional Features for sent-paid pay telephone-access 1994 as set forth in Section 6, following.

A. Bill Period Coin Revenue

The Telephone Company will establish a collection schedule for each NUSSOURI pay telephone station and will collect the monies from the coin and this schedule. this schedule during each bill period established by the Telephone Company will be identified by coin pay telephone station and summed to develop the Bill Period Coin Revenue for each coin record day (i.e., the day a record is prepared and dated to show the amount due the IC).

(AT)

Total IC Coin Revenue Β.

> The intrastate total IC Coin Revenue will be determined by the Telephone Company based on the IC message call detail received from the IC for each bill period and the IC's schedule of charges for sent-paid coin calls. Such total IC Coin Revenue will be developed each coin record day.

C. Recourse Adjustments

For each coin record day, the Telephone Company will subtract from the total IC Coin Revenue an amount for coin station shortages. Coin station shortages are amounts resulting from unauthorized calling at coin pay telephone stations, use of unauthorized coins (i.e., foreign coins, slugs and improper use of U.S. pennies), unauthorized removal of coins from coin pay telephone stations and coin refunds beyond the Telephone Company's control. Such amount for coin station shortages will be developed by the Telephone Company by multiplying the total IC Coin Revenue for each coin record day by a shortage factor. Such amount will be rounded to the nearest penny. The shortage factor will be determined by dividing the yearly total coin shortage amount by the yearly total coin revenue amount (i.e., total coin revenue equals the coin revenue, due under exchange tariffs, state toll tariffs, and interstate toll [5] tariffs). The total coin shortage amount and the total revenue amount will be determined by the Telephone Company through an annual special 1986 study. 86-84

Issued: JUN 27 1986

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

JUL

1

Access Services Tariff

DEC 28 1003

I.3ISSOURI
 Public Service Commission

たいに目し

JAN - 1 (92)

Public 8, 3 - 253

REGEIVEU

Section 3

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.6 Payment of Coin Sent-Paid Monies

The Telephone Company will collect the monies from coin pay telephone stations and will determine and remit amounts due to an IC which is provided Operator Trunk-Coin or Combined Coin and Non-Coin or Operator Trunk-Full Feature Optional Features for sent-paid pay telephone access as set forth in Section 6, following.

A. Bill Period Coin Revenue

The Telephone Company will establish a collection schedule for each coin pay telephone station and will collect the monies from the coin pay stations based on this collection schedule. The monies collected based on this schedule during each bill period established by the Telephone Company will be identified by coin pay telephone station and summed to develop the Bill Period Coin Revenue for each coin record day (i.e., the day a record is prepared and dated to show the amount due the IC).

B. Total IC Coin Revenue

The intrastate total IC Coin Revenue will be determined by the Telephone Company based on the IC message call detail received from the IC for each bill period and the IC's schedule of charges for sent-paid coin calls. Such total IC Coin Revenue will be developed each coin record day.

C. Recourse Adjustments

For each coin record day, the Telephone Company will subtract from the total IC Coin Revenue an amount for coin station shortages. Coin station shortages are amounts resulting from unauthorized calling at coin pay telephone stations, use of unauthorized coins (i.e., foreign coins, slugs and improper use of U.S. pennies), unauthorized removal of coins from coin pay telephone stations and coin refunds beyond the Telephone Company's control. Such amount for coin station shortages will be developed by the Telephone Company by multiplying the total IC Coin Revenue for each coin record day by a shortage factor. Such amount will be rounded to the nearest beint. The shortage factor will be determined by the Telephone Company through an annual special study.

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI

Lasued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 3rd Revised Sheet 8 Replacing 2nd Revised Sheet 8

ACCESS SERVICES

(RT)

3.7 Rate Regulations

- A. The transitional charges will be billed per access minute to each IC Switched Access Service provided under this Tariff in accordance with the regulations as set forth in Paragraph 3.7, following.
- B. The access minutes which are used to determine the transitional charges will be accumulated using call detail recorded by Telephone Company equipment, except as set forth in Paragraph 3.7, C., following, the FGC, BSA-C and operator and TSPS call detail such as pay telephone sent-paid.

Issued: June 24, 1994

Effective: August 17, 1994

By M. H. SCHULTEIS, Executive Director-External Affairs Southwestern Bell Telephone St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 8 Replacing 1st Revised Sheet 8

RECEIVED

MAR 29 1993

MISSOURI

Public Service Commission

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.6 Payment of Coin Sent-Paid Monies-(Continued)

D. Payment of Net IC Coin Revenue

The Telephone Company will determine the net IC Coin Revenue for each coin record day by subtracting from the total IC Coin Revenue determined as set forth in Paragraph 3.6, B., preceding, the amount for coin station shortages determined as set forth in Paragraph 3.6, C., preceding. On the date (payment date) determined by adding 45 days to the coin record day, the Telephone Company will remit payment to the IC for the net IC Coin Revenue.

E. Audit Provisions

Upon reasonable written notice by the IC to the Telephone Company, the IC shall have the right through its authorized representative to examine and audit, during normal business hours and at reasonable intervals as determined by the Telephone Company, all such records and accounts as may under recognized accounting practices contain information bearing upon the determination of coin revenues for which amounts may be payable to the IC. Adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit. Neither such right to examine and audit nor the right to receive such adjustment shall be affected by any statement to the contrary, appearing on checks or otherwise, unless such statement expressly waiving such right appears in a letter signed by the authorized representative of the party having such right and delivered to the other party.

All information received or reviewed by the IC or its authorized representative is to be considered confidential and is not to be distributed, provided or disclosed in any form to anyone not involved in the audit, nor is such information to be used for any other purpose.

3.7 Rate Regulations

- The transitional charges will be billed per access minute to each IC Α. Switched Access Service provided under this Tariff in accordance with the regulations as set forth in Paragraph 3.7, following.
- **B**. The access minutes which are used to determine the transitional charges will be accumulated using call detail recorded by Telephone Company equipment, except as set forth in Paragraph 3.7, C., following, the FGC, BSA-C and operator and TSPS call detail such as pay telephone sent-paid. CANCELLED

APR 11 1993 92 - 30 & Effective AUG 17 1994 APR 1

APR 1 1 1993

Public Service Commission

(AT)

Issued:

MAR 2 6 1993

Southwestern Bell Telephone Company St. Louis, Missouri

By R. D. BARRON, PHOIBURICASSOUTI DIVISION



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access	Services	: Tarij	Ef
	Se	ection	3
lst	Revised	Sheet	8
Replacing (Driginal	Sheet	8

REGEIVED

JUN 27 1985

APR 11 1993

1986

JUL 1

<u>86 - 84</u>

ACCESS SERVICES

CARRIER COMMON LINE ACCESS SERVICE-(Continued) 3.

3.6 Payment of Coin Sent-Paid Monies-(Continued)

Payment of Net IC Coin Revenue D.

> BIIZZOOK: The Telephone Company will determine the net IC Coin Revenue for each coin record day by subtracting from the total IC Coin Revenue determined as set forth in Paragraph 3.6, B., preceding, the amount for coin station shortages determined as set forth in Paragraph 3.6, C., preceding. On the date (payment date) determined by adding 45 days to the coin record day, the Telephone Company will remit payment to the IC for the net IC Coin Revenue.

E. Audit Provisions (AT)

> Upon reasonable written notice by the IC to the Telephone Company, the IC shall have the right through its authorized representative to examine and audit, during normal business hours and at reasonable intervals as determined by the Telephone Company, all such records and accounts as may under recognized accounting practices contain information bearing upon the determination of coin revenues for which amounts may be payable to the IC. Adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit. Neither such right to examine and audit nor the right to receive such adjustment shall be affected by any statement to the contrary, appearing on checks or otherwise, unless such statement expressly waiving such right appears in a letter signed by the authorized representative of the party having such right and delivered to the other party.

All information received or reviewed by the IC or its authorized representative is to be considered confidential and is not to be distributed, provided or disclosed in any form to anyone not involved in thCANCEL nor is such information to be used for any other purpose.

(CP) 3.7 Rate Regulations

The transitional charges will be billed per access minut Α. Switched Access Service provided under this Tariff in actual Service Commission the regulations as set forth in Paragraph 3.7, following. MISSOURI

Effective:

JUL

The access minutes which are used to determine the transitional charges B. will be accumulated using call detail recorded by Telephone Company-equipment, except as set forth in Paragraph 3.7, C., following, the Feature Group C and operator and TSPS call detail such as pay telephone sent-paid,

Issued: JUN 27 1986

1 1986 no Service Commission By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.6 Payment of Coin Sent-Paid Monies-(Continued)

D. Payment of Net IC Coin Revenue

The Telephone Company will determine the net IC Coin Revenue for each coin record day by subtracting from the total IC Coin Revenue determined as set forth in Paragraph 3.6, B., preceding, the amount for coin station shortages determined as set forth in Paragraph 3.6, C., preceding. On the date (payment date) determined by adding 45 days to the coin record day, the Telephone Company will remit payment to the IC for the net IC Coin Revenue.

- 3.7 Rate Regulations
- A. The transitional charges will be billed per access minute to each IC Switched Access Service provided under this Tariff in accordance with the regulations as set forth in Paragraph 3.7, G., following, except as set forth in Paragraphs 3.7, D., E. and F., following.
- B. The access minutes which are used to determine the transitional charges will be accumulated using call detail recorded by Telephone Company equipment, except as set forth in Paragraph 3.7, C., following, and Feature Group C operator and TSPS call detail such as pay telephone sent-paid, operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls recorded by the IC. The Telephone Company measuring and recording equipment, except as set forth in Paragraph 3.7, C., following, will be associated with end office or local tandem switching equipment and will record each originating and terminating access minutes where answer supervision is received. The accumulated access minutes will be summed by end office for each IC and then rounded to the nearest minute.
- C. When Carrier Common Line Access is provided in association with Feature Groups A and B end office switching in Telephone Company end offices that are not equipped for measurement capabilities, assumed average access minutes will be used to determine the transitional charges. The assumed average access minutes with the state of the second second

同心臣间 JUL 1 1986 JAN - 1 (981 At R S #9 PUBLIC SERVICE COMMISSION 83-253 OF MISSOUN

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Original Sheet 8 DEC 20 1983 LAISSOURI

Access Services Tariff

Section 3

Public Service Commission

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 3rd Revised Sheet 9 Replacing 2nd Revised Sheet 9

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
 - B. (Continued)

operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls recorded by the IC. The Telephone Company measuring and recording equipment, except as set forth in Paragraph 3.7, C., following, will be associated with end office or local tandem switching equipment and will record each originating and terminating access minute where answer supervision is received. The accumulated access minutes will be summed on a line-by-line basis, by line group or by end office, whichever type of account is used by the Telephone Company, for each IC and then rounded to the nearest minute.

- C. When Carrier Common Line Access is provided in association with FGA, FGB, BSA-A and BSA-B end office switching in Telephone Company end offices that are not equipped for measurement capabilities, assumed average access minutes will be used to determine the charges. The assumed average access minutes are as set forth in Section 6, Paragraph 6.10.1, D., following.
 - D. Resold Services
 - 1. Scope

Where the Interexchange Carrier is reselling MTS and/or MTS-type service(s) on which the Carrier Common Line and Switched Access charges have been assessed, the Interexchange Carrier may, at the option of the Interexchange Carrier obtain FGA, FGB, FGD, BSA-A, BSA-B, or BSA-D Switched Access Service under this Tariff as set forth in Section 6 for originating and/or terminating access in the local exchange. Such access group and BSA arrangements whether single lines or trunks or multiline hunt groups or trunk groups will have Carrier Common Line Access charges applied as set forth in Paragraph 3.8 in accordance with the resale rate regulations set forth in Paragraph 4. For purposes of administering this provision:

Resold intrastate terminating MTS and MTS-type service(s) shall include collect calls, third number calls and credit card calls where the reseller pays the underlying carrier's service charges; and shall not include interstate minutes of use.

Resold intrastate originating MTS and MTS-type service(s) shall not include collect, third number, credit card or interstate minutes of use.



(AT)

(AT)

(FC)

(AT)

(AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 9 Replacing 1st Revised Sheet 9

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

B. (Continued)

operator-DDD, operator-person, collect, credit-card, third-number and/or other like calls recorded by the IC. The Telephone Company measuring and recording equipment, except as set forth in Paragraph 3.7, C., following, will be associated with end office or local tandem switching equipment and will record each originating and terminating access minute where answer supervision is received. The accumulated access minutes will be summed on a line-by-line basis, by line group or by end office, whichever type of account is used by the Telephone Company, for each IC and then rounded to the nearest minute.

C. When Carrier Common Line Access is provided in association with Feature Groups A and B end office switching in Telephone Company end offices that are not equipped for measurement capabilities, assumed average access minutes will be used to determine the charges. The assumed average access minutes are as set forth in Section 6.7.1, Paragraph D., fo **CANCELLED**

(MT)

- D. Resold Services
 - 1. Scope

APR 11 1993 #9 BY 3rd R

Public Service Commission

Where the Interexchange Carrier is reselling MTS and/or MTS-WASSOURI (CT) service(s) on which the Carrier Common Line and Switched Access charges have been assessed, the Interexchange Carrier may, at the option of the (CT) (CT) Interexchange Carrier obtain Feature Group A, Feature Group B or Feature Group D Switched Access Service under this Tariff as set forth in (RT) Section 6 for originating and/or terminating access in the local exchange. Such access group arrangements whether single lines or trunks or multiline hunt groups or trunk groups will have Carrier Common Line Access charges (RT) applied as set forth in Paragraph 3.8 in accordance with the resale rate regulations set forth in Paragraph 4. For purposes of administering this (RT) provision:

> Resold intrastate terminating MTS and MTS-type service(s) shall include collect calls, third number calls and credit card calls where the reseller pays the underlying carrier's service charges; and shall not include interstate minutes of use.

Resold intrastate originating MTS and MTS-type service(s) shall not include collect, third number, credit card or interstate minutes of

Effective:

(HT)

Issued: NOV 2 1 1991

use.

FEB 1 2 1992

FILED

FEB 1 2 1992 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

RECEIVED

NOV 19 1922

MO. PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access	Services	s Tarif	Ef
	Se	ection	3
lst	Revised	Sheet	9
Replacing (Driginal	${\tt Sheet}$	9

REGEIVED

JUN 2 7 1960

MISSUUR

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

(CP) 3.7 Rate Regulations-(Continued)

B. (Continued)

operator-DDD, operator-person, collect, credit-card, third-number cand/oran other like calls recorded by the IC. The Telephone Company-measuring andrecording equipment, except as set forth in Paragraph 3.7, C., following, will be associated with end office or local tandem switching equipment and will record each originating and terminating access minute where answer supervision is received. The accumulated access minutes will be summed on a line-by-line basis, by line group or by end office, whichever type of account is used by the Telephone Company, for each IC and then rounded to the nearest minute.

C. When Carrier Common Line Access is provided in association with Feature Groups A and B end office switching in Telephone Company end offices that are not equipped for measurement capabilities, assumed average access minutes will be used to determine the charges. The assumed average access minutes are as set forth in Section 6.7.1, Paragraph D., following.

CANCELLED

FEB 12 1992 BY <u>2</u> R.S. 9 Public Service Commission MISSOURI

E.LED JUL 1986 86-84 Public Service Commission Issued: JUN 27 1986 Effective: 1986 JUL

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
 - 3.7 Rate Regulations-(Continued)
 - C. (Continued)
 - 1. When a Switched Access Service is provided where neither the originating access minutes nor the terminating access minutes are measured, the assumed average access minutes are 4076 originating and terminating minutes of use for each Switched Access Service provided.
 - 2. When a Switched Access Service is provided where the terminating access minutes are measured but the originating access minutes are not measured, the assumed average originating access minutes are 2297 originating minutes of use for each Switched Access Service provided.
 - 3. When a Switched Access Service is provided where the originating access minutes are measured but the terminating access minutes are not measured, the assumed average terminating access minutes are 1779 terminating minutes of use for each Switched Access Service provided.

GANGELLED JUL 1 1986 自己官问 LIC SERVICE COMM OF MISSOURI JAN = 1 (92-) 83-253 JAN 0 1 1984 Effective: Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 3 Original Sheet_9_ REGEIVED DEC 20 1003 I.I.SSCURE Public Service Commission

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 3rd Revised Sheet 10 Replacing 2nd Revised Sheet 10

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
 - D. Resold Services-(Continued)
 - 2. Interexchange Carrier Obligations Concerning the Resale of MTS and MTS-type Services

When the Interexchange Carrier is reselling MTS and/or MTS-type service as set forth in Paragraph 1, the Interexchange Carrier will be charged Carrier Common Line Access charges in accordance with the resale rate regulations as set forth in Paragraph 4, if the Interexchange Carrier or the provider of the MTS service furnishes documentation of the MTS usage and/or the Interexchange Carrier furnishes documentation of the MTS-type usage. Such documentation supplied by the Interexchange Carrier shall be supplied each month and shall identify the involved resold MTS and/or MTS-type services.

The monthly period used to determine the minutes of use for resold MTS and/or MTS-type service(s) shall be the most recent monthly period for which the Interexchange Carrier has received a bill for such resold service(s). This information shall be delivered to the Telephone Company, at a location specified by the Telephone Company, no later than 15 days after the bill date shown on the resold MTS and/or MTS-type service bill. If the required information is not received by the Telephone Company, the previously reported information, as described preceding will be used for the next two months. For any subsequent month, no allocation or credit will be made until the required documentation is delivered to the Telephone Company by the Interexchange Carrier.

Issued: November 21, 1991


No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 10 Replacing 1st Revised Sheet 10

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- 3.7 Rate Regulations-(Continued)
- (CP) D. Resold Services
 - 1. Scope

ALCOURI Public Service Commission

RECENED

SEP 2.5 1989

Where the customer is reselling MTS and/or MTS-type service(s) on which the Carrier Common Line and Switched Access charges have been assessed, the customer may, at the option of the customer, obtain Feature Group A, Feature Group B or Feature Group D Switched Access Service under this tariff as set forth in Section 6 following for originating and/or terminating access in the local exchange. Such access group arrangements whether single lines or trunks or multiline hunt groups or trunk groups will have Carrier Common Line Access charges applied as set forth in Paragraph 3.8 following in accordance with the resale rate regulations set forth in Paragraph 4, following. For purposes of administering this provision:

Resold intrastate terminating MTS and MTS-type service(s) shall include collect calls, third number calls and credit card calls where the reseller pays the underlying carrier's service charges; and shallingED include interstate minutes of use.

Resold intrastate originating MTS and MTS-type service(s) shall not include collect, third number, credit card or interstate Minutes of use. Public Service Commission

2. Customer Obligations Concerning the Resale of MTS and MTS-type Services

When the customer is reselling MTS and/or MTS-type service as set forth in Paragraph 1, preceding, the customer will be charged Carrier Common Line Access charges in accordance with the resale rate regulations as set forth in Paragraph 4, following, if the customer or the provider of the MTS service furnishes documentation of the MTS usage and/or the customer furnishes documentation of the MTS-type usage. Such documentation supplied by the customer shall be supplied each month and shall identify the involved resold MTS and/or MTS-type services.

The monthly period used to determine the minutes of use for resold MTS and/or MTS-type service(s) shall be the most recent monthly period for which the customer has received a bill for such resold service(s). This information shall be delivered to the Telephone Company, at a location specified by the Telephone Company, no later than 15 days after the bill date shown on the resold MTS and/or MTS-type service bill. If the required information is not received by the Telephone Company, the previously reported information, as described preceding, will be used for the next two months. For any subsequent month, no allocation or credit will be made until the required documentation is delivered to the Telephone Company by the customer.

Bffective:

Issued: SEP 2 5 1989

(CP)

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

1989

JCT 1

Access Services Tariff Section 3 1st Revised Sheet 10 Replacing_Original_Sheet 10 米巳頃ミルミリ

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- JUN 2 7 1986

(CP) 3.7 Rate Regulations-(Continued)

MISSUURI

1986

84

ision

D. When the customer orders a combined access group to be weed in Conjunction tion with the resale of MTS/WATS and/or MTS/WATS-type services-as-set forth in Paragraph 3.3, D., preceding, subject to the limitations of Carrier Common Line as set forth in Paragraph 3.2, preceding, and the Telephone Company receives the usage information required to calculate the proration of Carrier Common Line as set forth in Paragraph 3.4, F., preceding, the customer will be billed as set forth in Paragraph 1., following. When more than one combined access group is provided in a LATA, the Telephone Company will apportion the resold MTS/WATS and/or MTS/WATS-type services and originating minutes of use among the combined access groups. Such apportionment will be based on the relationship of the originating usage for each combined access group to the total originating usage for all combined access groups in the LATA. The involved resold minutes shall be only intrastate outward MTS, MTS-type, WATS and WATS-type minutes and shall not include collect, third number, credit card or interstate minutes of use.

In order for the rate regulations to apply as set forth in Paragraph 1, following, the combined access group and the resold MTS/WATS and/or MTS/WATS-type services must be provided in the same state in the same LATA, provided by the same Telephone Company and connected directly or indirectly. For those LATAs that have more than one state, the customer shall report the information by state within the LATA.

Each of the combined access groups arranged in a multiline hunt group or trunk group arrangement used by the customer in association with the resold MTS/WATS and/or MTS/WATS-type services must be connected either directly or indirectly to the customers' designated premises at which the resold MTS/WATS and/or MTS/WATS-type services are terminated. Direct connections are those arrangements where the combined access groups and resold MTS/WATS and/or MTS/WATS-type services are terminated at the same customers' designated premises. Indirect connections are those arrangements where the combined access groups and resold MTS/WATS and/or MTS/WATS-type services are terminated at different customer designated premises in the same LATA. Such different customer designated premises are connected by ____ facilities that permit a call to flow from combined Faccess groups' to ED resold MTS/WATS and/or MTS/WATS-type services

Issued: JUN 27 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St St. Louis, Missouri

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
- D. When the IC orders Switched Service Access (line side and/or trunk side) as set forth in Section 6, Paragraph 6.7.3, following, and reports the data as set forth in Paragraph 3.4, F., preceding, and in Section 6, Paragraph 6.7.3, following, the IC monthly transitional charge access minutes will be adjusted as follows:
 - 1. For the first month, both the Switched Access Service and the resold Dedicated Access Line Service provided under this Tariff are in service in the same state in the same LATA and provided by the same Telephone Company, the Carrier Common Line Access originating access minutes for the IC LATA account or end office account involved, whichever type of account the Telephone Company uses to bill the IC, will be adjusted as follows. After the Telephone Company verifies that the number of Dedicated Access Line Services in service as reported by the IC is accurate, the Carrier Common Line Access originating access minutes developed for the IC account involved will be reduced by an amount of minutes equal to the number of resold Dedicated Access Line Services reported as set forth in Paragraph 3.4, F., preceding, for trunk side Switched Access Service and Section 6, Paragraph 6.7.3, following, for line side Switched Access Service times 2,500. The adjustment will be made to the involved IC account no later than either the next bill date, or the one subsequent to that, depending on when the report is received.
 - 2. For every monthly period after the first monthly period, both the Switched Access Service and the resold Dedicated Access Line Service provided under this Tariff are in service in the same state in the same LATA and provided by the same Telephone Company, the Carrier Common Line Access originating access minutes for the IC LATA account or End Office account involved, whichever type of account the Telephone Company uses to bill the IC, will be adjusted as follows. After the Telephone Company verifies that the number of Dedicated Access Line Services with 2,500 or more minutes per month reported by the IC is accurate the Carrier Common Line Access originating accessmenties reveloped for the IC account

JUL 1 1986

PUBLIC SERVICE COMMISSION

NR. S. # 10

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff DEUS Section 3 Original Sheet 10 DEC 2 S 10ED **LUSSOUR** Public Service Commission

- いい二1 他下:

2 83-258

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 3rd Revised Sheet 10.01 Replacing 2nd Revised Sheet 10.01

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE (Continued)

- 3.7 Rate Regulations-(Continued)
- D. Resold Services-(Continued)
- 3. Resale Documentation Provided By the Customer

When the customer utilizes Switched Access Service as set forth in Paragraph 2, preceding, the Telephone Company may request a certified copy of the customer's resold MTS or MTS-type usage billing from either the customer or the provider of the MTS and/or MTS-type Service. Requests for billing will relate back no more than 12 months prior to the current billing period.

4. Regulations Concerning the Resale of MTS and MTS-type Services

When the customer is provided an access group or BSA to be used in conjunction with the resale of MTS and/or MTS-type services as set forth in Paragraph 1, preceding, subject to the limitations as set forth in Paragraph 3.2, preceding, and the Telephone Company receives the usage information required as set forth in Paragraph 2, preceding, to calculate the adjustment of Carrier Common Line Access charges, the customer will be billed as set forth following. In no event shall the adjusted access minutes be less than zero.

a. Apportionment and Adjustment of Resold Minutes of Use

When the customer is provided with more than one access group or BSA in a LATA in association with the resale of MTS and/or MTS-type services, the resold minutes of use will be apportioned as follows:

1. Originating Services

The Telephone Company will apportion the resold originating MTS and/or MTS-type services and originating minutes of use for which the resale credit adjustment applies, among the access groups and BSAs. Such apportionment will be based on the relationship of the originating usage for each access group and BSA to the total originating usage for all access groups and BSAs in the LATA. For purposes of administering this provision:

Resold originating MTS and/or MTS-type services minutes shall be only those attributable to intrastate originating MTS and/or MTS-type minutes and shall not include collect, third number, credit card or interstate minutes of use.



(AT) (AT)



AUG 15 1995

MO. PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 2nd Revised Sheet 10.01 Replacing 1st Revised Sheet 10.01

RECEIVED

MAR 291993

MISSOURI Public Service Commission

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- 3.7 Rate Regulations-(Continued)
- D. Resold Services-(Continued)
 - 3. Resale Documentation Provided By the Customer

When the customer utilizes Switched Access Service as set forth in Paragraph 2, preceding, the Telephone Company may request a certified copy of the customer's resold MTS or MTS-type usage billing from either the customer or the provider of the MTS and/or MTS-type Service. Requests for billing will relate back no more than 12 months prior to the current billing period.

- 4. Regulations Concerning the Resale of MTS and MTS-type Services
- (AT) When the customer is provided an access group or BSA to be used in conjunction with the resale of MTS and/or MTS-type services as set forth in Paragraph 1, preceding, subject to the limitations as set forth in Paragraph 3.2, preceding, and the Telephone Company receives the usage information required as set forth in Paragraph 2, preceding, to calculate the adjustment of Carrier Common Line Access charges, the CANDER will be billed as set forth following.
 - a. Apportionment and Adjustment of Resold Minutes of Use $SEP_{1} = 51995$

When the customer is provided with more than one access group or BSA in a LATA in association with the resale of MTS and or MTS in the Commission services, the resold minutes of use will be apport public as a solution of the services.

1. Originating Services

The Telephone Company will apportion the resold originating MTS and/or MTS-type services and originating minutes of use for which the resale credit adjustment applies, among the access groups and BSAs. Such apportionment will be based on the relationship of the originating usage for each access group and BSA to the total originating usage for all access groups and BSAs in the LATA. For purposes of administering this provision:

Resold originating MTS and/or MTS-type services minutes shall be only those attributable to intrastate originating MTS and/or MTS-type minutes and shall not include collect, third number, credit card or interstate minutes of use.

FILED

APR 1 1 1993

APR 1 1 1995

Issued: MAR 2 6 1993

Effective:

92-304 APF MO. PUBLIC SERVICE COMM.

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

(AT)

(AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 1st Revised Sheet 10.01 Replacing Original sheet 10.01

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

D. Resold Services-(Continued)

Macount Public Service Commission

CANCELLED

SEP 2 5 1989

RECEIVED

3. Resale Documentation Provided By the Customer

When the customer utilizes Switched Access Service as set forth in Paragraph 2, preceding, the Telephone Company may request a certified copy of the customer's resold MTS or MTS-type usage billing from either the customer or the provider of the MTS and/or MTS-type Service. Requests for billing will relate back no more than 12 months prior to the current billing period.

4. Regulations Concerning the Resale of MTS and MTS-type Services

When the customer is provided an access group to be used in conjunction with the resale of MTS and/or MTS-type services as set forth in Paragraph 1, preceding, subject to the limitations as set forth in Paragraph 3.2, preceding, and the Telephone Company receives the usage information required as set forth in Paragraph 2, preceding, to calculate the adjustment of Carrier Common Line Access charges, the customer will be billed as set forth following.

Apportionment and Adjustment of Resold Minutes of Use а.

> When the customer is provided with more than one accessAgroup 1993 LATA in association with the resale of MTS and/or MTS type AR STOO services, the resold minutes of use will be apportioned as follows: MISSOURI

1. Originating Services

The Telephone Company vill apportion the resold originating MTS and/or MTS-type services and originating minutes of use for which the resale credit adjustment applies, among the access groups. Such apportionment will be based on the relationship of the originating usage for each access group to the total originating usage for all access groups in the LATA. For purposes of administering this provision:

Resold originating MTS and/or MTS-type services minutes shall be only those attributable to intrastate originating MTS and/or MTS-type minutes and shall not include collect, third number, credit card or interstate minutes of use. FILED

0CT 1

1389

Issued: SEP 2 5 1989

OCT 1_ 1989 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company r'ublic Service Commission St. Louis, Missouri

Effective:

(CP)

(CP)

Access Services Tariff Section 3 Original Sheet 10.01

REGEI

JUN 2 7 1950

MISSUUR

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- (CP) 3.7 Rate Regulations-(Continued)
 - D. (Continued)

The adjustments as set forth following will be computed; separately for Olleach combined access group.

1. Combined Access Groups

When combined access groups are provided in a LATA, the following regulations apply:

- a. One intrastate MTS/WATS-type Feature Group A Switched Access Service for each resold intrastate WATS and WATS-type service reported as set forth in Paragraph 3.4, F., preceding, and Paragraphs d. and e., following, will be billed local business exchange service rates as set forth in Paragraph 6.7.3, B.1.a., following.
- b. For MTS/WATS-type Feature Group A provided in a combined access group, the minutes billed will be all the terminating intrastate access minutes and any adjusted originating intrastate access minutes for such combined access groups. The adjusted originating minutes billed will be the originating intrastate access minutes less the reported resold service(s) originating minutes of use.

		ARRON, President-Missouri Divis estern Bell Telephone Company St. Louis, Missouri	ion
Iss	sued: JUN 27 1986	Effective: JUL	Pupile Service Commission
		OCT 1 1989 BY MR.S. #10.01 Public Service Commission MISSOUHI	36-84
		CANCELLED	

Access Services Tariff Section 3 3rd Revised Sheet 11 Replacing 2nd Revised Sheet 11

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
- D. Resold Services-(Continued)
 - 4. Regulations Concerning the Resale of MTS and MTS-type Services-(Continued)
 - a. Apportionment and Adjustment of Resold Minutes of Use-(Continued)
 - 1. Originating Services-(Continued)

The resale credit adjustment shall apply for resold originating MTS and MTS-type services and minutes of use, provided Carrier Common Line and Switched Access Charges have been assessed on such services.

2. Terminating Services

The Telephone Company will apportion the resold terminating MTS and/or MTS-type services and terminating minutes of use for which the resale credit adjustment applies, among the access groups and

BSAs. Such apportionment will be based on the relationship of the terminating usage for each access group or BSA to the total terminating usage for all access groups and BSAs in the LATA. For purposes of administering this provision:

Resold terminating MTS and/or MTS-type services minutes shall be only those attributable to intrastate terminating MTS/MTS-type (i.e., collect calls, third number calls, and credit card calls) and shall not include interstate minutes of use or MTS/MTS-type minutes of use paid for by another party.

The resale credit adjustment shall apply for resold terminating MTS and MTS-type services and minutes of use, provided Carrier Common Line and Switched Access Charges have been assessed on such services.

b. Same State/Telephone Company/LATA Limitation

In order for the rate regulations to apply as set forth in Paragraph d, following, the access groups and BSAs and the resold MTS and/or MTS-type services must be provided in the same state in the same LATA, provided by the same Telephone Company and connected directly or indirectly. For those LATA's that encompass more than one state, the customer shall report the information by state within the LATA.

Issued:	March 26, 1993	Effective:	April 11, 1993	
	•	. BARRON, Presiden uthwestern Bell Telep		FIL
		St. Louis, Mis	1 2	MO F

(AT)

(AT)

(AT)

No supplement to this tariff vill be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 11 Replacing 1st Revised Sheet 11

ACCESS SERVICES

RECEIVED

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

SEP 2 5 1989

3.7 Rate Regulations-(Continued)

MEDOUTH Public Service Commission

- D. Resold Services-(Continued) (CP)
 - 4. Regulations Concerning the Resale of MTS and MTS-type Services-(Continued)
 - a. Apportionment and Adjustment of Resold Minutes of Use-(Continued)
 - 1. Originating Services-(Continued)

The resale credit adjustment shall apply for resold originating MTS and MTS-type services and minutes of user and Gred Gredier Common Line and Switched Access Charges have been assessed on such APR 11 1993 BY 3 A. R. S. # 11 services.

2. Terminating Services

The Telephone Company will apportion the Unic Commission and/or MTS-type services and terminating minutes SS use for which the resale credit adjustment applies, among the access groups. Such apportionment will be based on the relationship of the terminating usage for each access group to the total terminating usage for all access groups in the LATA. For purposes of administering this provision:

Resold terminating MTS and/or MTS-type services minutes shall be only those attributable to intrastate terminating MTS/MTS-type (i.e., collect calls, third number calls, and credit card calls) and shall not include interstate minutes of use or MTS/MTS-type minutes of use paid for by another party.

The resale credit adjustment shall apply for resold terminating MTS and MTS-type services and minutes of use, provided Carrier Common Line and Switched Access Charges have been assessed on such services.

b. Same State/Telephone Company/LATA Limitation

In order for the rate regulations to apply as set forth in Paragraph d, following, the access groups and the resold MTS and/or MTS-type services must be provided in the same state in the same LATA, provided by the same Telephone Company and connected directly or indirectly. For those LATA's that encompass more than one state, the customer shall report the information by state within the LATA.

FILED

Issued: SEP 2 5 1989

(CP)

Effective: OCT 1 1989)CT 1 1989 89-14 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commissio St. Louis. Missouri

Access Services Tariff Section 3 1st Revised Sheet 11 Replacing Original Sheet 11

ACCESS SERVICES

- CARRIER COMMON LINE ACCESS SERVICE-(Continued) CANCEL HEBBEIVED 3.
- 3.7 Rate Regulations-(Continued)
 - D. (Continued)
 - 1. Combined Access Groups-(Continued)
- Combined Access Groups-(Continued) BY 200 Commission C. Any trunk side Switched Access Service Commission MISSUUX Switched Access Service charges as set forth in Paragraph 6.7.3; B.1.b., following. Carrier Common Line charges will be billed on all the terminating intrastate access minutes and any adjusted originating minutes (i.e., originating minutes less the reported resold service(s) minutes of use).

OCT 1 1989

BY 2 BESHINUN 27 1980

- d. For the first month a combined access group is in service, each associated resold WATS and WATS-type service will be assumed to have 2500 or more originating intrastate minutes of use.
- e. After the first month that both the combined access group and the associated resold WATS and WATS-type services are in service, each resold WATS and WATS-type service shall have 2500 or more originating intrastate minutes of use per month on average. If the average originating intrastate usage per service for each resold WATS or WATS-type service group does not equal 2500 originating minutes of use in any month, then for combined access billing purposes, the number of WATS and WATS-type services for which local business exchange service rates will be applicable will be reduced until the average originating intrastate usage per month for the remaining services equals or exceeds 2500 originating intrastate minutes. The number of originating minutes per month per resold WATS and WATS-type service will be developed by adding the resold outward MTS and MTS-type minutes to the resold originating WATS and WATS-type minutes and dividing the resulting sum per service group by the number of services in the group.
- 2. The adjustment as set forth in Paragraph 1., preceding, will be made to the involved customer account no later than either the next bill date, or the one subsequent to that, depending on when the usage report is obtained.

Effective:

3. When the resold service(s) usage is shown in hours, the number WATS-type and WATS minutes of use. If the resold service(s) usage is shown in a unit that does not show hours or minutes, 1 the customer shall provide a factor to convert the shown units 1 1986 86-84 to minutes. Public Sentice Commission

1 1986

JUL

Issued: JUN 27 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

(Continued) D.

2. (Continued).

number of resold Dedicated Access Line Services reported as set forth in Paragraph 3.4, F., preceding, for trunk side Switched Access Service and Section 6, Paragraph 6.7.3, following, for line side Switched Access Service times 2,500. The adjustment will be made to the involved IC account no later than either the next bill date, or the one subsequent to that, depending on when the report is received.

3. The originating access minutes for Carrier Common Line Access, adjusted as set forth in Paragraphs 3.7, D.1. or 2., preceding, whichever is appropriate, or any other section of this Tariff, that are billed to an IC in a monthly period shall not be less than zero.

If the Telephone Company determines the number of Dedicated Access Line Services with 2,500 or more access minutes per month reported by the IC for the adjustment specified in Paragraphs 3.7, D.1. or 2., preceding, is not accurate, no adjustment will be made until a new report properly stating the number of Dedicated Access Line Services with 2,500 or more access minutes per month is delivered to the Telephone by the IC.

- 4. The adjustment as set forth in Paragraphs 3.7, D.1., 2. and 3., preceding, will be made to the involved IC account after making the adjustments to the IC account as set forth in Paragraphs 3.7, E. and F., following.
- E. When the IC reports exchange minutes of use, as long as the exchange minutes are not included in the minutes used to develop the reports as set forth in Section 2, Paragraph 2.3.14, preceding, as set forth in Paragraph 3.4, G., preceding, for in-service Switched Access Service in a LATA, the Carrier Common Line Access access of the IC LATA account or end office account involved, which ever type of account the Telephone Company uses to bill the IC, will be adjusted as follows. After the Telephone Company verifies the exchange minutes reported by the IC are accurate, the Carrier

PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

83 - 253

a Public Service Computation

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

	Poriginal Sheet]11	1
ļ	DEC 2 9 (866	N N
	LASSOURI Public Service Commission	, 1 :

Access Services Tariff

Access Services Tariff Section 3 3rd Revised Sheet 12 Replacing 2nd Revised Sheet 12

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
- D. Resold Services-(Continued)
 - 4. Regulations Concerning the Resale of MTS and MTS-type Services-(Continued)
 - c. Direct and Indirect Connections
- (AT) Each of the access group or BSA arrangements used by the customer in association with the resold MTS and/or MTS-type services must be connected either directly or indirectly to the customer designated premises at which the resold MTS and/or MTS-type services are terminated. Direct connections are those arrangements where the
 (AT) access groups or BSAs and resold MTS and/or MTS-type services are terminated at the same customer designated premises.
 - Indirect originating connections are those arrangements where the
 - access groups or BSAs and the resold originating MTS and/or MTS-type services are physically located at different customer designated premises in the same LATA. Such different customer designated premises are connected by facilities that permit a call to flow from access groups or BSAs to relsold MTS and/or MTS-type services.
 - Indirect terminating connections are those arrangements where the access groups or BSAs and resold terminating MTS and/or MTS-type services are physically located at different customer designated premises in the same LATA. Such different customer designated premises are connected by facilities that permit a call to flow from resold terminating MTS and/or MTS-type services to access groups or BSAs.
 - d. Access Groups and BSAs

The minutes billed Carrier Common Line Access Service charges will be the adjusted originating intrastate access minutes and the adjusted terminating intrastate access minutes for the appropriate access groups and BSAs.

The adjusted originating access minutes will be the originating intrastate access minutes less the reported resold originating MTS and/or MTS-type service minutes of use as set forth in Paragraph a., 1, preceding; but not less than zero. The adjusted terminating access minutes will be the terminating intrastate access minutes less the reported resold terminating MTS and/or MTS-type service minutes of use as set forth in Paragraph b., 1, preceding; but not less than zero.

Issued:



(AT)

(AT)

(AT)

(AT)

(AT)

(AT)

(AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 12 Replacing 1st Revised Sheet 12

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued) 3.7 Rate Regulations-(Continued)

SEP 2.5 1989 N ISSOURI

RECEIVED

(CP) D. Resold Services-(Continued)

Public Service Commission 4. Regulations Concerning the Resale of MTS and MTS-type Services-(Continued)

c. Direct and Indirect Connections

Bach of the access group arrangements used by the customer in association with the resold MTS and/or MTS-type services must be connected either directly or indirectly to the customer designated premises at which the resold MTS and/or MTS-type services are terminated. Direct connections are those arrangements where the access groups and resold MTS and/or MTS-type services are terminated at the same customer designated premises.

Indirect originating connections are those arrangements where the access groups and the resold originating MTS and/or MTS-type services are physically located at different customer designated premises in the same LATA. Such different customer designated premises are connected by facilities that permit a call to flow from access groups to resold MTS and/or MTS-type services.

Indirect terminating connections are those arrangements where the access groups and resold terminating MTS and/or MTS-type services are physically located at different customer designated premises in the same LATA. Such different customer designated premises are connected by facilities that permit a call to flow from resold CANCERLED and/or MTS-type services to access groups.

d. Access Groups

APR 11 1993

OCT 1 1989 89-14

The minutes billed Carrier Common Line Access Service Charges will the adjusted originating intrastate access minutes and the adjusted terminating intrastate access minutes and the adjusted mission groups.

The adjusted originating access minutes will be the originating intrastate access minutes less the reported resold originating MTS and/or MTS-type service minutes of use as set forth in Paragraph a., 1, preceding; but not less than zero. The adjusted terminating access minutes will be the terminating intrastate access minutes less the reported resold terminating MTS and/or MTS-type service minutes of use as set forth in Paragraph b., 1, preceding; but not less than zero.

Effective:

OCT 1

Issued: SEP 2 5 1989

(CP)

Figesc Service Commissio. By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 3 lst Revised Sheet 12 Replacing Original Sheet 12

REGEIVED

JUN 2 7 1985

ACCESS SERVICES

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- (CP) 3.7 Rate Regulations-(Continued)
 - D. (Continued)
 - 4. The originating intrastate access minutes for Carrier Common Line Access, adjusted as set forth in Paragraph 1., preceding, orlany Sich other section of this tariff, that are billed to a customer in a monthly period shall not be less than zero.
 - 5. If the required documentation is not received by the Telephone Company, the previously reported information as described in Paragraph 3.4., F., preceding, will be used for the next two months. For any subsequent month, no adjustment or credit will be made until the required documentation is delivered to the Telephone Company by the customer.
 - 6. The adjustment as set forth in Paragraphs 1. and 2., preceding, will be made to the involved customer's account after making the adjustments to the customer's account as set forth in Paragraph E., following.
 - E. When the customer reports interstate and intrastate use of in service Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Paragraph 2.3.12. The intrastate Switched Access Service access minutes will, after adjustment as set forth in Paragraph D., preceding, when necessary, be used to determine the Carrier Common Line Charges as set forth in Paragraph F., following.

ued: JUN 27 1986 Effective: JUL 1 19	86
· · · · · · · · · · · · · · · · · · ·	Public Senice Commissio
	86-84
Public Semisson	
Public Service Commission	FILED
OCT 1 St#12 By 2 m S. H12 Commission	
CANCELLED	
CANT	

St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

Ε. (Continued)

> Common Line Access access minutes developed by the Telephone Company for the IC account involved will be reduced by the exchange minutes of use reported by the IC. If the Telephone Company determines the reported exchange minutes are not accurate, no adjustment will be made until a new report properly stating the exchange minutes is delivered to the Telephone Company by the IC. The access minutes, adjusted as set forth in this paragraph or any other section of this Tariff, that are billed to the IC in a monthly period shall not be less than zero.

This adjustment will be made to the involved IC account prior to making the adjustments to the IC account as set forth in Paragraphs 3.7, D., preceding, and Paragraph 3.7, F., following.

- F. When the IC reports interstate and intrastate use of in-service Switched Access Service, the Carrier Common Line Access access minutes will be adjusted as follows. The Carrier Common Line Access access minutes developed by the Telephone Company will, after adjustment as set forth in Paragraph 3.7, E., preceding, when necessary, be multiplied by the intrastate percentage reported by the IC as set forth in Section 2. Paragraph 2. Paragraph 3. Carrier Comm
- G. After the ad preceding. h Access acces determined a
 - 1. The access minutes for a Premium Access Carrier will be multiplied by the transitional charge as set forth in

7, D., preceding, when	, after adjustment as set forth in necessary, be used to determine the
on Line Charges as set	forth in Paragraph 3.7, G., following.
ave been applied, when	in Paragraphs 3.7, D., E. and F., in necessary, to the Carrier Common Line is for the involved IC account will be

Access Services Tariff

Original_Sheet 12

REGEIVED

DEC 20 12.3

ETESOURI

Public Service Commission

Section 3

Paragraph 3.8, following, to determine the charges. 上記[1] JUL 1 1986 JAN - 1 (93) BY 1 At R.S. # 12 PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 8th Revised Sheet 13 Replacing 7th Revised Sheet 13

Missouri Public Service Commission

RECT OCT 1 5 2002

ACCESS SERVICE

3. CARRIER COMMON LINE ACCESS SERVICE (cont'd)

- 3.7 Rate Regulations (cont'd)
 - E. When the customer reports interstate and intrastate use of in service Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section 2, paragraph 2.3.12.

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

	Current (1) Rate
InterLATA Access	
Access Minute, each terminating	\$0.0179919(CR)
Access Minute, each originating	0.0099222(CR)
IntraLATA Access	
Access Minute, each terminating	0.0179919(CR)
Access Minute, each originating	0.0099222(CR)

Missouri Public Service Commission

FLED DEC 01 2002

(1) See Sheet 14 for maximum allowable rates.

Issued: October 15, 2002

Effective: December 1, 2002

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri





Access Services Tariff Section 3 7th Revised Sheet 13 Replacing 6th Revised Sheet 13

ACCESS SERVICES TARIFF

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
 - E. When the customer reports interstate and intrastate use of in service Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section 2, Paragraph 2.3.12.
- 3.8 Rates and Charges

The rates for Carrier Common Line Access are:

Rate InterLATA Access Access Minute, each terminating Access Minute, each originating IntraLATA Access Access Minute, each terminating \$0.0178(CR) 0.0098(CR)

CANCELLED

DEC 0 1 2002 minssign

(1) See Sheet 14 for maximum allowable rates.

Issued: October 1, 2001

Effective: December 1, 2001

By JAN NEWTON., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Missouri Public

FILED DEC 01 2001

Service Commission

Missouri Public

REC'D OCT 01 2001

Service Commission

Current (1) \$0.0178315(CR) 0.0098337(CR)

Access Minute, each originating

P.S.C. Mo.-No. 36

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 3 except for the purpose 6th Revised Sheet 13 of canceling this tariff. Replacing 5th Revised Sheet 13 Missouri Public Service Commission ACCESS SERVICES 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued) REC'D OCT 0 2 2000 3.7 Rate Regulations-(Continued) E. When the customer reports interstate and intrastate use of in service Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section (RT)2, Paragraph 2.3.12. (\mathbf{RT}) 3.8 Rates and Charges The rates for Carrier Common Line Access are: (AT)Current (1) (AT)Rate InterLATA Access Access Minute, each terminating \$0.0179662(CR) Access Minute, each originating 0.0099080(CR) IntraLATA Access Access Minute, each terminating \$0.0179662(CR) Access Minute, each originating 0.0099080(CR)



Missouri Public Service Commission FILED DEC 01 2000

(AT) (1) See Sheet 14 for maximum allowable rates.

Issued: October 2, 2000

Effective: December 1, 2000

By JAN NEWTON., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

i

ł

Access Services Tariff Section 3 5th Revised Sheet 13 Replacing 4th Revised Sheet 13 **RECEIVED**

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

forth in Paragraph F, following.

- SEP 22 1994
- E. When the customer reports interstate and intrastate user pinises viewer common Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section 2, Paragraph 2.3.12. The intrastate Switched Access Service access minutes will, after adjustments as set forth in Paragraph D, preceding, when necessary, be used to determine the Carrier Common Line Charges as set

(RT)	
	CANCELLED
	UNIVELLED.
	DEC 01 2000 BY 642 RS 13 Public Fervice Commission WildSouth
(RT)	
	3.8 Rates and Charges
	The rates for Carrier Common Line Access are:
	Rate
(RT)	
	InterLATA Access
	Access Minute, each terminating (CR)\$9.018113
	Access Minute, each originating (CR) 0.010000
	IntraLATA Access 95 - 85
	MISSOURI Access Minute, each terminating
-	Access Minute, each originating (CR) 0.010000
	SSUEd: SEP 2 2 1994 Effective:
	By HORACE WILKINS, JR., President-Missouri DCT 0 1 1994 Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 5th Revised Sheet 13 Replacing 4th Revised Sheet 13

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

Rate Regulations-(Continued) 3.7

DFC 23 1993 MISSOURI

Cubile Service Commission E. When the customer reports interstate and intrastate use of in service Switched Access Service, the Carrier Common Line Access Transitional Charge will be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section 2, Paragraph 2.3.12. The intrastate Switched Access Service access minutes will, after adjustments as set forth in Paragraph D, preceding, when necessary, be used to determine the Carrier Common Line Charges as set forth in Paragraph F, following.

(RT)

(RT)

3.8 Rates and Charges

The rates for Carrier Common Line Access NCELLEL

Rate (RT) InterLATA Access Access Minute, each terminating S (CR)\$0.0148 (CR) 0,0100-3 Access Minute, each originating . IntraLATA Access Access Minute, each terminating . . (CR)\$0 Crrblic S Commission Access Minute, each originating . . . Issued: DEC 2 3 1993 Effective: JAN 0 1 1994

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 4th Revised Sheet 13 Replacing 3rd Revised Sheet 13

ACCESS SERVICES

RECEIVED

SEP 25 1989

CANCELLED

OCT 0 1 1994

Rate

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

- 3.7 Rate Regulations-(Continued)
 - When the customer reports interstate and intrastate use of in service E. Switched Access Service, the Carrier Common Line Access [Transitional Charge) vill be billed only to intrastate Switched Access Service access minutes based on the data reported by the customer as set forth in Section 2,
- Paragraph 2.3.12. The intrastate Switched Access Service access minutes (AT) vill, after adjustment as set forth in Paragraph D, preceding, when necessary, be used to determine the Carrier Common Line Charges as set forth in Paragraph F, following.
 - F. After the adjustments as set forth in Paragraphs 3.7, D., and B., preceding, have been applied, when necessary, to the Carrier Common Line Access access minutes, the charges for the involved IC account will be determined as follows:
 - 1. The originating and terminating access minutes for a Premium Access Carrier will be multiplied by the transitional charges as set forth in Paragraph 3.8, following, to determine the charges.
 - 2. The originating and terminating access minutes for a non-Premium Access Carrier will be multiplied by the transitional charges in Paragraph 3.8, following and by:

0.65 for access minutes in 1984, 0.77 for access minutes in 1985, 0.88 for access minutes from January 1, BY Sth R. S. 13 Public Service Commission 1986, through August 31, 1986.

to determine the charges.

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

Transitional Charge

InterLATA Access

Access Minute, each terminating (CR)\$0.0192 Access Minute, each originating (CR) 0.0111 FILED IntraLATA Access

Access Minute, each terminating \$0.0192 8 9 - 14 Access Minute, each originating Public Service Commission

Issued: SEP 2 5 1989 Bffective: UUI I 1989 By R. D. BARRON, President-Missouri Division

Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 3 3rd Revised Sheet 13 Replacing 2nd Revised Sheet 13

MAY 2

ACCESS SERVICES

RECEIVED

1988

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.7 Rate Regulations-(Continued)

MISSOURI

- F. After the adjustments as set forth in Paragraphs 3.7, D., and E., preceding, have been applied, when necessary, to the Carrier Common Line Access access minutes, the charges for the involved IC account will be determined as follows:
 - 1. The originating and terminating access minutes for a Premium Access Carrier will be multiplied by the transitional charges as set forth in Paragraph 3.8, following, to determine the charges.
 - 2. The originating and terminating access minutes for a non-Premium Access Carrier will be multiplied by the transitional charges in Paragraph 3.8, following and by:

0.65 for access minutes in 1984,
0.77 for access minutes in 1985,
0.88 for access minutes from January 1, 1986, through August 31, 1986.

to determine the charges.

3.8 Rates and Charges

The rates for Carrier Common Line Access are: ED

		Transitional Charge	OCT 1 1989 BY 4 13 Rate Public Service Commission MISSOUR: 50 0463	
	(AT)	InterLATA Access	Public Service Comments MISSOUR	
		Access Minute, each	terminating \$0.0463	
)		Access Minute, each	originating 0.0266	
	(TA)	IntraLATA Access		
	(NR)	Access Minute, each	terminating \$0.0192	
	(NR)	Access Minute, each	originating 0.0111	

Issued: MAY 2 1988 By R. D. BARRON, President-Missouri Division JUL 1 1988 Southwestern Bell Telephone Company

St. Louis, Missouri

94-222 stal. Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

3.7 Rate Regulations-(Continued)

3.

2nd_Revised_Sheet_13. Replacing ACCESS SERVICES OCT 3 1 1986 CARRIER COMMON LINE ACCESS SERVICE-(Continued) MISSUNKI

Rate

Access Services Tariff

Section 3

- F. After the adjustments as set forth in Paragraphs 3.7, Dublic Service Commission preceding, have been applied, when necessary, to the Carrier Common Line Access access minutes, the charges for the involved IC account will be determined as follows:
 - The originating and terminating access minutes for a Premium Access 1. Carrier will be multiplied by the transitional charges as set forth in Paragraph 3.8, following, to determine the charges.
 - The originating and terminating access minutes for a non-Premium 2. Access Carrier will be multiplied by the transitional charges in Paragraph 3.8, following and by:
 - 0.65 for access minutes in 1984, 0.77 for access minutes in 1985, 0.88 for access minutes from January 1, 1986, through August 31, 1986.

to determine the charges.

Transitional Charge

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

CANCELLED JUL 1 1988 Public Service Commission

Access Minute, each terminating \$0.0463 0.0266 Access Minute, each originating

JAN - 1 1987 Public Service Commission



(CR)

CR)

Effective: JAN 01 1987

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 1st Revised Sheet 13 Replacing Original Sheet 13

ACCESS SERVICES

REGEIVED

- 3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)
- (CP) 3.7 Rate Regulations-(Continued)

JUN 2 7 1985

- F. After the adjustments as set forth in Paragraphs 3.7, D., and E., preceding, have been applied, when necessary, to the Carrier Common Line Access access minutes, the charges for the involved IC account will be determined as follows:
 - 1. The originating and terminating access minutes for a Premium Access Carrier will be multiplied by the transitional charges as set forth in Paragraph 3.8, following, to determine the charges.
 - 2. The originating and terminating access minutes for a non-Premium Access Carrier will multiplied by the transitional charges in Paragraph 3.8, following and by:
 - 0.65 for access minutes in 1984,
 0.77 for access minutes in 1985,
 0.88 for access minutes from January 1, 1986, through August 31, 1986.

to determine the charges.

Transitional Charge

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

Access Minute, each terminating

Access Minute, each originating

RAR BELLED

JAN 1 1987

\$0,0521,21d R.S. #13 PUBLIC SERVICE COMMISSION







Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose	Access Services Tariff Section 3 Original Sheet 13
of canceling this tariff.	ACCESS SERVICES REGENCED
3. CARRIER COMMON LINE ACCESS	SERVICE-(Continued) DEC 20 (SS3
3.7 Rate Regulations-(Continue	ed) / Elissouri
G. (Continued)	Public Service Commission
	r a non-Premium Access Carrier will be nsitional charge in Paragraph 3.8, fol-
0.77 for ac 0.88 for ac	fess minutes in 1984, fess minutes in 1985, fess minutes from January 1, thru August 31, 1986
to determine the charg	ges.
3.8 Rates and Charges	
The rate for Carrier Comm	on Line Access is:
	Rate
Transitional Charge	2
Access Minute, each	n

GARBELLED JUL 1 1986 行礼民间 JAN - 1 198.) PUBLIC SERVICE COMMISSION 83-253 BY. of Missouri ing Chemicission ៍ខ្ម

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 3 2nd Revised Sheet 14 Replacing 1st Revised Sheet 14

ACCESS SERVICE

3. CARRIER COMMON LINE ACCESS SERVICE (cont'd)

3.8 Rates and Charges (cont'd)

The rates for Carrier Common Line Access are:

InterLATA Access

Missouri Public Service Commission

RECTO OOT 1 5 2002

Maximum Rate

Access Minute, each terminating\$0.0179919(CR)Access Minute, each originating0.0099222(CR)IntraLATA Access0.0179919(CR)Access Minute, each terminating0.0179919(CR)Access Minute, each originating0.0099222(CR)

Missouri Public Service Commission

FILED DEC 01 2002

يرييها المتوجم مرمانه

Issued: October 15, 2002

Effective: December 1, 2002

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

ł

ļ

1

Access Services Tariff Section 3 1st Revised Sheet 14 Replacing Original Sheet 14

ACCESS SERVICES TARIFF

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

	Maximum Rate
InterLATA Access	
Access Minute, each terminating	\$0.0178315 (CR)
Access Minute, each originating	0.0098337 (CR)
IntraLATA Access	
Access Minute, each terminating	\$0.0178 (CR)
Access Minute, each originating	0.0098 (CR)

CANCELLED

DEC 0 1 2002 2 Jul RS 14 Public Service Comunission MASSOURI

Issued: October 1, 2001

Effective: December 1, 2001

By JAN NEWTON., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri Missouri Public

FILED DEC 01 2001

Service Commission

Missouri Public

RECT OCT 01 2001

Service Commission

No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 3 Original Sheet 14

Missouri Public Service Commission

RECT OCT 0 2 2000

Maximum

ACCESS SERVICES

3. CARRIER COMMON LINE ACCESS SERVICE-(Continued)

3.8 Rates and Charges

The rates for Carrier Common Line Access are:

	Rate
InterLATA Access	
Access Minute, each terminating	\$0.0179662
Access Minute, each originating	0.0099080
IntraLATA Access	
Access Minute, each terminating	\$0.0179662
Access Minute, each originating	0.0099080

CANCELLED DEC 0 1 2001 และอเฉก Pub

Missouri Public Service Commission

FILED DEC 01 2000



ĺ

Issued: October 2, 2000

Effective: December 1, 2000

By JAN NEWTON., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 4 Original Sheet 1

ACCESS SERVICES

Reserved for Future Use

Issued: December 29, 1983

Effective: January 01, 1984

By R. D. Barron, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE

5.1 General

This Section sets forth the regulations and order-related charges for Access Orders for Switched and Special Access Services. These charges are in addition to other applicable charges as set forth in other sections of this Tariff.

An Access Order is an order to provide the IC with Switched Access Service or Special Access Service or to provide changes to existing services.

5.1.1 Ordering Conditions

A customer may order any number of services of the same type and between the same premises on a single Access Order. All details for services for a particular order must be identical except for those for multipoint service.

The customer shall provide all information necessary for the Telephone Company to provide and bill for the requested service. In addition to the order information required in Paragraph 5.2, following, the customer must also provide:

- Customer name and premises address(es).
- Billing name and address (when different from customer name and address).
- Customer contact name(s) and telephone number(s) for the following provisioning activities: order negotiation, order confirmation, interactive design, installation and billing.

The Telephone Company will establish a service date when the customer has provided an Access Order that contains the information as required for each respective service described in this and other sections of the tariff. The date on which the service date is established is considered to be the Application Date. The Telephone Company will provide a firm order confirmation to the customer and will advise the customer of the application Date and the Service Date.

Orders for FGA and BSA-A Switched Access Service must be in multiples of 30 busy-hour minutes of capacity.

Issued: March 26, 1993

Effective:

April 11, 1993

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 5 2nd Revised Sheet 1 Replacing 1st Revised Sheet 1

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE

AUG 9 1991

RECEIVED

5.1 General

MISSOURI

This Section sets forth the regulations and order-rela**RedbinaService Commission** Orders for Switched and Special Access Services. These charges are in addition to other applicable charges as set forth in other sections of this Tariff.

An Access Order is an order to provide the IC with Switched Access Service or Special Access Service or to provide changes to existing services.

5.1.1 Ordering Conditions

A customer may order any number of services of the same type and between the same premises on a single Access Order. All details for services for a particular order must be identical except for those for multipoint service.

The customer shall provide all information necessary for the Telephone Company to provide and bill for the requested service. In addition to the order information required in Paragraph 5.2, following, the customer must also provide:

- Customer name and premises address(es).
- Billing name and address (when different from customer name and address).
- Customer contact name(s) and telephone number(s) for the following provisioning activities: order negotiation, order confirmation, interactive design, installation and billing.

The Telephone Company will establish a service date when the customer has provided an Access Order that contains the information as required for each respective service described in this and other sections of the tariff. The date on which the service date is established is considered to be the Application Date. The Telephone Company will provide a firm order confirmation to the customer and will advise the customer of the application Date and the Service Date.

Orders for Feature Group A Switched Access Service must be in multiples of 30 busy-hour minutes of capacity. CANCELLED

Issued: 116 0 9 1991

Effective: SEP 3 0 1991 SEP 3 0 1991 SFP 3 0 1991

APR 11 1993 #

FILED

Public Service Commission

MISSOURI

By R. D. BARRON, President-Missouri DivisionPublic Service Commis Southwestern Bell Telephone Company St. Louis, Missouri

Access	Services	Tariff
	See	ction 5
	Revised_S	
Replacing)(riginal ;	Sheethl
Replacing) (i)	LOLU	VEU

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE JUN 2.7 1980

MISSUUR

5.1 General

This Section sets forth the regulations and order-related charges for Access Orders for Switched and Special Access Services. These charges are in addition to other applicable charges as set forth in other sections of this Tariff.

An Access Order is an order to provide the IC with Switched Access Service or Special Access Service or to provide changes to existing services.

5.1.1 Ordering Conditions

A customer may order any number of services of the same type and between the same premises on a single Access Order. All details for services for a particular order must be identical except for those for multipoint service.

The customer shall provide all information necessary for the Telephone Company to provide and bill for the requested service. In addition to the order information required in Paragraph 5.2, following, **WANCELED** must also provide:

- Customer name and premises address(es).
- Billing name and address (when different from custome By 2^{md} R.S. 1 name and address).
- Customer contact name(s) and telephone number(s) For the Commission following provisioning activities: order negotiation, MISSOURI order confirmation, interactive design, installation and billing.

The Telephone Company will establish a service date when the customer has provided an Access Order that contains the information as required for each respective service described in this and other sections of the tariff. The date on which the service date is established is considered to be the Application Date. The Telephone Company will provide a firm order confirmation to the customer and will advise the customer of the Application Date and the Service Date.

Orders for Feature Group A Switched Access Service must be in multiples of 30 busy-hour minutes of capacity.

5.69 1 1986 86-84 Public Service Commission

SEP 3 0 1991

Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff

ACCESS SERVICES

DEC 2 2 1003

Elissouri Public Service Commission

> JAN - 1 (931 83 - 253

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE

5.1 General

This Section sets forth the regulations and order-related charges for Access Orders and Planned Facilities Orders for Switched and Special Access Services. These charges are in addition to other applicable charges as set forth in other sections of this Tariff.

An Access Order is an order to provide the IC with Switched Access Service or Special Access Service or to provide changes to existing services. A Planned Facilities Order is an order to make facilities available on an established activation date for an IC's future Access Service needs.

For purposes of ordering Access Service, an End User who orders Access Service will be considered to be an IC and will be required to provide the same information that a non-End User IC must provide when placing an order.

5.1.1 Ordering Conditions

An IC may order any number of services of the same type (e.g., Feature Group, interface group, facility type, etc.) between the same locations (i.e., the IC terminal location and end office, Hub or End User's premises) on a single Access Order or Planned Facilities Order.

The IC shall supply all the necessary information to provide service, e.g., IC name and terminal location, End User's name and location, facility interface, etc. All details for a particular order must be identical except for those for multipoint services. The minimum order requirements are set forth in Paragraphs 5.2 and 5.4, following.

Orders for Feature Group A Swindfiel CCESS Service must be in multiples of 30 busy-hour minutes of cepacity.

Dedicated Access Line Special Access Service must be ordered for use with a Feature Group C or D Switched Access Service which is in service or on order. BY ACCESS Service which is in service or on

5.1.2 Provision of Other Services OF MUSSION

A. Testing Service, Additional Engineering, Additional Labor, Restoration Priority and Special Facilities Routing may be ordered with an Access Order or Planned Facilities Order as additional features at the time the order is accepted by the Telephone Company. The rates and charges for these services, as set forth in other sections of this Tariff, will apply in addition to the ordering charges set forth in this Section and the rates and charges for the Access Service or Planned Facilities with which they are associated.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 2nd Revised Sheet 1.01 Replacing 1st Revised Sheet 1.01

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

AUG 9 1991

- 5.1 General-(Continued)
- 5.1.2 Provision of Other Services

MISSOURI Public Service Commission

A. Testing Service, Additional Labor, Telecommunications Service Priority and Special Facilities Routing shall be ordered with an Access Order or as set forth in Paragraph 5.1.2, B., following. The rates and charges for these services, as set forth in other sections of this Tariff, will apply in addition to the ordering charges set forth in this section and the rates and charges for the Access Service with which they are associated.

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

FILED Issued: AUG 0 9 1991 By R. D. BARRON, President-Missouri Divisipablic Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 1st Revised Sheet 1.01 Replacing Original Sheet 1.01 RECEIVED

ACCESS SERVICES

JUN 28 1991

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) MISSOUR

5.1 General-(Continued)

Public Service Commission

5.1.2 Provision of Other Services

(CT)

A. Testing Service, Additional Labor, Telecommunications Service Priority and Special Facilities Routing shall be ordered with an Access Order or as set forth in Paragraph 5.1.2, B., following. The rates and charges for these services, as set forth in other sections of this Tariff, will apply in addition to the ordering charges set forth in this section and the rates and charges for the Access Service with which they are associated.

CANCELLED

SEP 3 0 1991 BY 2 R.S. I. OI Public Service Commission MISSOURI

FILED

Issued: JUL - 2 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 5 Original Sheet 1.01

REGEIVED

JUN 2 7 1980

MISSUURI

Cablic Service Commission

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE

5.1 General-(Continued)

5.1.2 Provision of Other Services

A. Testing Service, Additional Labor, Restoration Priority and Special Facilities Routing shall be ordered with an Access Order or as set forth in Paragraph 5.1.2, B., following. The rates and charges for these services, as set forth in other sections of this Tariff, will apply in addition to the ordering charges set forth in this section and the rates and charges for the Access Service with which they are associated.

CANCELLED

AUG 2 1991 #, of BY <u>John R.S.</u>, of Public Service Commission MISSOURI





Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri
Access Services Tariff Section 5 2nd Revised Sheet 2 Replacing 1st Revised Sheet 2

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.1 General-(Continued)

5.1.2 Provision of Other Services-(Continued)

- B. With the agreement of the Telephone Company, the items listed in Paragraph A., preceding, may subsequently be added to the order at any time, u p to and including the service date for the Access Service. When added subsequently, charges for a design change as set forth in Paragraph 5.2.3, C., following, will apply when an engineering review is required.
- C. Additional Engineering is not an ordering option, but will be applied to an Access Order when the Telephone Company determines that Additional Engineering is necessary to accommodate a customer's request. Additional Engineering will only be required as set forth in Paragraph 13.1, following. When it is required, the customer will be so notified and will be furnished with a written statement setting forth the justification for the Additional Engineering as well as an estimate of the charges. If the customer agrees to the Additional Engineering, a firm order will be established. If the customer does not want the service or facilities after being notified that Additional Engineering of Telephone Company facilities is required, the order will be withdrawn and no charges will apply. Once a firm order has been established, the total charge to the customer for the Additional Engineering may not exceed the estimated amount by more than 10 percent.

The regulations, rates and charges for Additional Engineering are as set forth in Paragraph 13.1, following, and are in addition to the regulations, rates and charges specified in this Section.

5.1.3 Special Construction

The regulations, rates and charges for special construction are set forth in Section 14, following, and are in addition to the regulations, rates and charges specified in this Section.

5.2 Access Order

An Access Order is used by the Telephone Company to provide to an IC Access Service as follows:

- Switched Access Services as set forth in Section 6, following.
- Special Access Services as described in Section 7, following, and
- Other Services as set forth in Paragraph 5.1.2, preceding.

August 9, 1991

Effective:

September 9, 1991



CT)

C)

(CP)ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Section 5 1st Revised Sheet 2 Replacing_Original Sheet 2 REGE

Access Services Tariff

MISSUURI

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) JUN 2 7 1980

- 5.1 General-(Continued)
- 5.1.2 Provision of Other Services-(Continued)
- Fublic Service Commission B. With the agreement of the Telephone Company, the items listed in Paragraph A., preceding, may subsequently be added to the order at any time, up to and including the service date for the Access Service. When added subsequently, charges for a design change as set forth in Paragraph 5.2.2, C., following, will apply when an engineering review is required. CANCELLED
- C. Additional Engineering is not an ordering option, but will be applied to an Access Order when the Telpehone Company determines SEP 3 0 1991 that Additional Engineering is necessary to accommodate a customer's $\sim R, S, J$ request. Additional Engineering will only be required as set forth in Paragraph 13.1, following. When it is required, the cuRuchlie Service Commission will be so notified and will be furnished with a written statementMISSOURI setting forth the justification for the Additional Engineering as well as an estimate of the charges. If the customer agrees to the Additional Engineering, a firm order will be established. If the customer does not want the service or facilities after being notified that Additional Engineering of Telephone Company facilities is required, the order will be withdrawn and no charges will apply. Once a firm order has been established, the total charge to the customer for the Additional Engineering may not exceed the estimated amount by more than 10 percent.

The regulations, rates and charges for Additional Engineering are as set forth in Paragraph 13.1, following, and are in addition to the regulations, rates and charges specified in this Section.

5.1.3 Special Construction

The regulations, rates and charges for special construction are set forth in Section 14, following, and are in addition to the regulations, rates and charges specified in this Section.

5.2 Access Order

An Access Order is used by the Telephone Company to provide to an ICLACCES Service as follows:

- Switched Access Services as set forth in Section 6, following, \$5 * 84 1986

- Other Services as set forth in Paragraph 5.1.2, preceding DONG SCHICE COMMISSION

Issued: JUN 27 1986

1 1986 JUL Effective:

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff

L'ISSOURI

Public Service Commission

Section 5

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5.

- 5.1 General-(Continued)
- 5.1.2 Provision of Other Services-(Continued)
 - B. With the agreement of the Telephone Company, the items listed in Paragraph 5.1.2, A., preceding, may subsequently be added to the order at any time, up to and including the service date for the Access Service or the activation date for the Planned Facilities. When added subsequently, charges for order modifications as set forth in Paragraph 5.2.2 or 5.4.5, following, will apply.
- 5.1.3 Special Construction

Special Construction as set forth in Section 14, following, is not an ordering option, but may be applied to either an Access Order or a Planned Facilities Order to accommodate an IC request. The IC will be notified that Special Construction Charges will apply. Such charges and liabilities are in addition to the regulations, rates and charges specified in this Section.

If the IC agrees to the Special Construction, a firm order will be established. If the IC does not want the service after being notified that Special Construction is required, the order will be withdrawn and no charges will apply unless the IE was requested a quotation.

5.2 Access Order

1986

An Access Order is used by the Telephone Company to provide to an IC Access Service as follows: - Switched Access ServicesEPANCE COMMISSION - Switched Access ServicesEPANCE COMMISSION of Section 6, following,

- Switched Access Services as described in Section 7, following, and EU
- Other Services as set forth in Paragraph 5.1.2, preceding.

When placing an order for Access Service, the IC must provide, 44.% - 1.193%minimum, the following information: 83 - 253

Puhlin Par in Da-- For Switched Access Service, the IC must specify the number of busyhour minutes of capacity (BHMC) from the IC terminal location to the end office by Feature Group and by type of BHMC. In addition, for Feature Group A, the IC must specify whether the ordered capacity is for foreign dial tone service, second dial tone service or switched private network service. This information is forwarded to the Circuit Administration Center (CAC) for networking design, then back to the IC for specifying the Interface Group Supervisory Signaling and Local Switching options.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 5th Revised Sheet 3 Replacing 4th Revised Sheet 3

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

When placing an order for Access Service, the IC must identify itself as a certified LEC, a certificated IXC, or an ESP for the services being ordered and provide, at a minimum, the following information:

- For the DNAL Switched Access BSA, the customer shall specify the designated customer premises, the channel interface and the line side or trunk side Switched Access BSA to be associated with the DNAL.
- For FGA and BSA-A Switched Access Service, the customer shall specify the number of lines and the first point of switching (i.e., dial tone office). The first point of switching shall be selected in accordance with the provisions set forth in Section 2, Paragraph 2.1.2.A., and cannot be an end office served by an RSS or RSM because all traffic originating from or terminating at a remote switching office must switch through its host office. The customer shall also specify the directionality of the service, the Local Transport and Local Switching features desired. The IXCs or LECs must also specify whether the ordered lines or capacity is for FX/ONAL or MTS/WATS-type service. The customer shall also specify which MTS/WATS-type FGA or BSA-A lines are to be arranged in multiline hunt group arrangements and which lines are to be provided as single lines. The customer shall also specify that the FX/ONAL is to be provided with an extension to a different LATA, if applicable. When such an extension is specified on the order, the customer must also specify the customer-designated premises within the LATA of the Switched Access

Feature Group A

service, at which the FGA or BSA-A extension is to be terminated.

For FGB or BSA-B Switched Access Service, the customer shall specify the number of trunks, directionality of the service, the entry switch and the Local Transport and Local Switching features desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. When an end office is the specified entry switch, the Telephone Company will work cooperatively with the customer to determine whether direct trunking of the service to the end office is necessary. Direct trunking to an end office will be installed when a customer orders features that can only be provided through direct trunking.

Issued:	March 26, 1993	Effective:	April 11, 1993	
	5	ON, President-Missouri ern Bell Telephone Com		FILED
	St. Louis, Missouri			MO PSC

(AT)

(A'T)

(AT)

(AT)

(MT)

(MT)

(CT)

Access Services Tariff Section 5 4th Revised Sheet 3 Replacing 3rd Revised Sheet 3 RECEIVED

ACCESS SERVICES

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) NOV 19 1971
- 5.2 Access Order-(Continued)

- MO. PUBLIC SERVICE COMM.
- When placing an order for Access Service, the IC must identify itself as a certified LEC, a certificated IXC, or an ESP for the services being ordered and provide, at a minimum, the following information:
 - For Feature Group A Switched Access Service, the customer shall specify the number of lines and the first point of switching (i.e., dial tone office). The first point of switching shall be selected in accordance with the provisions set forth in Section 2, Paragraph 2.1.2.A., and cannot be an end office served by an RSS or RSM because all traffic originating from or terminating at a remote switching office must switch through its host office. The customer shall also specify the directionality of the service, the Local Transport and Local Switching features desired. The IXCs or LECs must also specify whether the ordered lines or capacity is for FX/ONAL or MTS/WATS-type service. The customer shall also specify which MTS/WATS-type FGA lines are to be arranged in multiline hunt group arrangements and which lines are to be provided as single lines.

The customer shall also specify that the FX/ONAL is to be provided with an extension to a different LATA, if applicable. When such an extension is specified on the order, the customer must also specify the customer-designated premises within the LATA of the Switched Access Feature Group A service, at which the FGA extension is to be terminated.

- For Feature Group B Switched Access Service, the customer shall specify the number of trunks, directionality of the service, the entry switch and the Local Transport and Local Switching features desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. When an end office is the specified entry switch, the Telephone Company will work cooperatively with the customer to determine whether direct trunking of the service to the end office is necessary. Direct trunking to an end office will be installed when a customer orders features that can only be provided through direct trunking.

CANCELLED

APR 11 1993 BY ST QSF3 Public Service Commission MISSOURI FEB 12 1992 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri Public Service Commissio

Access Services Tariff Section 5 3rd Revised Sheet 3 Replacing 2nd Revised Sheet 3 RECEIVED

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 9 1991

5.2 Access Order-(Continued)

MISSOURI

When placing an order for Access Service, the IC must ployide, yice Commission minimum, the following information:

- For Feature Group A Switched Access Service, the customer shall specify the number of lines and the first point of switching (i.e., dial tone office), the directionality of the service, the Local Transport and Local Switching features desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. In addition, the customer shall specify whether the ordered lines or capacity is for FX/ONAL or MTS/WATS-type service. The customer shall also specify which MTS/WATS-type FGA lines are to be arranged in multiline hunt group arrangements and which lines are to be provided as single lines.

The customer shall also specify that the FX/ONAL is to be provided with an extension to a different LATA, if applicable. When such an extension is specified on the order, the customer must also specify the customer-designated premises within the LATA of the Switched Access Feature Group A service, at which the FGA extension is to be terminated.

- For Feature Group B Switched Access Service, the customer shall specify the number of trunks, directionality of the service, the entry switch and the Local Transport and Local Switching features desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. When an end office is the specified entry switch, the Telephone Company will work cooperatively with the customer to determine whether direct trunking of the service to the end office is necessary. Direct trunking to an end office will be installed when a customer orders features that can only be provided through direct trunking.

CANCELLED

FEB 12 1992 BY <u>Hth</u> R.S. 3 Public Service Commission MISSOURI

FILED

SEP 3 0 1991

Issued: AUG 0 9 1991

By R. D. BARRON, President-Missouri Division Schrödent-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

(CT)

(RT)

(CT)

(RT)

Access Services Tariff Section 5 2nd Revised Sheet 3 Replacing 1st Revised Sheet 3

RECEIVED

ACCESS SERVICES

JAN 1 9 1988

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued SOURI Public Service Commission

5.2 Access Order-(Continued)

When placing an order for Access Service, the IC must provide, at a minimum, the following information:

- For Feature Group A Switched Access Service, the customer shall specify the number of lines and the first point of switching (i.e., dial tone office), the directionality of the service, the Local Transport options and Local Switching options desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. In addition, the customer shall specify whether the ordered lines or capacity is for FX/ONAL or MTS/WATS-type service. The customer shall also specify which MTS/WATS-type FGA lines are to be arranged in multiline hunt group arrangements and which lines are to be pro**CANCELLED** single lines.

The customer shall also specify that the FX/ONAL is to be provided with an extension to a different LATA, if applicable. When such an extension K.S.^{*}3 is specified on the order, the customer must also specify the customer designated premises within the LATA of the Switched Access Feature Group Areas service, at which the FGA extension is to be terminated.

- For Feature Group B Switched Access Service, the customer shall specify the number of trunks, directionality of the service, the entry switch and the Local Transport and Local Switching options desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. When an end office is the specified entry switch, the Telephone Company will work cooperatively with the customer to determine whether direct trunking of the service to the end office is necessary. Direct trunking to an end office will be installed when a customer orders optional features that can only be provided through direct trunking.

FILED

FEB 19 1988

Issued: JAN 20 1988

Effective: FEB 19 19 Solic Service Commission

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



(CP)

(CP)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff
Section 5
1st Revised Sheet 3
Replacing Original Sheet 3
REGEOVED

(CP)ACCESS SERVICES

JUN 2 7 1986

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) Public Service Commission

5.2 Access Order-(Continued)

When placing an order for Access Service, the IC must provide, at a minimum, the following information:

- For Feature Group A Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) and the first point of switching (i.e., dial tone office), the directionality of the service, the Local Transport options and Local Switching options desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. In addition, the customer shall specify whether the ordered lines or capacity is for FX/ONAL or MTS/WATS-type service. The customer shall also specify which MTS/WATS type FGA lines are to be arranged in multiline hunt group arrangements and which lines are to be provided as single lines.

The customer shall also specify that the FX/ONAL is to be provided with an extension to a different LATA, if applicable. When such an extension is specified on the order, the customer must also specify the customer-designated premises within the LATA of the Switched Access Feature Group A service, at which the FGA extension is to be terminated.

- For Feature Group B Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC), directionality of the service, the entry switch and the Local Transport and Local Switching options desired. The first point of switching cannot be an end office served by an RSS or RSM because all traffic originating or terminating at a remote switching office must switch through its host office. When an end office is the specified entry switch, the Telephone Company will work cooperatively with the customer to determine whether direct trunking of the service to the end office is necessary. Direct trunking to an end office will be installed when a customer orders optional features that can only be provided through direct trunking.

			in the second se				
			CANCELLED				1
			FEB 19 1988 BY 9 nub & S#3 BY 9 nub & S#3	6	JUL 1 86.	1986 - 84	1
			Public Service Commission		Public Sanne	<u>e Commiss</u>	SIUn
-	Issued:	JUN 27 1986	Public Service MISSOLHEctive:	JUL	1 1986		
			. BARRON, President-Missouri Div Athwestern Bell Telephone Compan				

St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 3

RECEIVED

EIISSUURI.

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE- (Continued) (CC3

- 5.2 Access Order-(Continued)
 - Directory Assistance Service is ordered in the same manner (as(SwitchedS'CN Access Service and requires the same information.
 - For Special Access Service, the IC must specify the IC terminal location and End User's premises or Hub, the type of service (e.g., Narrowband 1, Voice Grade 3, High Capacity 1, etc.) and the facility interface combination. When ordering to a Hub, no interface combinations are required but Hub functions must be specified. For multipoint services, the facility interface at each End User's premises may be different, but they must all work in combination with a common facility interface at the IC terminal location.
 - The IC must always specify how Special Access Service is to be used so that the Telephone Company may determine the appropriate application of the Special Access Service Surcharge. The surcharge and its applications are described in Section 7, Paragraph 7.4.2, following.
 - For Dedicated Access Lines, the IC must specify the End User's premises, the type of facility (i.e., two-wire or four-wire), the type of calling (i.e., originating or terminating) and the type of supervisory signaling. When the necessary screening functions are not provided at the End User's serving wire center, the IC may specify, with the agreement of the Telephone Company, a specific foreign serving wire center where capacity exists.
 - 5.2.1 Access Order Service Date Intervals

Access Service is provided with one of the following Service Date Intervals:

- Standard Interval
- Negotiated Interval
- Short Notice Interval



原品[[[]]) JAN = 1 (9.11 83-253 Dight for the o

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

When ordering FGB or BSA-B capacity to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same categories as described in Paragraph 6.1.1, following, to enable efficient provisioning and billing functions.

For FGC, FGD, BSA-C and BSA-D Switched Access Service, the customer s hall specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an RSS or RSM),

by Feature Group or BSA and traffic type. The Telephone Company shall design and determine the routing, including the selection of the entry switch in accordance with Section 6, Paragraph 6.8.2, following. This information is used to determine the number of transmission paths as set forth in Section 6, Paragraph 6.8.5, following. The customer then specifies the Local Transport and Local Switching features.

Customers may, at their option, order FGD or BSA-D by specifying the number of trunks desired between the customer's premises and an entry switch. When ordering by trunk quantities to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements.

When a customer orders FGD or BSA-D Switched Access Service, the capacity ordered in an Access Tandum Network must be ordered in either BHMCs of trunks and cannot be mixed. When ordering FGD or BSA-D trunks with SS7 Signaling, in addition to the other information listed in Section 5, the customer shall specify the switching point codes and trunk circuit identification codes. The customer must also identify the Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service link associated with the FGD or BSA-D trunk group.

Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report.

Issued:	March 7, 1994
	By M. H. SCHULTEIS, Divisi

Effective:

April 7, 1994



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 4th Revised Sheet 3.01 Replacing 3rd Revised Sheet 3.01 RECEIVED

ACCESS SERVICES

MAR 29 1993

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

- (AT) When ordering FGB or BSA-B capacity to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same
 (RT) categories as described in Paragraph 6.1.1, following, to enable efficient provisioning and billing functions.
- (AT) For FGC, FGD, BSA-C and BSA-D Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an RSS or RSM),
 (AT) by Feature Group or BSA and traffic type. The Telephone Company shall design and determine the routing, including the selection of the entry
 (FC) switch in accordance with Section 6, Paragraph 6.8.2, following. This information is used to determine the number of transmission paths as set
 (FC) forth in Section 6, Paragraph 6.8.5, following. The customer then specifies the Local Transport and Local Switching features.
- (AT) Customers may, at their option, order FGD or BSA-D by specifying the number of trunks desired between the customer's premises and an entry switch. When ordering by trunk quantities to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements.
- (AT) When a customer orders FGD or BSA-D Switched Access Service, the capacity ordered in an Access Tandum Network must be ordered in either BHMCs or trunks and cannot be mixed.

Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report. CANCELLED

#301 vice Commission Public Ser MISSOURI

FILED

APR 1 1 1993 92 - 304 MO. PUBLIC SERVICE COMM.

Issued: MAR 2 6 1993

Effective:

APR 1 1 1993

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 3rd Revised Sheet 3.01 Replacing 2nd Revised Sheet 3.01 **RECEIVED**

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

AUG 9 1991

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

When ordering FGB capacity to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same categories as described in Section 6.1.1(E), following, to enable efficient provisioning and billing functions.

For Feature Group C and D Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an RSS or RSM), by Feature Group and traffic type. The Telephone Company shall design and determine the routing, including the selection of the entry switch in accordance with Section 6.5.2, following. This information is used to determine the number of transmission paths as set forth in Section 6.5.5, following. The customer then specifies the Local Transport and Local Switching features.

Customers may, at their option, order FGD by specifying the number of trunks desired between the customer's premises and an entry switch. When ordering by trunk quantities to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements.

When a customer orders FGD Switched Access Service, the capacity ordered in an Access Tandem Network must be ordered in either BHMCs or trunks and cannot be mixed.

Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report.

CANCELLED

APR 11 1993 BY <u>HR R S 3.01</u> Public Service Commission <u>MISSOURI FILED</u>

Issued: AUG 0 9 1991

(CT)

Effective: SEP 3 0 1991 SEP 3 0 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 2nd Revised Sheet 3.01 Replacing 1st Revised Sheet 3.01

RECEIVED

ACCESS SERVICES

JAN 1 9 1988 MISSOURI

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE - Commission

5.2 Access Order-(Continued)

No supplement to this

tariff will be issued

(RT)

(AT)

(CP)

except for the purpose

of canceling this tariff.

When ordering FGB capacity to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same categories as described in Section 6.1.1(E), following, to enable efficient provisioning and billing functions.

For Feature Group C and D Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an RSS or RSM), by Feature Group and traffic type. The Telephone Company sGANCELLED design and determine the routing, including the selection of the entry switch in accordance with Section 6.5.2, following. This information Pig 0 1991 used to determine the number of transmission paths as set forth in Section 6.5.5, following. The customer then specifies the Local BY <u>3 CRS 3</u>.
 Transport and Local Switching options.

(CP) Customers may, at their option, order FGD by specifying the number of trunks desired between the customer's premises and an entry switch. When ordering by trunk quantities to an access tandem, the Telephone Company may request an estimate of the amount of traffic the customer will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements.

> When a customer orders FGD Switched Access Service, the capacity ordered in an Access Tandem Network must be ordered in either BEMCs or trunks and cannot be mixed.

- Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report.

FEB **19** 1988

6411

Issued: JAN 20 1988

Effective: (FEB 1 9 1988 Public Service Commissio)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 1st Revised Sheet 3.01 Replacing Original Sheet: 3) 01

ACCESS SERVICES

OCT 1 3 1987

- 5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) Public Service Commission
- 5.2 Access Order-(Continued)

When ordering FGB capacity to an access tandem, the customer must also provide the Telephone Company an estimate of the amount of traffic it will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same categories as described in Section 6.1.1(E), following, to enable efficient provisioning and billing functions.

- For Feature Group C and D Switched Access Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an RSS or RSM), by Feature Group and traffic type. The Telephone Company shall design and determine the routing, including the selection of the entry switch in accordance with Section 6.5.2, following. This information is used to determine the number of transmission paths as set forth in Section 6.5.5, following. The customer then specifies the Local Transport and Local Switching options.
- Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report.

CANCELLED FEB 2 9 1988 BY2MDR S#301 Public Service Commission MISSOURI

FILED

OCT 16 1987 TO-87-42 Public Service Commission

Issued: OCT 1 4 1987

(RT)

(RT)

* Effective: OCT 16 1987.

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 Original Sheet 3.01

12.1SSUAR

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

When ordering FGB capacity to an access tandem, the customer-must. also provide the Telephone Company an estimate of the amount of traffic it will generate to and/or from each end office subtending the access tandem to assist the Telephone Company in its own efforts to

project further facility requirements. Traffic estimates to end offices served by remote switching offices should be uniquely identified and not included in traffic estimates to the host. The traffic type must also be specified using the same categories as described in Section 6.1.1(E), following, to enable efficient provisioning and billing functions.

- For Feature Group C and D Switched Access Service, the customer challed commission specify the number of busy-hour minutes of capacity (BHMC) from the IC terminal location to end offices (including those served by an MSS or RSM), by Feature Group and traffic type. The Telephone Company shall design and determine the routing, including the selection of the entry switch in accordance with Section 6.5.2, following. This information is used to determine the number of transmission paths as set forth in Section 6.5.5, following. The customer then specifies the Local Transport and Local Switching options.
- Remote switching office to host relationships are provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report.
- The WATS Access Line Optional Feature must be ordered for use with FGC or FGD Switched Access Service. For the WATS Access Line optional feature, the customer shall specify the customer's premises at which the WATS Access Line terminates, the type of line (i.e., two-wire or four-wire), the type of calling (i.e., originating or terminating) and the type of Supervisory Signaling. When the necessary screening functions are not provided at the wire center which serves the customer's originating or terminating premises, the Telephone Company will use the nearest wire center premises where the screening capacity exists.

The customer must also specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if appliable. When such an extension is specified, the customer must provide either (1) the end user premises (for an IntraLATA extension) or (2), the customertomer-designated premises (for an extension in a different LATA) to which the extension is to be provided. 86-84

Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 7th Revised Sheet 3.02 Replacing 6th Revised Sheet 3.02

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

(MT) For 900 Access Service, customers must have LATA-wide Switched Access Service and shall specify the LATAs that they wish to receive originating 900 Access Service calls from, the 900-NXX codes that are to be activated in a given LATA and the desired due date of the order. 900 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

- (AT) 1+900 Access Service will be combined with either FGB, FGC, FGD, BSA-B,
 (MT) BSA-C or BSA-D Switched Access Service. The customer shall designate which FGB, FGC,
 FGD, BSA-B, BSA-C or BSA-D Switched Access Service trunk groups are to be associated with 900 Access Service. However, when FGD or BSA-D
 (AT) Switched Access Service becomes available in an end office, 1+900 Access Service traffic
- (AT) Switched Access Service becomes available in an end office, 1+900 Access Service traffic originating from that end office must be provided with FGD
 (MT) or BSA-D Switched Access Service.
 - 0+900 Access Service is available only when combined with 1+900 Access Service provided with FGD or BSA-D.

Information regarding 900 Access Service Screening Offices is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications section of Southwestern Bell's Tariff, FCC No. 73.

For 800 Number Portability Access Service (NPAS), direct routing will be provided from SSP equipped end offices, i.e., end offices equipped to provide customer identification. All 800 traffic originating from end offices not equipped to provide customer identification will require routing to an access tandem where the customer identification function is available. Information regarding 800 NPAS identification function is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications Section of Southwestern Bell Telephone Company's Interstate Tariff, FCC No. 73.

800 NPAS requires FGD or BSA-D Switched Access Service. The customer shall designate which originating FGD or BSA-D Switched Access Service trunk groups are to be associated with 800 NPAS. Calls originating from an Area of Service in which the Customer has not ordered sufficient originating FGD or BSA-D Switched Access Service will be blocked.



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 6th Revised Sheet 3.02 Replacing 5th Revised Sheet 3.02

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) MAR 13 1993

5.2 Access Order-(Continued)

(AT)

BC. BULLS TRUCK BANKS

(RT) For 900 Access Service, customers shall specify the LATAs that they wish to
 (RT) receive originating 900 Access Service calls from, the 900-NXX codes that are to be activated in a given LATA and the desired due date of the order.
 (RT) 900 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

- (RT) 900 Access Service will be combined with either FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service. The customer must have LATA-wide Switched
 (RT) Access Services in order to receive 900 Access Service traffic. The customer shall designate which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched
 (RT) Access Service trunk groups are to be associated with 900 Access Service. However, when FGD or BSA-D Switched Access Service becomes available in an
 (RT) end office, 900 Access Service traffic originating from that end office must be provided with FGD or BSA-D Switched Access Service. The customer must
 (RT) specify whether the 900 Access Service to be provided will be joint provided or non-joint provided.
- (RT) Information regarding 900 Access Service Screening Offices is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to
 (CT) Technical Publications section of Southwestern Bell's Tariff, FCC No. 73.

(AT) For 800 Number Portability Access Service (NPAS), direct routing will be provided from SSP equipped end offices, i.e., end offices equipped to provide customer identification. All 800 traffic originating from end offices not equipped to provide customer identification will require routing to an access tandem where the customer identification function is available. Information regarding 800 NPAS identification function is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications' Section of Southwestern Bell Telephone Company's Interstate (AT)

800 NPAS requires FGD or BSA-D Switched Access Service. The customer shall designate which originating FGD or BSA-D Switched Access Service trunk groups are to be associated with 800 NPAS. Calls originating from an Area of Service in which the Customer has not ordered sufficient originating FGD or BSA-D Switched Access Service CANOTEL blacked.

(MT) LAN 6 1995 BY <u>TUR</u> R. 5 # 3,050 Public Service Commission MISSOFFECTIVE: MAY - 1 1993 MAY 0 1 1993 MC. FUELIC SERVICE CONTA. Issued: MAR 2 2 1993 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 5th Revised Sheet 3.02 Replacing 4th Revised Sheet 3.02 RECEIVED

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MAR 29 1993

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

For 800 Access Service and 900 Access Service, customers shall specify the LATAs that they wish to receive originating 800 Access Service and 900 Access Service calls from, the 800-NXX and 900-NXX codes that are to be activated in a given LATA and the desired due date of the order. 800 Access Service and 900 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

800 Access Service and 900 Access Service will be combined with either FGB, (AT) FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service. The customer must (AT) have LATA-wide Switched Access Services in order to receive 800 Access Service and 900 Access Service traffic. The customer shall designate which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service trunk groups (AT) are to be associated with 800 Access Service and 900 Access Service. However, when FGD or BSA-D Switched Access Service becomes available in an (AT) end office, 800 Access Service and 900 Access Service traffic originating from that end office must be provided with FGD or BSA-D Switched Access (AT) Service. The customer must specify whether the 800 Access Service and 900 Access Service to be provided will be joint provided or non-joint provided.

> Information regarding 800 Access Service and 900 Access Service Screening Offices is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications section of Southwestern Bell's Tariff, FCC No. 68.

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be commbined with a FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service, the customer shall also specify which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport features.

For all Special Access Services, the customer must specify the customerdesignated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical specification package and features desired. For multipoint services, the channel interface at each premises may, at the nequest of the customer be different, but all such interfaces interfaces interface.

1 1993 82

Issued: MAR 2 6 1993

(AT)

(AT)

5 1993 BY GREENISSION PUBLIC SERVICE (SAMI 1 1993 By R. D. BAERON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 4th Revised Sheet 3.02 Replacing 3rd Revised Sheet 3.02

ACCESS SERVICES

RECEIVED

1991

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MISSOURI

AUG 9

5.2 Access Order-(Continued)

For 800 Access Service and 900 Access Service, customers shall specify the LATAs that they wish to receive originating 800 Access Service and 900 Access Service calls from, the 800-NXX and 900-NXX codes that are to be activated in a given LATA and the desired due date of the order. 800 Access Service and 900 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

800 Access Service and 900 Access Service will be combined with either Feature Group B, C or D Switched Access Service. The customer must have LATA-wide Switched Access Services in order to receive 800 Access Service and 900 Access Service traffic. The customer shall designate which Feature Group B, C or D Switched Access Service trunk groups are to be associated with 800 Access Service and 900 Access Service. However, when Feature Group D Switched Access Service becomes available in an end office, 800 Access Service and 900 Access Service traffic originating from that end office must be provided with Feature Group D Switched Access Service. The customer must specify whether the 800 Access Service and 900 Access Service to be provided will be joint provided or appoint provided.

Information regarding 800 Access Service and 900 Access Service Servic

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a Feature Group B, C or D Switched Access Service, the customer shall also specify which Feature Group B, C or D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport features.

(CT)

For all Special Access Services, the customer must specify the customerdesignated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical

(CT) specification package and features desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

ULED

Issued: AUG 0 9 1991

Effective SEP 3 0 1991 SEP 3 0 1991

By R. D. BARRON, President-Missouri Division Public Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 3rd Revised Sheet 3.02 Replacing 2nd Revised Sheet 3.02

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 1989

5.2 Access Order-(Continued)

MISSCUR

RECEIVED

 (AT) For 800 Access Service and 900 Access Service, customers shall specifysthe LATAs that they wish to receive originating 800 Access Service and 900
 (AT) Access Service calls from, the 800-NXX and 900-NXX codes that are to be activated in a given LATA and the desired due date of the order. 800
 (AT) Access Service and 900 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

800 Access Service and 900 Access Service will be combined with either (AT) Feature Group B, C or D Switched Access Service. The customer must have LATA-wide Switched Access Services in order to receive 800 Access Service (AT) and 900 Access Service traffic. The customer shall designate which Feature Group B, C or D Switched Access Service trunk groups are to be associated with 800 Access Service and 900 Access Service. However, when (AT) Feature Group D Switched Access Service becomes available in an end office. 800 Access Service and 900 Access Service traffic originating from (AT) that end office must be provided with Feature Group D Switched Access Service. The customer must specify whether the 800 Access Service and 900 (AT) Access Service to be provided will be joint provided or non-joint (AT) provided.

CANCELLED

 (AT) Information regarding 800 Access Service and 900 Access Service Screening
 (DT) Offices is contained in the Southwestern Bell End Office Profile report, Customers wishing to receive this report may obtain ordering internations
 (AT) from the Reference to Technical Publications section of Southwestern
 (AT) Bell's Tariff, FCC No. 68. Public Service Commission

MISSOURI

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a Feature Group B, C or D Switched Access Service, the customer shall also specify which Feature Group B, C or D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport options.

For all Special Access Services, the customer must specify the customer-designated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical specification package and options desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

Issued: DEC 2 9 1989

Effective: JAN 2 9 1990

JAN 29 1990

FILED

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Public Service Commission

5.

(AT)

(AT)

(AT)

Access Services Tariff Section 5 2nd Revised Sheet 3.02 Replacing 1st Revised Sheet 3.02 RECEIVED

ACCESS SERVICES

OCT 2 3 1987 ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI

Public Service Commission

For 800 Access Service, customers shall specify the LATAs that they wish to receive originating 800 Access Service calls from, the 800-NXX codes that are to be activated in a given LATA and the desired due date of the order. 800 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

800 Access Service will be combined with either Feature Group B, C or D Switched Access Service. The customer must have LATA-wide Switched Access Services in order to receive 800 Access Service traffic. The customer shall designate which Feature Group B, C or D Switched Access Service trunk groups are to be associated with 800 Access Service. However, when Feature Group D Switched Access Service becomes available in an end office, 800 Access Service traffic originating from that end office must be provided with Feature Group D Switched Access Service. The customer must specify whether the 800 Access Service to be provided will be joint provided or non-joint provided.

800 Access Service Screening Offices will be provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications section of this Tariff.

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a Feature Group B, C or D Switched Access Service, the customer shall also specify which Feature Group B, C or D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport options.

For all Special Access Services, the customer must specify the customer-designated premises or hubs involved, the type of service (e.g., Voice Grade, Eigh Capacity, etc.), the channel interface, technical specification package and options desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but **AD** such interfaces shall be compatible.

CANCEL FEB 1 1988 JAN 29 390 lic Service Commission 2 3 1987 BY 3 Commission FEB 1 BY 3 1987 BY 3 Commission FEB 1 By R. D. BARRON, Mesident-Missouri Division OCT 2 3 1987 . Issued: Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

(AT)

Access Services Tariff Section 5 1st Revised Sheet 3.02 Replacing Original Sheet 3.02

ACCESS SERVICES

RECEIVED

- ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5. OCT 1 3 1987
 - 5.2 Access Order-(Continued)
 - For 800 Access Service, customers shall specify the LATAS that they wish to receive originating 800 Access Service calls from, the 800 NXX codes solution that are to be activated in a given LATA and the desired due date of the order. 800 Access Service calls originating in LATAs which the customer has not ordered will be blocked.

800 Access Service will be combined with either Feature Group B, C or D Switched Access Service. The customer must have LATA-wide Switched Access Services in order to receive 800 Access Service traffic. The customer shall designate which Feature Group B, C or D Switched Access Service trunk groups are to be associated with 800 Access Service. However, when Feature Group D Switched Access Service becomes available in an end office, 800 Access Service traffic originating from that end office must be provided with Feature Group D Switched Access Service.

800 Access Service Screening Offices will be provided to all customers by the Telephone Company. This information is contained in the Southwestern Bell End Office Profile report. Customers wishing to receive this report may obtain ordering information from the Reference to Technical Publications section of this Tariff.

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a Feature Group B, C or D Switched Access Service, the customer shall also specify which Feature Group B, C or D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport options.

For all Special Access Services, the customer must specify the customer-designated premises or hubs involved, the type of service nical specification package and options desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

vine Commission

FILED

OCT 16 1987 TO-87-42 Public Service Commission

Issued: OCT 1 4 1987

Effective: OCT 1 6 1987.

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

MISSOURI

PUD

Access Services Tarif	
Section	
MEGEUV2U	2
MEARIAGM	
	4
JUN 27 1022	,

(CP)ACCESS SERVICES

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE- (Continued)) 5.

5.2 Access Order-(Continued)

Public Service Commission

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a Feature Group B, C or D Switched Access Service, the customer shall also specify which Feature Group B, C or D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport options.

For all Special Access Services, the customer must specify the customer-designated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical specification package and options desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

CANCELLED OCT 10 1987 BY SEC. S. C. Public Service Commission

FILED JUL 1 1986 86-84 Public Service Commission

Issued:

JUN 27 1986

Effective: 1 1986 JUL

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 7th Revised Sheet 3.03 Replacing 6th Revised Sheet 3.03

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service, the customer shall also specify which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport features.

For all Special Access Services, the customer must specify the customer-designated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical specification package and features desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.

(MT)

(MT)

Issued: Mar. 5, 1999





No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 6th Revised Sheet 3.03 Replacing 5th Revised Sheet 3.03

ACCESS SERVICES

MAR 13 1993

- NO. PULLE CALLECOM. 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)
- 5.2 Access Order-(Continued)

(MT)

(MT)

N

1999

APR 05

(MT)

Issued: MAR 2 2 1993

CANCEL

For Directory Assistance Service, the customer shall specify the number of busy-hour minutes of capacity (BHMC) from the customer's premises to the Directory Assistance location. If the Directory Assistance Service is to be combined with a FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service, the customer shall also specify which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service trunk group is to be associated with the Directory Assistance Service. This information is used to determine the number of transmission paths as set forth in Section 9.2, E.3., following. The customer then specifies the Directory Transport features.

For all Special Access Services, the customer must specify the customerdesignated premises or hubs involved, the type of service (e.g., Voice Grade, High Capacity, etc.), the channel interface, technical specification package and features desired. For multipoint services, the channel interface at each premises may, at the request of the customer, be different, but all such interfaces shall be compatible.

For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at . the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a g different LATA, if applicable. When such an extension is specified, the E customer must provide either (1) the end user premises (for an intraLATA

Section of the end user premises (for an extension in a section of (2) the customer-designated premises (for an extension in a Generator Call Processing, the customer must specify the specific LATAs section of the customer desires Operator Call Processing and whether O-Transfer Inward Assistance, or both are to be provided. The customer must also > g specify if FGC, FGD, BSA-C or BSA-D Switched Access Service will be used to interconnect between the OSS Tandem(s) and the customers premises and a whether or not operator functionality, coin station control, or both are to be provided by the customer.

511 - 14

MAY 0 11993

Effective: MAY - 1 1993 MO. PUPILIC SERVICE GIMM. By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 5th Revised Sheet 3.03 Replacing 4th Revised Sheet 3.03 RECEIVED

ACCESS SERVICES

MAR 23 1993 ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5.

5.2 Access Order-(Continued)

MISSOUR

MAY 1 1993 BY 64 R.S. # 3.03

For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.

For Operator Call Processing, the customer must specify the specific LATAs where the customer desires Operator Call Processing and whether O-Transfer Inward Assistance, or both are to be provided. The customer must also specify if FGC, FGD, BSA-C or BSA-D Switched Access Service will be used to interconnect between the OSS Tandem(s) and the customers premises and whether or not operator functionality, coin station control, or both are to be provided by the customer.

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.

- For Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service, the customer must provide the following information to the CANCELLED Telephone Company at the time of ordering:
 - Number of access links
 - Link Type
 - Signaling Link Code

 - Common Language Location Identifier (CLLI) code of cublic Service Commission Company interconnecting Signal Transfer Point
 - Contact telephone number for installation and maintenance of the HILED customer's designated premises

Issued: 92-304 Effective: MAR 2 6 1993 APR 1 1 1993 By A. D. ROBERTSON, Assistant Vice Presider RUBPICeSTRAL offering Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 4th Revised Sheet 3.03 Replacing 3rd Revised Sheet 3.03 RECEIVEN

ACCESS SERVICES

MAY 11 1892 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MO. PUELIC SETUCE COMMA

For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.

For Operator Call Processing, the customer must specify the specific LATAs where the customer desires Operator Call Processing and whether O-Transfer Inward Assistance, or both are to be provided. The customer must also specify if FGC or FGD Switched Access Service will be used to interconnect between the OSS Tandem(s) and the customers premises and whether or not operator functionality, coin station control, or both are to be provided by the customer.

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.

- For Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service, the customer must provide the following information to the Telephone Company at the time of ordering:

CANCELLED

APR 11 1993

BY STR.S

- Number of access links
- Link Type

(AT)

(AT)

- Signaling Link Code
- Customer Signaling Point Code
- Common Language Location Identifier (CLLI) Public Service Coepuission - Contact telephone number for installation and maintenance of the customer's designated promises
- customer's designated premises

MO. PHBLIC SERVICE COMM.

NOV 16 1992

Effective: Issued: MAY 1 3 1992 NUV 16 1992 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 3rd Revised Sheet 3.03 Replacing 2nd Revised Sheet 3.03

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 9 1991

5.2 Access Order-(Continued)

MISSOURI

blic Service Commission

For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.

For Operator Call Processing, the customer must specify the specific LATAs where the customer desires Operator Call Processing and whether O- Transfer, Inward Assistance, or both are to be provided. The customer must also specify if Feature Group C or Feature Group D Switched Access Service will be used to interconnect between the OSS Tandem(s) and the customers premises and whether or not operator functionality, coin station control, or both are to be provided by the customer.

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.

CANCELLED

NOV 16 1392 BY <u>4 R.S. # 3</u>03 Public Service Commission MISSOURI

Issued: AUG 0 9 1991

1991 Effective: SEP 3 0 1991 SEP 3 0 1991SEP 3 0 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

(C)

(RT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 2nd Revised Sheet 3.03 Replacing 1st Rec E 3.02D

ACCESS SERVICES

JUL 17 1990

- ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5. MISSOURI
- 5.2 Access Order-(Continued)

^Dublic Service Commission For WATS Access Line Service the customer must also specify the type of

calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified in the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.

For Operator Call Processing, the customer must specify the specific LATAs where the customer desires Operator Call Processing and whether 0- Transfer, Inward Assistance, or both are to be provided. The customer must also specify if Feature Group C or Feature Group D Switched Access Service will be used to interconnect between the OSS Tandem(s) and the customers premises and whether or not operator functionality, coin station control, or both are to be provided by the customer.

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook. Customers wishing to receive this information may obtain ordering information from the Reference to Technical Publications section of this tariff.

CANCELLED

SEP 3 0 1991 🖛 BY 3 ~ R. S. 3,03 **Public Service Commission** MISSOURI

Issued: JUL 1 7 1990

(AT)

(AT)

Effective: AUG 1 7 1990

FILED

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

AUG 17 1990

Public Service Commission

(CT)

(RT)

RT)

Access Services Tariff Section 5 1st Revised Sheet 3.03 Replacing Original Sheet 3.03

ACCESS SERVICES

RECEIVED

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) UC 1 2 3 1987

5.2 Access Order-(Continued)

MISSOUR!

For WATS Access Line Service the customer must also specify chertype Gommission calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., non-joint provided or joint provided, WAL), the type of address signaling and the type of supervisory signaling.

Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified in the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to which the extension is to be provided.



1988

FILED

FEB 1 1988

Public Service Commission

Issued: OCT 2 3 1987

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective: 0

5.

Access Services Tariff Section 5 Original Sheet 3.03

RECEIVED

ACCESS SERVICES

OCT 1 3 1987 ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

(AT) For WATS Access Line Service the customer must also specify the type of calling (i.e., originating or terminating) for which the service is to be provided, the type of WAL (i.e., interLATA only or jointly provisioned WAL), the 800 telephone number the customer has assigned to the terminating WATS Access Line Service, the type of address signaling and the type of supervisory signaling. When WATS Access Line Service is ordered for use in terminating 800 Access Service traffic, the customer shall specify the Switched Access Feature Group that will carry this terminating traffic. When Feature Group A (FGA) access is used in conjunction with WATS Access Line Service, the customer must also identify the FGA seven digit local telephone number used to complete the 800 Access Service calls. Additionally, when the necessary screening functions are not provided at the wire center which serves the customer premises, the Telephone Company will provide the service to the nearest wire center where capacity exists. In these circumstances, the customer will be so notified and the order will be changed to designate the appropriate premises. No charge will apply for the order modification. In addition, the customer must specify that the WATS Access Line is to be provided with an extension in the same or a different LATA, if applicable. When such an extension is specified, the customer must provide either (1) the end user premises (for an intraLATA extension) or (2) the customer-designated premises (for an extension in a different LATA) to (AT) which the extension is to be provided.



FILED

OCT 16 1987 TO-87- 42 Public Service Commission

Issued: OCT 1 4 1987

Effective: OCT 16 1987

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

For Operator Call Processing, the customer must specify the FGC, FGD, BSA-C or
BSA-D switched Access Service that will be used to transfer calls from the OSS tandem to the access customer. For Operator Transfer, a separate trunk group and CIC must be established for each name to which Operator Transfer calls are to be transferred.
Customers who wish to participate in Operator Transfer and do not presently have a CIC, will be required to obtain a four-digit CIC.

The customer must also specify the specific LATAs where the customer desires Operator Call Processing and whether Operator Transfer, Inward Assistance or both are to be provided. In addition, the customer shall specify whether or not operator functionality, coin station control or both are to be provided by the customer.

(CT)

Issued: Mar. 5, 1999

By PRISCILLA HILL-ARDOIN, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.

(AT) Carrier Identification Code Parameter (CIP)

The customer must specify per trunk group or tandem, and all associated Carrier Identification Codes, the end offices that will be used to pass originating Carrier Identification Code Parameter (CIP) to the customer's designated premises. The customer's end office must be equipped with SS7 Signaling and is provided in conjunction with Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnnection Service as set forth in Section 23. End offices so equipped may be found in NECA Tariff F.C.C. No. 4.

- For Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service, the customer must provide the following information to the Telephone Company at the time of ordering:
- Number of access links
- Link Type
- Signaling Link Code
- Customer Signaling Point Code
- Common Language Location Identifier (CLLI) code of the Telephone Company interconnecting Signal Transfer Point
- Contact telephone number for installation and maintenance of the customer's designated premises

When ordering CCS/SS7 Interconnection Service, the customer will provide an estimate of total annual volume and busy hour busy month volume projected for a period of three years. The forecast should be itemized by message type. The Telephone Company will utilize this forecast in its own efforts to project further facility requirements.

 For LIDB Validation Service, the customer shall provide a LIDB Validation Service Order Form which specifies the OPCs of the customer's designated OSSs sending the query or queries, the PIU per OPC of the customer's OSS location, and the desired due date of the order.

LIDB Validation Service is provided in conjunction with CCS/SS7 Interconnection Service, as set forth in Section 20. In order to utilize LIDB Validation Service, the customer must have CCS/SS7 Interconnection Service to the two Telephone Company STPs designated by the Telephone Company as the interconnecting STP pair to be utilized for interconnection to the CCS/SS7 network. The Telephone Company's STP locations are provided in the National Exchange Carriers Association, Inc. Tariff F.C.C. No. 4.

Issued: JULY 27, 1999

Effective:

AUGUST 26, 1999

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 2nd Revised Sheet 3.04 Replacing 1st Revised Sheet 3.04

ACCESS SERVICES

RECEIVED

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)
 - AUG 18 1993

5.2 Access Order-(Continued)

MISSOURI

Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.

- For Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service, the customer must provide the following information to the Telephone Company at the time of ordering:
- Number of access links
- Link Type
- Signaling Link Code
- Customer Signaling Point Code
- Common Language Location Identifier (CLLI) code of the Telephone Company interconnecting Signal Transfer Point
- Contact telephone number for installation and maintenance of the customer's designated premises

When ordering CCS/SS7 Interconnection Service, the customer will provide an estimate of total annual volume and busy hour busy month volume projected for a period of three years. The forecast should be itemized by message type. The Telephone Company will utilize this forecast in its own efforts to project further facility requirements.

- For LIDB Validation Service, the customer shall provide a LIDB Validation Service Order Form which specifies the OPCs of the customer's designated OSSs sending the query or queries, the PIU per OPC of the customer's OSS location, and the desired due date of the order.

LIDB Validation Service is provided in conjunction with CCS/SS7 Interconnection Service, as set forth in Section 20. In order to utilize LIDB Validation Service, the customer must have CCS/SS7 Interconnection Service to the two Telephone Company STPs designated by the Telephone Company as the interconnecting STP pair to be utilized for interconnection to the CCS/SS7 network. The Telephone Company's STP Associations are provided in the National Exchange Carriers

AUG 2 6 1999

By 3 RS # 3.04 Public Service Commission MISSOURI

SEP 2 0 1393

THISSOURT

Issued: AUG 1 8 1993

(AT)

(A'T)

Ē

Pui die Sprvice Commission By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

Effective: SEP 2 0 1993

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 1st Revised Sheet 3.04 Replacing Original Sheet 3.04

ACCESS SERVICES

RECENED

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)
 - MAR 18 1983

5.2 Access Order-(Continued)

- Operator Service System (OSS) Tandem interconnection requirements are specified in Section 17, Paragraph 17.2, following. Information regarding OSS Tandem locations is contained in the Southwestern Bell Interexchange Customer Information Handbook.
 - For Common Channel Signaling/Signaling System 7 (CCS/SS7) Interconnection Service, the customer must provide the following information to the Telephone Company at the time of ordering:
 - Number of access links
 - Link Type
 - Signaling Link Code
 - Customer Signaling Point Code
 - Common Language Location Identifier (CLLI) code of the Telephone Company interconnecting Signal Transfer Point
 - Contact telephone number for installation and maintenance of the customer's designated premises

When ordering CCS/SS7 Interconnection Service, the customer will provide an estimate of total annual volume and busy hour busy month volume projected for a period of three years. The forecast should be itemized by message type. The Telephone Company will utilize this forecast in its own efforts to project further facility requirements.

CANCELLED

Commission Public Ser MISSOURI

5 M ())

MAY 0 1 1993

Issued: MAR 2 2 1993

Effective: MAY - 1 1993

MC. 292110 SERVICE COMM.

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

(MT)

(MT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

(AT)

Access Services Tariff Section 5 Original Sheet 3.04

RECEIVED

ACCESS SERVICES

MAY 11 1992

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MO. PUELIC SERVICE COMM

5.2 Access Order-(Continued)

When ordering CCS/SS7 Interconnection Service, the customer will provide an estimate of total annual volume and busy hour busy month volume projected for a period of three years. The forecast should be itemized by message type. The Telephone Company will utilize this forecast in its own efforts to project further facility requirements.

CANCELLED MAY 1 1393# BY Lat R.S. 2 Public Service Commission MISSOUR

FILED

NOV 16 1992

MO. PUBLIC SERVICE COMM.

Issued: Effective: Effective: NOV 1 6 1992 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri
Access Services Tariff Section 5 1st Revised Sheet 3.05 Replacing Original Sheet 3.05

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

The BHMC may be determined by the customer in the following manner. For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). The customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

Where the Special Access Service is exempt from the Special Access Surcharge as set forth in Paragraph 7.3.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order the certification as set forth in Paragraph 7.3.2, following.

For Frame Relay Service (FRS), the customer must specify the customer's designated premises, the Primary Market Area (PMA) involved and the type of access termination. For each access termination requested, the customer must specify the data transmission speed.

In addition, the customer must provide both points connected by a Logical Link, along with the data link connection identifier (DLCI), or address, assigned for each point when establishing the permanent virtual circuit.

When ordering from a point outside the PMA, the Telephone Company shall determine the frame relay switch(es) within the PMA as well as the nearest serving wire center within the PMA.

When FGD or BSA-D Switched Access Service is ordered for the provision of Multiple 64 CCC, the customer must have direct routed digital transport facilities between the customer designated premises and the Multiple 64 CCC end office for originating and terminating traffic. To ensure availability of transporting Multiple 64 CCC rates at speeds up to 1536 Kbps, the customer must, at a minimum, order 24 FGD or BSA-D trunks or contiguous increments of 24 FGD or BSA-D trunk groups, equipped with the following:

- SS7 Signaling - 64 CCC

- Multiple 64 CCC

Issued: October 17, 1994



AT)

(AT)

(MT)

(AT)

(AT)

Access Services Tariff Section 5 Original Sheet 3.05

ACCESS SERVICES

RECEIVED

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MAY 24 1994

5.2 Access Order-(Continued)

(MT) The BHMC may be determined by the customer in the following manner Commission For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). The customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

> Where the Special Access Service is exempt from the Special Access Surcharge as set forth in Paragraph 7.3.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order the certification as set forth in Paragraph 7.3.2, following.

For Frame Relay Service (FRS), the customer must specify the customer's designated premises, the Primary Market Area (PMA) involved and the type of access termination. For each access termination requested, the customer must specify the data transmission speed.

In addition, the customer must provide both points connected by a Logical Link, along with the data link connection identifier (DLCI), or address, assigned for each point when establishing the permanent virtual circuit.

When ordering from a point outside the PMA, the Telephone Company shall determine the frame relay switch(es) within the PMA as well as the nearest serving wire center within the PMA.

CANCELLED

NOV 2/21994 3 05 BY 101 R.S. T.3.05 Public Service Commission

JUN D T

JUL 1 5 1994

MISSOURI Public Service Commission JUL 1 5 1994

Issued: MAY 2 4 1994

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

Access Services Tariff Section 5 1st Revised Sheet 3.0501 Replacing Original Sheet 3.0501

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

In addition, the customer must specify one of three trunk allocation schemes: fixed, floating or flexible. In the fixed allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call are contiguous and the first FGD or BSA-D trunk is constrained to certain fixed starting points. In the floating allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call are contiguous, but the position of the first trunk can float. For the flexible allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call may occupy non-contiguous positions within a group of 24 FGD or BSA-D trunks.

Customers may segregate their originating and terminating Multiple 64 CCC traffic by specifying dedicated Multiple 64 CCC trunk group(s) on the order. A Multiple 64 CCC trunk group(s) represents access capacity for carrying only Multiple 64 CCC traffic.

For Advanced Carrier Identification Service (ACIS), the customer must have LATA-wide Switched Access in order to receive originating ACIS routed traffic. The customer must specify the:

- LATAs from which they wish to receive originating ACIS calls,
- the ACIS Access Code(s) to be activated in a given LATA,
- the Carrier Identification Code (CIC) of each ACIS Code in a given LATA, and
- the desired due date of the order.

Originating ACIS routed calls from LATAs in which the customer h as not ordered ACIS will be blocked.

Originating traffic that is to be routed using ACIS may be combined with either FGB, FGC, BSA-B, BSA-C or BSA-D Switched Access Service. The customer shall designate which FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service trunk groups are to be associated with ACIS. However, when FGD or BSA-D Switched Access Service becomes available in an end office, ACIS traffic originating from that end office must be provided with FGD or BSA-D Switched Access Service.

(AT)

Issued:

March 20, 1995

Effective: April 20, 1995

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

(AT)

Access Services Tariff Section 5 Original Sheet 3.0501

ACCESS SERVICES

RECEIVED

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

OCT 17 1994

5.2 Access Order-(Continued)

In addition, the customer must specify one of three Muniparlocation schemes: fixed, floating or flexible. In the fixed allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call are contiguous and the first FGD or BSA-D trunk is constrained to certain fixed starting points. In the floating allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call are contiguous, but the position of the first trunk can float. For the flexible allocation scheme, the FGD or BSA-D trunks selected for a Multiple 64 CCC call may occupy non-contiguous positions within a group of 24 FGD or BSA-D trunks.

Customers may segregate their originating and terminating Multiple 64 CCC traffic by specifying dedicated Multiple 64 CCC trunk group(s) on the order. A Multiple 64 CCC trunk group(s) represents access capacity for carrying only Multiple 64 CCC traffic.

CANCELLED

APR 201995 ommission Public Ser MISSOURI

<u>NOV 241994</u>

Issued: OCT 1 7 1994

Effective

By Horace Wilkins, Jr., President-Missouri Public Service Commission Southwestern Bell Telephone St. Louis, Missouri

Access Services Tariff Section 5 6th Revised Sheet 4 Replacing 5th Revised Sheet 4

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd)

5.2.1 Access Order Service Date Intervals

The time required to provision the service (i.e., the interval between the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

Schedules that specify installation intervals will also specify the services and quantities of the services that can be provided as specified in Section 5.2.2.1(B) and (C). The Telephone Company will adhere to the intervals as specified in Section 5.2.3(C)(1), except during circumstances beyond its direct control.

A schedule of intervals applicable for Expanded Interconnection is located in the Telephone Company's Technical Publication for Expanded Interconnection. The Telephone Company's intervals for equipment are subject to equipment availability from the manufacturer and the published intervals will be adhered to except:

- During circumstances beyond its direct control (i.e., acts of God, governmental requirements, work stoppages and civil commotions) and
- When an interconnector designates equipment that is not used by the Telephone Company to provision its other access services and the Telephone Company must, therefore, first obtain training regarding the installation, maintenance and administration of the equipment designated as well as obtain the equipment itself.

Access Services provided will be installed during Telephone Company business days. If an IC requests that installation be done outside of scheduled work hours, and the Telephone Company agrees to this request, the IC will be subject to applicable Additional Labor Charges as set forth in Section 13, following.

5.2.2 Access Order Charge

An Access Order Charge will apply per LIDB Validation Service Order for the installation, addition, change or rearrangement of LIDB Validation Service.

Issued: December 20, 2002

Effective: January 20, 2003

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

(RT)

(AT)

Access Services Tariff Section 5 5th Revised Sheet 4 Replacing 4th Revised Sheet 4

ACCESS SERVICES

RECEIVED

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) MAY 24 1994

5.2.1 Access Order Service Date Intervals

MISSOURI

The time required to provision the service (i.e., the Interval Detween the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

 (AT) A schedule of intervals applicable for Switched, Special Access and
 (AT) Frame Relay Services is located in the Southwestern Bell Interexchange Customer Information Handbook. The schedule specifies the services and quantities of these services that can be provided within the published interval.

> Access Services provided will be installed during Telephone Company business days. If an IC requests that installation be done outside of scheduled work hours, and the Telephone Company agrees to this request, the IC will be subject to applicable Additional Labor Charges as set forth in Section 13, following.

5.2.2 Access Order Charge

An Access Order Charge will apply per LIDB Validation Service Order for the installation, addition, change or rearrangement of LIDB Validation Service.

JAN 2 0 2003 on ission



JUL 1 5 1994

JMISSOUR!

Issued:

MAY 2 4 1994

Effective: 900-9-4 100

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone St. Louis, Missouri

Access Services Tariff Section 5 4th Revised Sheet 4 Replacing 3rd Revised Sheet 4

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 18 1993

5.2 Access Order-(Continued)

MISSOURI

Public Service Commission

RECEIVED

The BHMC may be determined by the customer in the following manner. For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

Where the Special Access Service is exempt from the Special Access Surcharge as set forth in Paragraph 7.3.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order the certification as set forth in Paragraph 7.3.2, following.

5.2.1 Access Order Service Date Intervals

The time required to provision the service (i.e., the interval between the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

A schedule of intervals applicable for Switched and Special Access Services is located in the Southwestern Bell Interexchange Customer Information Handbook. The schedule specifies the services and quantities of these services that can be provided within the published interval.

Access Services provided will be installed during Telephone Company business days. If an IC requests that installation be done outside of scheduled work hours, and the Telephone, Company agrees to this request, the IC will be subject to applicable odditional Labor Charges as set forth in Section 13, following.

(AT) 5.2.2 Access Order Charge

JUL 15139

An Access Order Charge will apply per LIDB valuation Service Order for the installation, addition, change of rearrangement of LIDB valuations Service.

Issued:

Effective: SEP 2 0-1993 Service Commission AUG 1 8 1993 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 3rd Revised Sheet 4 Replacing 2nd Revised Sheet 4

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 9 1991

5.2 Access Order-(Continued)

MISSOURI

Public Service Commission The BHMC may be determined by the customer in the following manner. For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). The customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

Where the Special Access Service is exempt from the Special Access Surcharge as set forth in Paragraph 7.3.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order the certification as set forth in Paragraph 7.3.2, following.

5.2.1 Access Order Service Date Intervals

The time required to provision the service (i.e., the interval between the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

A schedule of intervals applicable for Switched and Special Access Services is located in the Southwestern Bell Interexchange Customer Information Handbook. The schedule specifies the services and quantities of these services that can be provided within the published interval.

Access Services provided will be installed during Telephone Company business days. If an IC requests that installation be done outside of scheduled work hours, and the Telephone Company agrees to this request, the IC will be subject to applicable Additional Labor Charges as set forth in Section 13, following.

SEP 20 1993 rvice Commission FILED MISSO Public

Effective: **SEP 0 9 1991** SEP 3 0 1991

Issued: AUG 0 9 1991

By R. D. BARRON, President-Missouri Division Public Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(F()

(F())

Access Services Tariff Section 5 2nd Revised Sheet 4 Replacing 1st Revised Sheet 4 RECEIVED

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

Public Service Commission

Public Service Commission

The BHMC may be determined by the customer in the following manner. For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). The customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

Where the Special Access Service is exempt from the Special AGANGELLED charge as set forth in Paragraph 7.3.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order 1991 the certification as set forth in Paragraph 7.3.2, following. RV 3^{-1} R.S.*

5.2.1 Access Order Service Date Intervals

The time required to provision the service (i.e., the interval between the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

A schedule of intervals applicable for Switched and Special Access Services is located in the Southwestern Bell Interexchange Customer Information Handbook. The schedule specifies the services and quantities of these services that can be provided within the published interval.

Access Services provided will be installed during Telephone Company business days. If an IC requests that installation be done outside of scheduled work hours, and the Telephone Company agrees to this request, the IC will be subject to applicable Additional Labor Charges as set forth in Section 13, following.

FILED OCT 1 1989 89 - 14 Public Service Commission 0CT 1 Issued: SEP 2 5 1989 Effective: 1989 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 5 lst Revised Sheet 4 Replacing Original Sheet 4 DELVED

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSUAR

The BHMC may be determined by the customer in the following mainer minission For each day, the customer shall determine the highest number of minutes of use for a single hour (e.g., 55 minutes in the 10-11 a.m. hour). The customer shall, for the same hour period (i.e., busy hour), pick the 20 consecutive business days in a calendar year which add up to the largest number of minutes of use. The customer shall then determine the average busy-hour minutes of capacity (i.e., BHMC) by dividing the largest number of minutes-of-use figure for the same hour period for the consecutive 20-day period by 20. This computation shall be performed to establish the BHMC's for each end office the customer wishes to serve. These determinations thus establish the forecasted BHMC for each end office.

Where the Special Access Service is exempt from the Special Access Surcharge as set forth in Paragraph 7.4.2, following, and the customer desires an exemption from the surcharge, it shall furnish with the order the certification as set forth in Paragraph 7.4.2, following.

5.2.1 Access Order Service Date Intervals

The time required to provision the service (i.e., the interval between the Application Date and the Service Date) is known as the service interval. Such intervals will be established in accordance with published service date interval guidelines and, where possible, will reflect the customer's requested service date.

A schedule of intervals applicable for Switched and Special Access Services is located in the Southwestern Bell Interexchange Customer Information Handbook. The schedule specifies the services and quantities of these services that can be provided within the published interval.

Public Service Commission MISSOURI

Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff

Original Sheet 4

REGEIVED

MISCOURI

Public Service Commission

后礼国间

JAN - 1 1984

83-253

Section 5

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 29 1283

- 5.2 Access Order-(Continued)
 - 5.2.1 Access Order Service Date Intervals-(Continued)

To the extent the Access Service can be made available with reasonable effort, the Telephone Company will provide the Access Service in accordance with the IC's requested interval, subject to the following conditions:

A. Standard Interval

The Telephone Company shall publish and make available to all IC's, a schedule of Standard Intervals applicable for Switched and Special Access Services. This schedule shall specify which services and the quantities of services that can be provided within the standard intervals.

Access Services provided in a Standard Interval will be installed during normally scheduled Telephone Company work hours. If an IC requests that installation be done outside of scheduled work hours, and the Telephone Company agrees to this request, the IC will be subject to applicable Additional Labor Charges as set forth in Section 13, following.

B. Negotiated Interval

The Telephone Company will negotiate a service date interval with the IC when:

- 1. There is no Standard Interval for the service, or
- 2. The Short Notice Interval is not applicable, or
- 3. The IC requests a service date beyond the applicable Standard Interval service date.



DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd)

- 5.2 Access Order (cont'd)
 - 5.2.2 Access Order Charge (cont'd)
 - 5.2.2.1 Negotiated Interval

The Telephone Company will negotiate a service date interval with the customer when:

- A. There is no standard interval for the service, or
- B. The quantity of Access Services ordered exceeds the quantities specified in Section 5.2.3(C)(1), or
- C. The customer requests a service date beyond the applicable standard interval service date.

Standard Intervals	
Analog/Voice Grade/DS0	10 days
DS1	5 days

The Telephone Company will offer a service date based on the type and quantity of Access Services the customer has requested. The negotiated interval may not exceed by more than six months the standard interval service date, or, when there is no standard interval, the Telephone Company offered service date.

Issued: December 20, 2002

Effective: January 20, 2003

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Complete St. Louis, Missouri



ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.2 Access Order Charge-(Continued)

An Access Order Charge will apply per Access Order for the installation, addition, change or rearrangement of Switched Access Service. An Access Order Charge will also apply per Access Order for the installation, addition, change or rearrangement of Special Access Service and will apply to the addition, change or rearrangement of the WATS Access Line features, specified in Section 6, Paragraph 6.11, following, when requested subsequent to the installation of the WATS Access Line.

An Access Order Charge will apply, in addition to other applicable Access tariff charges, with the following exceptions:

- Nonchargeable Administrative Changes in 6.10.1, C.4.a, and where noted in 7.3.1, B.3., following;
- Access Order Modifications as specified in 5.2.3, following;
- Initial or subsequent activation of NXX's for 1+900 Access Service;
- Establishment of 0+900 Access Service
- Complete or partial disconnection of Access Service(s) and/or feature(s) (except when a Carrier Identification Code (CIC) is deleted);
- Upgrades from FGA, FGB, or FGC service to FGD service; and upgrades from BSA-A, BSA-B, BSA-C to BSA-D service.
- Discontinuance of Telecommunications Service Priority System.
- Conversions during the transition period of FGA to BSA-A, FGB to BSA-B, FGC to BSA-C and FGD to BSA-D.
- Change of Access Services Bill Period.
- As specified in Paragraph 6.10.1, C.5.g.

Switched Access Order Charge	<u>USOC</u>	<u>Charge</u>
Switched Access Order Charge, per order	NRBIX	\$16.00(1)
Special Access and Frame Relay Order Charge,		
per order	NRBIX	\$14.00(1)
LIDB Validation Service Access Order Charge, per LIDB		
Validation Service Order Form	NRBIX	\$19.00

(1) This nonrecurring charge will be waived when the charge is the result of the return of a Carrier Identification Code (CIC) to Bellcore.

Issued:	December 16, 1994	Effective:	January 16, 1995	FILED
	2	WILKINS, JR., Pre thwestern Bell Teler		MO PSC
	504	St. Louis, Missour	L	

Access Services Tariff Section 5 8th Revised Sheet 5 Replacing 7th Revised Sheet 5

ACCESS SERVICES

ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE- (Continued) 5.

DEC.

all. PEBERSON CARL

- 5.2 Access Order-(Continued)
- 5.2.2 Access Order Charge-(Continued)

An Access Order Charge will apply per Access Order for the installation, addition, change or rearrangement of Switched Access Service. An Access Order Charge will also apply per Access Order for the installation, addition, change or rearrangement of Special Access Service and will apply to the addition, change or rearrangement of the WATS Access Line features, specified in Section 6, Paragraph 6.11, following, when requested subsequent to the installation of the WATS Access Line.

An Access Order Charge will apply, in addition to other applicable Access tariff charges, with the following exceptions:

- Nonchargeable Administrative Changes in 6.10.1, C.4.a, and where noted in 7.3.1, B.3., following;
- Access Order Modifications as specified in 5.2.3, following;
- Initial or subsequent activation of NXX's for 1+900 Access Service;
- Establishment of 0+900 Access Service
- Complete or partial disconnection of Access Service(s) and/or feature(s) (except when a Carrier Identification Code (CIC) is deleted);
- Upgrades from FGA, FGB, or FGC service to FGD service; and upgrades from BSA-A, BSA-B, BSA-C to BSA-D service.
- Discontinuance of Telecommunications Service Priority System.
- Conversions during the transition period of FGA to BSA-A, FGB to CANCELLED BSA-B, FGC to BSA-C and FGD to BSA-D.

- As specified in Paragraph 6.10.1, C.5.g.

Switched Access Order Charge, per order

Special Access and Frame Relay Order Charge, per order

LIDB Validation Service Access Order Charge, per LIDB Validation Service Order Form

USOC JAN 151995 <u>Charge</u> 9 Th R. S. # 5 \$16.00(1) NRBIX Public Service Commission MISSOURI NRBIX \$14.00(1)

NRBIX

\$19.00

(1)	This nonrecurring of	charge will be	waived when	the charge is	the result	of the
	return of a Carrier	dentificatio	on Code (CIC)	to Bellcore.	JAN 6	1005
					JANU	1333

JAN 0 6 1995 MISSOURI DEC 0 6 1994 Effective:

By HORACE WILKINS, JR., President-Missouri Public Service Commission Southwestern Bell Telephone St. Louis, Missouri

(AT) (AT)

Issued:

Access Services Tariff

No supplement to this

tariff will be issued Section 5 except for the purpose 7th Revised Sheet 5 of canceling this tariff. Replacing 6th Revised Sheet 5 ACCESS SERVICES 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) ECEIVED 5.2 Access Order-(Continued) MAY 24 1994 5.2.2 Access Order Charge-(Continued) MISSOURI An Access Order Charge will apply per Access Order for the instantions addition, change or rearrangement of Switched Access Service. An Access Order Charge will also apply per Access Order for the installation, addition, change or rearrangement of Special Access Service and will apply to the addition, change or rearrangement of the WATS Access Line features, specified in Section 6, Paragraph 6.11, following, when requested subsequent to the installation of the WATS Access Line. An Access Order Charge will apply, in addition to other application (Netes) tariff charges, with the following exceptions: - Nonchargeable Administrative Changes in 6.10.1, C.4.a, 6 1995 JAN and where noted in 7.3.1, B.3., following; - Access Order Modifications as specified in 5.2.3, following; $\mathcal{A} R.S.5$ - Initial or subsequent activation of NXX's for 900 Access Service Commistant - Complete or partial disconnection of Access Service(s) AutoMorService Commistant feature(s) (except when a Carrier Identification Code (CIC) is MISSOURI deleted); - Upgrades from FGA, FGB, or FGC service to FGD service; and upgrades from BSA-A, BSA-B, BSA-C to BSA-D service. - Discontinuance of Telecommunications Service Priority System. - Conversions during the transition period of FGA to BSA-A, FGB to BSA-B, FGC to BSA-C and FGD to BSA-D. - As specified in Paragraph 6.10.1, C.5.g. USOC Charge Switched Access Order Charge, per order NRBIX \$16.00(1) (AT) Special Access and Frame Relay Order Charge, per order \$14.00(1) NRBIX LIDB Validation Service Access Order Charge, per LIDB Validation Service Order Form \$19.00 NRBIX (1) This nonrecurring charge will be waived when the charge is the result of the return of a Carrier Identification Code (CIC) to Bellcore. JUL 1 5 1994 Issued: 100) 1050UR 1994 Effective: MAY 2 4 1994 JUN des Commission By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

_

- -----

		nager-Regulatory & Í Bell Telephone Compan uis, Missouri	ndust: 1y	ry Relations U 1994 MO. PUBLIC SERVICE CON
Issued:		Effective: DEC JAN 2 0 1994	, 	
	is nonrecurring charge will be v turn of a Carrier Identification			
T)	LIDB Validation Service Access Order Charge, per LIDB, Validation Service Order Form	NRBIX		\$19.00
T)				
T)	Special Access Order Charge, per order	NR ALS SE PUDITE SE	IVICE (MISSO	Commission URI \$14.00(1)
T) T)		JUL	151 R.S.	Toreion
Τ)	Switched Access Order Charge, per order	NRBLX	CELLE	\$16.00(1)
T) T)	- As specified in Paragraph 6.	•		D Charge
T \	from BSA-A, BSA-B, BSA-C to - Discontinuance of Telecommun - Conversions during the trans BSA-B, FGC to BSA-C and FGD	BSA-D service. Dications Service Pri Sition period of FGA to BSA-D.	ority	System.
	feature(s) (except when a Ca deleted); - Upgrades from FGA, FGB, or I			
	 Nonchargeable Administrative and where noted in 7.3.1, B. Access Order Modifications a Initial or subsequent activa Complete or partial disconne 	3., following; as specified in 5.2.3 ation of NXX's for 90	, fol 0 Acc	lowing; ess Service;
	An Access Order Charge will an tariff charges, with the follo	wing exceptions:		
	addition, change or rearrangem Order Charge will also apply p addition, change or rearrangem apply to the addition, change features, specified in Section requested subsequent to the in	ment of Switched Acce oer Access Order for ment of Special Acces or rearrangement of 6, Paragraph 6.11,	ss Ser the in s Serv the W follow	rvice. An Access nstallation, vice and will ATS Access Line wing, when
	An Access Order Charge will ap			
	Access Order Charge-(Continued)	1)	i	MO. PUBLIC SERVICE COMM.
	ERING FOR SWITCHED AND SPECIAL A	COLDO DERVICE-(CONTI	nueu)	NOV 22 1993
5. ORD		SS SERVICES		
	eling this tariff.			Revised Sheet 5
	will be issued for the purpose		(th	Section 5 Revised Sheet 5

ļ

.....

- -

No supplement to this		Access	Services Tariff
tariff will be issued except for the purpose		5+6	Section 5 Revised Sheet 5
of canceling this tariff.	Re		Revised Sheet 5
5	ACCESS SERVICES	2	
5. ORDERING FOR SWITCHED AND	SPECIAL ACCESS SERVICE-	-(Continued)	RECEIVED
5.2 Access Order-(Continued)			AUG 18 1993
5.2.2 Access Order Charge-(MISSOURI blic Service Commission
	e will apply per Access earrangement of Switche		
installation, additio and will apply to the Access Line features,	e will also apply per A on, change or rearrangen addition, change or re specified in Section 6 uent to the installation	nent of Spec earrangement 5, Paragraph	ial Access Service of the WATS 6.11, following,
	e will apply, in additi the following exception		applicable Access
 Nonchargeable Admini in 7.3.1, B.3, follo 	strative Changes in 6.1	10.1, C.4.a,	and where noted
 Access Order Modific Initial or subsequen Complete or partial feature(s) (except w 	ations as specified in at activation of NXX's in disconnection of Access when a Carrier Identific	for 900 Acce s Service(s) cation Code	ss Se GANGELLED and/or (CIC) is
 Discontinuance of Te 	GB, or FGC service to BSA-C to BSA-D service. elecommunications Service the transition period of and FGD to BSA-D.	ce Priority ,	System to determinission
The Access Order Charg	ge for Switched Access	Service is:	HAN 20155 # 5
		SUC	BY CHISSOURI
Access Order Charge, per order	N	RBIX	\$16.00(1)
The Access Order Charg	ge for Special Access S	ervice is:	
Access Order Charge, per order	N	RBIX	\$14.00(1)
The Access Order Charg	ge for LIDB Validation	Service is:	
Access Order Charge, p Validation Service Ord		RBIX	(NR)5649209 (593
 This nonrecurring charge return of a Carrier Ident 	tification Code (CIC) t	o Bellcore.	MISSOURI
Issued: AUG 1 8 1993	Effective:	SEP 2 0 199	3
By A. D. ROBERTSON,	, Assistant Vice Presid hwestern Bell Telephone		l Affairs

St. Louis, Missouri

• •			P.S.C. MoNo. 36		
4		lement to this vill be issued		Acc ess Ser v	ices Tariff Section 5
		For the purpose			sed Sheet 5
	of cance	eling this tariff.		Replacing 3rd Revi	sed Sheet 5
			ACCESS SERVICES		
		ERING FOR SWITCHED AND S	PECIAL ACCESS SERVI	CE-(Continued)AR 1	3 1993
		ccess Order-(Continued)		NO. PULLES	LIVE COMM
	5.2.2	Access Order Charge-(C	continued)		
		An Access Order Charge addition, change or re			
		An Access Order Charge installation, addition and will apply to the Access Line features, when requested subsequ	, change or rearran addition, change of specified in Section	ngement of Special A r rearrangement of t on 6, Paragraph 6.11	ccess Service the WATS , following,
		An Access Order Charge tariff charges, with t			l icable Access
(RT)		 Nonchargeable Adminis in 7.3.1, B.3, follow Access Order Modifica Initial or subsequent 	ving; tions as specified activation of NXX	in 5.2.3, following 's for 900 Acc ess So	;; ervice;
		 Complete or partial d feature(s) (except wh deleted); 			
		- Upgrades from FGA, FG			upgrades
		from BSA-A, BSA-B, BS			
		 Discontinuance of Tel Conversions during the BSA-B, FGC to BSA-C a 	e transition perio	* *	
		The Access Order Charge is:	e for Switched Acce	ss Service, except	the DNAL BSA,
		Assess Orden Channe		USOC	Charge
		Access Order Charge, per order		NRBIX	\$16.00(1)
		The Access Order Charge	e for the DNAL BSA	and Special Access	Service is:
		Acc ess Order Charge, per order	CANCELLED	NRBIX	\$14.00(1)
					•1+100(1)
			SEP 20 1993 SEP 20 1993 Cth R.S. S	Ion	
		is nonrecurring charge	BY Service Commune		FILD
	(1) Thi	s nonrecurring charge	All bM vaived when	the charge is the	result of the MAY 0 1 1993
	Issued:	urn of a Carrier Ident: MAR 2 2 1993	Effectiv		
				MUCCERI T - YAM	Public Service Comm.
		By A. D. ROBERTSON, A South	vestern Bell Teleph		1113
			St. Louis, Missou	ıri	

r i

1

1

ļ

	tariff w except f	lement to this vill be issued for the purpose ling this tariff.		3rd	Services Tariff Section 5 Revised Sheet 5 Revised Sheet 5
	5 000T		ACCESS SERVICES	TE (Continued)	RECEIVED
		ERING FOR SWITCHED AND S	SPECIAL ACCESS SERVIO	.E-(Continued)	MAR 29 1993
	5.2 Ac	cess Order-(Continued)			MISSOURI
	5.2.2	Access Order Charge-(6	Continued)	Pu	blic Service Commissic.
		An Access Order Charge addition, change or re			
(FC)		An Access Order Charge installation, addition and will apply to the Access Line features, when requested subsequ	n, change or rearran addition, change or specified in Sectio	gement of Spec rearrangement n 6, Paragraph	<pre>ial Access Service of the WATS 6.11, following,</pre>
		An Access Order Charg tariff charges, with			applicable Access
(FC)		 Nonchargeable Administin 7.3.1, B.3, follow Access Order Modification Initial or subsequent Service; Complete or partial of feature(s) (except with deleted); 	wing; ations as specified t activation of NXX' disconnection of Acc	in 5.2.3, foll s for 800 or 9 ess Service(s)	owing; 00 Access and/or
(AT) (AT) (AT) (AT)		 Upgrades from FGA, F from BSA-A, BSA-B, B Discontinuance of Te Conversions during t BSA-B, FGC to BSA-C 	SA-C to BSA-D servic lecommunications Ser he transition period	e. vice Priority	System.
(AT)		The Access Order Charg is:	e for Switched Acces	s Service, exc	ept the DNAL BSA,
		Access Order Charge,		USOC	Charge
		per order		NRBIX	\$16.00(1)
(AT)		The Access Order Charg Access Order Charge,	INCEILED	-	cess Service is:
		per order	MAY 1 1393	NRBIX	FILED .00(1)
		. 	MAY 1 1393 BY TL R.S. 5 BY TL R.S. 5 Iblic Service Commiss MISEOURI	sion	PR 11 1995
_	re	is nonrecurring charge turn of a Carrier Ident	ification Code (CIC)	to Bellcore.	the result of the
	Issued:	MAR 2 6 1993	Effective]	APR 1 1 1993
		By A. D. ROBERTSON, South	Assistant Vice Presi western Bell Telepho St. Louis, Missour	ine Company	Affairs

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) RECEIVED

(MT)

MISSOURI Public Service Commission

AUG 9 1991

(AT) 5.2.2 Access Order Charge

5.2 Access Order-(Continued)

An Access Order Charge will apply per Access Order for the installation, addition, change or rearrangement of Switched Service.

An Access Order Charge will also apply per Access Order for the installation, addition, change or rearrangement of Special Access Service and will apply to the addition, change or rearrangement of the WATS Access Line features, specified in Section 6.8 following, when requested subsequent to the installation of the WATS Access Line.

An Access Order Charge will apply, in addition to other applicable Access tariff charges, with the following exceptions:

- Nonchargeable Administrative Changes in 6.7.1, C.4.a, and where noted in 7.3.1, B.3, following;
- Access Order Modifications as specified in 5.2.3, following;
- Initial or subsequent activation of NXX's for 800 or 900 Access Service;
- Complete or partial disconnection of Access Service(s) and/or feature(s) (except when a Carrier Identification Code (CIC) CANCELLED deleted);
- Upgrades from FGA, FGB, or FGC service to FGD service;
- Discontinuance of Telecommunications Service Priority System APR 11 1993 BY 3MRS.

The Access Order Charge for Switched Access Service is: Public Service Commissi

	USOC	MIRGOURI
Access Order Charge, per order	NRBIX	(NR) \$16.00(1)
The Access order Charge for Special	Access Service is:	
	USOC	Charge
Access Order Charge, per order	NRBIX	(NR) \$14.00(1)
(1) This nonrecurring charge will be waived return of a Carrier_Identification Code	when the charge is (CIC) to Bellcore.	the resulFILEDe
Issued: AUG 0 9 1991	ective: 6EP=0.9=	91 30 1991 SEP 3 0 1991

5 1991 SEP 3 0 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

Replacing 1st Revised Sheet 5

2nd Revised Sheet 5

Section 5

Access Services Tariff

Access	Services	; Tariff
	Se	ection 5
<u>lst</u>	Revised	Sheet 5
Replacing (Driginad	Sheet 5
- 18121	6121177	י וווק
		עורי אירי

MISSUGK. Public Service Commission

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.2 Access Order Modifications

The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of Special Access Service channels or Switched Access Service busy-hour minutes of capacity will be treated as a new Access Order (for the increase amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire ANDEL du pper authorizes the suggested order modifications, these changes will be made without order modification charges being incurred by the ICSE Howeyegg charges for the Special Access Service provided to the customer will BY 2 mil apply. **Public Service Commission**

A. Service Date Change Charge

Access Order service dates for the installation of new services or rearrangements of existing services may be changed, but the new service date may not exceed the original service date by more than 45 calendar days. When, for any reason, the customer indicates that service can not be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly detays the start of service, a Service Date Change Charge will apply [[If]] Design Change has been requested as set forth in 5.2.2.(C), following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by

86-84 Public Service Commission

MISSOURI

Issued: JUN 27 1986 Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff

ACCESS SERVICES

DEC 20 1000

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) South

Public Service Commission

5.2 Access Order-(Continued)

5.2.1 Access Order Service Date Intervals-(Continued)

B. Negotiated Interval-(Continued)

The Telephone Company will offer a service date based on the type and quantity of services the IC has requested. The negotiated interval may not exceed by more than 30 days the standard interval service date, or when there is no standard interval, the Telephone Company-offered service date.

When the negotiated interval is shorter than that which the Telephone Company originally offered, additional charges, including but not limited to Special Construction Charges and charges set forth in Section 13, following, for Additional Engineering and Additional Labor, may also apply.

- C. Short Notice Interval
 - 1. The following types of Access Service will be available to an IC on a Short Notice Interval:
 - a. Access Service required for short-term Regional, National and International Special Events, including sports and entertainment, news coverage and conventions, or

GANGELLED JUL 1 1986 FILED SERVICE COMMISSION OF MISSOUN JAM = 1 (SU) 89-253

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

5.

(AT)

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Access Services Tariff Section 5 6th Revised Sheet 5.01 Replacing 5th Revised Sheet 5.01

ACCESS SERVICES

RECEIVED ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MAY 24 1994

5.2.3 Access Order Modifications

MISSOURI Public Service Commission

The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of DNAL Channels, Special Access Service channels, Frame Relay Service (FRS) Links or Ports, CCS/SS7 STP Port Terminations, STP Access Connections, Switched Access Service busy-hour minutes of capacity or LIDB OPCs will be treated as a new Access Order (for the increased amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order modifications, these changes will be made without order modification charges being incurred by the IC. However, charges for the Special Access Service provided to the customer will apply.

Service Date Change Charge Α.

> Access Order service dates for the installation of new services or rearrangements of existing services may be changed, but the new service date may not exceed the original service date by more than 45 calendar days. When, for any reason, the customer indicates that service can not be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. The Application Date will not change as a result of a Service Date Change. If a Design Change has been requested as set forth in 5.2.3, C. following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more than 45 calendar days. In all other cases, if the customer-requested service date exceeds the allowable service date change period previously described, the order will be canceled by the Telephone JON SUPPOSITION SOURCE

> > St. Louis, Missouri

Effective:

(MT)

Issued:

MAY 2 4 1994 By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 5th Revised Sheet 5.01 Replacing 4th Revised Sheet 5.01

ACCESS SERVICES

ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) RECEIVED 5.

5.2 Access Order-(Continued)

MAR 17 1994

MISSOURI

5.2.3 Access Order Modifications

The customer may request a modification of its Access Order at any onmaission prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of DNAL Channels, Special Access Service channels, CCS/SS7 STP Port Terminations, STP Access Connections, Switched Access Service busy-hour minutes of capacity or LIDB OPCs will be treated as a new Access Order (for the increased amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order CANCELLED modifications, these changes will be made without order modification charges being incurred by the IC. However, charges for the Special

Α.

Access Order service dates for the installation of new spublices price Commission date may not exceed the original service date to be accepted f be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. The Application Date will not change as a result of a Service Date Change If a Design Change has been requested as set forth in 5.2.3, C., following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more than 45 calendar days. In all other cases, THE He 1994 customer-requested service date exceeds the allowable service date change period previously described, the order will be canceled by the Totes one Public Service Commission Company,

Issued: MAR 2 1 1994

APR 3 0 1994 Effective

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

(AT) (AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 4th Revised Sheet 5.01 Replacing 3rd Revised Sheet 5.01

ACCESS SERVICES

RECEIVED

AUG 18 1993

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

5.2.3 Access Order Modifications

The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. Ιf the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of DNAL Channels, Special Access Service channels, CCS/SS7 STP Port Terminations, STP Access Connections, Switched Access Service busy-hour minutes of capacity or LIDB OPCs will be treated as a new Access Order (for the increased amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order modifications, these changes will be made without order modification charges being incurred by the IC. However, charges for the Special Access Service provided to the customer will apply. APR 301994

Α. Service Date Change Charge

cth R. Access Order service dates for the installation of new service COMM rearrangements of existing services may be changed, but the new service UPI date may not exceed the original service date by more than 45 calendar days. When, for any reason, the customer indicates that service can not be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. If a Design Change has been requested as set forth in 5.2.3, C., following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more." than 45 calendar days. In all other cases, if the customer-requested service date exceeds the allowable service date change period previously gr described, the order will be canceled by the Telephone Company,

AUG 1 8 1993 Issued:

Effective SEP 2 0 1993 Public Service Commission

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

(RT) (AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 3rd Revised Sheet 5.01 Replacing 2nd Revised Sheet Sol **D**1

ACCESS SERVICES

ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5.

MISSOURI

MAR 29 1993

5.2 Access Order-(Continued)

Public Service Commission

5.2.3 Access Order Modifications

The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

(AT) Any increase in the number of DNAL Channels, Special Access Service channels, CCS/SS7 STP Port Terminations, STP Access Connections or Switched Access Service busy-hour minutes of capacity will be treated as a new Access Order (for the increased amount only).

> If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order modifications, these changes will be made without order modification charges being incurred by the IC. However, charges for the Special for the customer will apply.

A. Service Date Change Charge

SEP 20 1993 Access Order service dates for the installation of new services orth R.S rearrangements of existing services may be changed, but the new bervice commissed date may not exceed the original service date by more than 45 case of the service days. When, for any reason, the customer indicates that service the service the service the service that service the s be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. If a Design Change has been requested as set forth in 5.2.3, C., following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more than 45 calendar days. In all other cases, if the customer-requested service date exceeds the allowable service date change period previously described, the order will be canceled by the Telephone Company,

Issued: Effective: MAR 2 6 1993 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company APR 1 1 1995 St. Louis, Missouri 92-304 MO. PUBLIC SERVICE COMPL

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 2nd Revised Sheet 5.01 Replacing 1st Revised Sheet 5.01

ACCESS SERVICES

RECEIVED

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MAY 11 1992

NO. PUBLIC SERVICE COMMA.

BY 3MR.S. S.U.

- 5.2 Access Order-(Continued)
- 5.2.3 Access Order Modifications

The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

(AT) Any increase in the number of Special Access Service channels, CCS/SS7 STP
 (AT) Port Terminations, STP Access Connections or Switched Access Service busy-hour minutes of capacity will be treated as new Access Order (for the increase amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order modifications, these changes will be made without order modi**CANOELLED** charges being incurred by the IC. However, charges for the Special Access Service provided to the customer will apply. APR 11 1993

A. Service Date Change Charge

Access Order service dates for the installation of new services **Public Service Commiss** rearrangements of existing services may be changed, but the new service date may not exceed the original service date by more than 45 calendar days. When, for any reason, the customer indicates that service can not be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. If a Design Change has been requested as set forth in 5.2.3, C., following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more than 45 calendar days. In all other cases, if the customer-requested service date exceeds the allowable service date change period previously described, the order will be canceled by the Telephone Company, whether the service will be canceled by the Telephone Company.

Issued: WAY 1 3 1992

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company MO. PUBLIC SERVICE COMM. St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 1st Revised Sheet 5.01 Replacing Original Sheet 5.01

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

HED

AUG 9 1991

(FC)(MT) 5.2.3 Access Order Modifications

> The customer may request a modification of its Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The Telephone Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. If the modification cannot be made with the normal work force during normal business hours, the Telephone Company will notify the customer. If the customer still desires the Access Order modification, the Telephone Company will schedule a new service date. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of Special Access Service channels or Switched Access Service busy-hour minutes of capacity will be treated as new Access Order (for the increase amount only).

If the Telephone Company specifies that order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by a customer (e.g., the upgrading by the Telephone Company of the channel termination from effective two-wire to effective four-wire) and the customer authorizes the suggested order ANGE ED modifications, these changes will be made without order modification ED charges being incurred by the IC. However, charges for the Special Access Service provided to the customer will apply. NOV 16 1392 ervice Date Change Charge $R_{10} = \frac{1392}{R_{10}}$

A. Service Date Change Charge

Public Service Commission Access Order service dates for the installation of new services Sources rearrangements of existing services may be changed, but the new service date may not exceed the original service date by more than 45 calendar days. When, for any reason, the customer indicates that service can not be accepted for a period not to exceed 45 calendar days beyond the original service date, and the Telephone Company accordingly delays the start of service, a Service Date Change Charge will apply. If a Design Change has been requested as set forth in 5.2.3, C., following, and the engineering review cannot be completed within the 45-calendar-day time frame, the new service date may exceed the original service date by more than 45 calendar days. In all other cases, if the customer-requested service date exceeds the allowable service date change period previously described, the order will be cancelled by the Telephone Company,

Issued: **AUG 0 9** 1991

Effective: SEP 0 9 1991 SEP 3 0 1991 SEP 3 0 1991 By R. D. BARRON, President-Missouri Division Public Service Commissic: Southwestern Bell Telephone Company St. Louis, Missouri

(CT)

(CP)AC	CESS	SERVI	[CES
---	----	-----	------	-------	------

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.2 Access Order Modifications-(Continued)

A. Service Date Change Charge-(Continued)

more than 45 calendar days. In all other cases, if the customerrequested service date exceeds the allowable service date change period previously described, the order will be cancelled by the Telephone Company, appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.2(D), following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is CANCELLED

Service Date Change Charge, per Order

OMC

USOC

\$17BY <u>Lat R.S.S.</u> Public Service Commissio MISSOURI

Charge SEP 3 0 1991

B. Partial Cancellation Charge

Any decrease in the number of Special Access Service channels or Switched Access Service busy-hour minutes of capacity will be created as a partial cancellation, and the charges set forth in Paragraph 5.2.3, following will apply.

JUL 1 1986 86-84

Public Service Commission

Access Services Tariff

MISSUUK: Public Service Commissio

REGE

Original Sheet 5.01

Section 5

Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 5th Revised Sheet 5.02 Replacing 4th Revised Sheet 5.02

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd)

- 5.2 Access Order (cont'd)
 - 5.2.3 Access Order Modifications (cont'd)
 - A. Service Date Change Charge (cont'd)

Company, appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply will be in addition to the Service Date Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is;

USOC

	<u></u>	
Service Date Change Charge, per Order	OMC	\$13.00

(RT)

(RT)

Issued: December 20, 2002

Effective: January 20, 2003

Charge

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 4th Revised Sheet 5.02 Replacing 3rd Revised Sheet 5.02

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.3 Access Order Modifications-(Continued)

A. Service Date Change Charge-(Continued)

Company, appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is:

		
Service Date Change Charge,		
per Order	OMC	\$13.00

USOC

Charge

JUL 1 5 1994

MISSOURI

Public Service Commission

B. Partial Cancellation Charge

Any decrease in the number of DNAL channels, Special Access Service channels, FRS Links or Ports, CCS/SS7 STP Port Terminations, STP Access Connections, Switched Access Service busy-hour minutes of capacity or LIDB OPCs will be treated as a partial cancellation, and the charges Set forth in Paragraph 5.2.4, following, will apply.



Issued: MAY 2 4 1994

Effective: JUL 1 5 1994

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

MAY 24 1994

RECEIVED

MISSOURI

(MT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 3rd Revised Sheet 5.02 Replacing 2nd Revised Sheet 5.02

ACCESS SERVICES

RECEIVED

AUG 18 1993 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

5.2.3 Access Order Modifications-(Continued)

A. Service Date Change Charge-(Continued)

appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is:

	USOC	Charge
Service Date Change Charge,		
per Order	OMC	\$13.00

B. Partial Cancellation Charge

Any decrease in the number of DNAL channels, Special Access Service channels, CCS/SS7 STP Port Terminations, STP Access Connections, Switched Access Service busy-hour minutes of capacity or LIDB OPCs will be treated as a partial cancellation, and the charges set forth in Paragraph 5.2.4, following, will apply.

JUL_151994

Public Service Commission

TFR (#5.02)

SEP 2 0 1993

MISSOURI Public Service Commission

Issued: AUG 1 8 1993

(RT)

(AT)

Effective:SEP 2 0 1993

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 2nd Revised Sheet 5.02 Replacing 1st Revised Sheet 5.02

Charge

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

- 5.2 Access Order-(Continued)
- 5.2.3 Access Order Modifications-(Continued)
 - A. Service Date Change Charge-(Continued)

appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is:

Service Date Change Charge,		
per Order	OMC	\$13.00

(AT)

B. Partial Cancellation Charge

MAR 2 6 1993

Any decrease in the number of DNAL channels, Special Access Service channels, CCS/SS7 STP Port Terminations, STP Access Connections or Switched Access Service busy-hour minutes of capacity will be treated as a partial cancellation, and the charges set forth in Paragraph 5.2.4, following, will apply. CANCELLED

SEP 20 19:3 # 5.02 FILED ice Commission APR 11 1955 Public Ser 92 - 304 MO. PUBLIC SERVICE COMM Effective:

USOC

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

Issued:

RECEIVED MAR 29 1993

MISSOURI **Public Service Commission**

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 1st Revised Sheet 5.02 Replacing Original Sheet 5.02

ACCESS SERVICES

RECEIVED

MAY 11 1992

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)
- 5.2 Access Order-(Continued)

MO. PUBLIC SERVICE COMM

- 5.2.3 Access Order Modifications-(Continued)
- A. Service Date Change Charge-(Continued)

appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is:

	USOC	Charge
Service Date Change Charge,		
per Order	OMC	\$13.00

- B. Partial Cancellation Charge
- (AT) Any decrease in the number of Special Access Service channels, CCS/SS7 STP
 (AT) Access Connections or Switched Access Service busy-hour minutes of capacity will be treated as a partial cancellation, and the charges set forth in Paragraph 5.2.4, following, will apply.

CANCELLED

APR 11 1993 BY 2 R.S. 5.02 Public Service Commission MISSOUR Effective:

Issued: MAY 1 3 1992

By A. D. ROBERTSON, Assistant Vice President-External Affaigs/ 16 1992 Southwestern Bell Telephone Company St. Louis, Missouri MO. PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 5.02

ACCESS SERVICES

RECEIVED

AUG 9 1991 (RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

(FC)(MT) 5.2.3 Access Order Modifications-(Continued)

A. Service Date Change Charge-(Continued)

appropriate cancellation charges applied and a new order issued with the new service date. If a service date change is necessary to accommodate a customer-requested Design Change, both the Service Date Change Charge and the Design Change Charge apply.

A new service date may be established that is prior to the original service date if the Telephone Company determines it can accommodate the customer's request without delaying service dates for orders of other customers. If the service date is changed to an earlier date and the Telephone Company determines that additional labor or extraordinary costs are necessary to meet the earlier service date requested by the customer, the customer will be notified by the Telephone Company that an Expedited Order Charge, as set forth in 5.2.3, D., following, applies. Such charges to expedite service should they apply, will be in addition to the Service Date Change Charge.

A Service Date Change Charge will apply, on a per-order, per-occurrence basis, for each service date changed. The applicable charge is:

USOC

Service Date Change Charge,		
per Order	OMC	(CR)\$13.00

B. Partial Cancellation Charge

Any decrease in the number of Special Access Service channels or Switched Access Service busy-hour minutes of capacity will be treated as a partial cancellation, and the charges set forth in Paragraph 5.2.4, following, will apply.

CANCELLED

Charge

NOV 16 1992 BY Lot R.S. # 5.02 Public Service Commission MISSOURI

FILED

Issued: AUG 0 9 1991

SEP 0 9 1991 SEP SEP 3 0 1991 By R. D. BARRON, President-Missouri Division Public Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

(CT)

(CT)

MISSOURI

olic Service Commission

Access Services Tariff Section 5 4th Revised Sheet 6 Replacing 3rd Revised Sheet 6

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd)

5.2 Access Order (cont'd)

5.2.3 Access Order Modifications (cont'd)

B. Design Change Charge

The customer may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. An engineering review is a review by Telephone Company personnel of the service ordered and the requested changes, to determine what changes in the design, if any, are necessary to meet the changes requested by the customer. Design changes to a pending order include such things as the addition or deletion of features, BSEs, or functions or a change in the type of Transport Termination (Switched Access only), type of channel interface, type of interface group or technical specification package. Design changes do not include a change of customer premises, End User premises, End Office Switch, Feature Group type, BSA type, or Special Access Service channel type. Changes of this nature will require the issuance of a new order and the cancellation of the original order with appropriate cancellation charges applied.

The Telephone Company will review the requested change, notify the customer whether the change is a design change, if it can be accommodated and if a new service date is required. If the customer authorized the Telephone Company to proceed with the design change, a Design Charge will apply. The Design Charge Charge will apply on a per-order, per-occurrence basis, for each customer-issued order requiring a design change. The applicable charge is:

	<u>USOC</u>	Rate
Design Change Charge, per Order	H28	\$22.00

If a change of service date is required, the Service Date Change Charge as set forth in Paragraph 5.2.3, A., preceding, will also apply.

Issued: December 20, 2002

Effective: January 20, 2003

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

(FC)
Access Services Tariff Section 5 3rd Revised Sheet 6 Replacing 2nd Revised Sheet 6

ACCESS SERVICES

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)
- 5.2 Access Order-(Continued)

5.2.3 Access Order Modifications-(Continued)

C. Design Change Charge

The customer may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. An engineering review is a review by Telephone Company personnel of the service ordered and the requested changes, to determine what changes in the design, if any, are necessary to meet the changes requested by the customer. Design changes to a pending order include such things as the addition or deletion of features, BSEs, or functions or a change in the type of Transport Termination (Switched Access only), type of channel interface, type of interface group or technical specification package. Design changes do not include a change of customer premises, End User premises, End Office Switch, Feature Group type, BSA type, or Special Access Service channel type. Changes of this nature will require the issuance of a new order and the cancellation of the original order with appropriate cancellation charges applied.

The Telephone Company will review the requested change, notify the customer whether the change is a design change, if it can be accommodated and if a new service date is required. If the customer authorized the Telephone Company to proceed with the design change, a Design Charge will apply. The Design Change Charge will apply on a per-order, per-occurrence basis, for each customer-issued order requiring a design change. The applicable charge is:

	USOC	Rate
Design Change Charge,		
per Order	H28	\$22.00

If a change of service date is required, the Service Date Change Charge as set forth in Paragraph 5.2.3, A., preceding, will also apply.



(AT)

(AT)

Ó

RECEIVED

MISSOURI

Public Service Commission

MAR 291993

Access Services Tariff Section 5 2nd Revised Sheet 6 Replacing 1st Revised Sheet 6

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

MISSOURI

RECEIVED

AUG 9 1991

- (FC) 5.2.3 Access Order Modifications-(Continued)
 - C. Design Change Charge

The customer may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. An engineering review is a review by Telephone Company personnel of the service ordered and the requested changes, to determine what changes in the design, if any, are necessary to meet the changes requested by the customer. Design changes to a pending order include such things as the addition or deletion of features or functions or a change in the type of Transport Termination (Switched Access only), type of channel interface, type of interface group or technical specification package. Design changes do not include a change of customer premises, End User premises, End Office Switch, Feature Group type or Special Access Service channel type. Changes of this nature will require the issuance of a new order and the cancellation of the original order with appropriate cancellation charges applied.

The Telephone Company will review the requested change, notify the customer whether the change is a design change, if it can be accommodated and if a new service date is required. If the customer authorized the Telephone Company to proceed with the design change, a Design Charge will apply. The Design Change Charge will apply on a per-order, per-occurrence basis, for each customer-issued order requiring a design change. The applicable charge is:

	<u>USUC</u>	Kate
Design Change Charge,		
per Order	H28	(CR)\$22.00

(CT)

If a change of service date is required, the Service Date Change Charge as set forth in Paragraph 5.2.3, A., preceding, will also apply.

CANCELLED

Date

APR 11 1993 BY <u>3^{cr} R.5^{cr} C</u> Public Service Commission MISSOURI

Issued: AUG 0 9 1991

1 Effective: SEP 3 0 1991 EILED SEP 3 0 1995 SEP 3 0 1991 Southwestern Bell Telephone Company St. Louis, Missouri Public Service Commission

11000

(RT)

(CP) ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

	Access Services Tarif	-
	Section 1st Revised Sheet	6
	Replacing Original Sheet	6
	Replacing Original Sheet REGEIVED	
°CF 5 5	JUN 27 1985 SERVICE-(Continued)	;

MISSUUR;

Public Service Commissio

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SER

- 5.2 Access Order-(Continued)
- 5.2.2 Access Order Modifications-(Continued)
 - C. Design Change Charge

The customer may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. An engineering review is a review by Telephone Company personnel of the service ordered and the requested changes, to determine what changes in the design, if any, are necessary to meet the changes requested by the customer. Design changes to a pending order include such things as the addition or deletion of optional features or functions or a change in the type of Transport Termination (Switched Access only), type of channel interface, type of interface group or technical specification package. Design changes do not include a change of customer premises, End User premises, End Office Switch, Feature Group type or Special Access Service channel type. Changes of this nature will require the issuance of a new order and the caCANACELLED tion of the original order with appropriate cancellation charges applied. SEP 3 0 1991

The Telephone Company will review the requested change, notify $Bke_2 R$ customer whether the change is a design change, if it can Public Survice Commission dated and if a new service date is required. If the customer authorsource con rized the Telephone Company to proceed with the design change, a Design Charge will apply. The Design Change Charge will apply on a per-order, per-occurrence basis, for each customer-issued order requiring a design change. The applicable charge is:

Design	Change	Charge,	
per or	rder		



Issued: JUN 27 1986

Effective: JUL

1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell TElephone Company St. Louis, Missouri

Access Services Tariff Original Sheet 6

「同心医し

U/M ~ 1 1920 83 - 253

DEC 2 0 1003

ACCESS SERVICES

- ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 1.11SSOURI 5.
 - 5.2 Access Order-(Continued)
 - 5.2.1 Access Order Service Date Intervals-(Continued)
 - C. Short Notice Interval-(Continued)
 - 1. (Continued)
 - b. Access Service required for the Government for Command and Control Communications and National Security and Emergency Preparedness.
 - 2. Access Service required for the Government will be provided on the date requested, or as soon thereafter as possible, when the emergency falls into one of the following categories:
 - a. State of crisis declared by the National Command (includes commitments made to the "National Plan Authorities for Emergencies and Major Disasters"), or
 - b. Efforts to protect endangered U.S. personnel or property both in the U.S. and abroad (includes space vehicle recovery and protection efforts), or
 - Communications requirements resulting from hostile c. action, a major disaster or a major civil disturbance, or
 - d. The Director (Cabinet Level) of a Federal Department, Commander of a Unified/Specified Command, or Head of a Military Department has certified that a communications requirement is so critical to the protection of life and property or to the National Defense that it must be processed immediately, or
 - Political unrest A Adre Baldoult El which e. affects the National Interest. or
 - f. Presidential Service. JUL 1 1986

R.S. #6 SERVICE COMMISSION OF MISSOURI

Issued: DEC 29 1983 Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

C.

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

(FC)

(AT)

1

(AT)

Access Services Tariff Section 5 3rd Revised Sheet 7 Replacing 2nd Revised Sheet 7

Missouri Public Service Commission

RECD DEC 20 2002.

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd)

- 5.2 Access Order (cont'd)
 - 5.2.3 Access Order Modifications (cont'd)
 - Expedited Order Charge
 - (1) Analog (Metallic, Telegraph, and Wideband Analog), Voice Grade, DS0 (Wideband Data, MegaLink Data and DovLink) and DS1 Access Services

If a customer desires that service be provided on a due date earlier than the standard interval, the customer may request that service be provided on an expedited service interval. To qualify for an expedited interval the customer must provide End User premises access, where needed, until 11PM (CT), Monday-Friday.

If, upon reviewing availability of equipment, facilities and scheduled workload, the Telephone Company agrees to provide service on an expedited basis and the customer accepts this proposal, an Expedite Order Charge will apply.

The maximum number of circuits, which may be expedited, is limited to twelve (12) two-point or six (6) multi-point Analog/DS0 circuits at the same location; and a limit of four (4) DS1 circuits at the same location. When the number of access circuits exceeds the maximum threshold the interval will be negotiated and expediting will not be an option.

If the Telephone Company determines that service can be provided on an expedited basis, the following charges will apply based upon agreed upon expedited service interval. The Expedited Order Charge applies on a per order basis, regardless of the number of circuits on the order.

Missouri Public

FILED JAN 2 0 2003

Issued: December 20, 2002

Service Commission Effective: January 20, 2003

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 2nd Revised Sheet 7 Replacing 1st Revised Sheet 7

ACCESS SERVICES

RECEIVED

AUG 9 1991

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

(FC) 5.2.3 Access Order Modifications-(Continued)

MISSOURI Public Service Commission

D. Expedited Order Charge

If a customer desires that service be provided on an earlier date than that which has been established for the Access Order or the provision of the Access Service, the customer may request that service be provided on an expedited basis. If the Telephone Company determines that service can be provided on the requested date and that additional labor cost or extraordinary costs are required to meet the requested service date, the customer will be notified and will be provided with an estimate of the additional charges involved. If the customer instructs the Telephone Company to proceed, such additional charges will be determined and billed to the customer as follows:

To calculate the additional labor charges, the Telephone Company will keep track of the additional labor hours used to meet the request of the customer and will bill the customer at the applicable Additional Labor charges as set forth in 13.2.6, following.

To develop, determine and bill the customer for extraordinary costs which may be involved, the special construction terms and conditions as set forth in Section 14, following, will be used by the Telephone Company. Authorization to incur the costs and to bill the customer will be in accordance with the terms and conditions in Section 14, following.

When the request for expediting occurs subsequent to the issuance of the Access Order, a Service Date Change Charge as set forth in Paragraph 5.2.3, A., preceding, also applies.



Issued: AIIR 0 9 1991

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri Public S

Public Service Commiss

FILED

SEP 3 0 1991

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

5.

supplement to this	Access Dervices Tariri
iff will be issued	Section 5
ept for the purpose	lst Revised Sheet 7
canceling this tariff.	Replacing Original/Sheet 7
(CP)ACCESS SERVICES	
ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS	SERVICE-(Continued)
2 Access Order-(Continued)	MISSUUR:
.2.2 Access Order Modifications-(Continued)	Fublic Service Commission

..... Commisso Tomiff

D. Expedited Order Charge

5.2 Access Order-(Continued)

If a customer desires that service be provided on an earlier date than that which has been established for the Access Order or the provision of the Access Service, the customer may request that service be provided on an expedited basis. If the Telephone Company determines that service can be provided on the requested date and that additional labor cost or extraordinary costs are required to meet the requested service date, the customer will be notified and will be provided with an estimate of the additional charges involved. If the customer instructs the Telephone Company to proceed, such additional charges will bCANCELLED termined and billed to the customer as follows:

To calculate the additional labor charges, the Telephone Company will 3 0 1991 keep track of the additional labor hours used to meet the requesport 2nd the customer and will bill the customer at the applicable Applicable Commis-Labor charges as set forth in 13.2.6 following. MISSOURI

To develop, determine and bill the customer for extraordinary costs which may be involved, the special construction terms and conditions as set forth in Section 14, following, will be used by the Telephone Company. Authorization to incur the costs and to bill the customer will be in accordance with the terms and conditions in Section 14, following.

When the request for expediting occurs subsequent to the issuance of the Access Order, a Service Date Change Charge as set forth in Paragraph A., preceding, also applies.

1986 -6 Service Commission

Issued: JUN 27 1986 Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

DEC 29 %

Section 5

Original Sheet 7

Access Services Tariff

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5.

5.2 Access Order-(Continued)

i..!SSCURI Public Service Commission

- 5.2.1 Access Order Service Date Intervals-(Continued)
- C. Short Notice Interval-(Continued)
 - 3. For Short Notice Interval Access Orders, the nonrecurring charge is two times the appropriate nonrecurring charges for the service provided as set forth in other sections of this Tariff.
- 5.2.2 Access Order Modifications

An Access Order may be modified by the IC prior to the service date as set forth following. One or more of the following charges will apply when such modifications are undertaken. When modifications are undertaken, the Telephone Company may specify that the service date will be changed. All charges for Access Order modifications will apply on a per-occurrence basis.

Any increase in the number of Special Access Service channels or Switched Access Service busy-hour minutes of capacity will be treated as a new Access Order (for the increase amount only).

If order modifications are necessary to satisfy the transmission performance for a Special Access Service ordered by an IC, these changes will be made without order modification charges being incurred by the IC.

Α. Service Date Change Charge

> Access Order service dates may be changed, but the new service date may not exceed the original service date by more than 30 calendar days. If the ICrequested service date is more than 30 calendar days after the original service date, the order will be cancelled by the Telephone Company and reissued with the appropriate cancellation charges applied. With the agreement of the Telephone Company, a new service date may be established that is prior to the original standard or negotiated interval service date.

A Service Date Change Charge SIL uppy, on a per-order, per-occurrence basis, for each service date changed. The applicable charge $\overline{is}:|L| \in [U]$ JUL 1 1986 JAN - 1 1931 PUBLIC SERVICE COMMISSION

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Public Service Comm

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

OF MISSOURI

No Supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 7.01

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd) Missouri Public Service Commission

- 5.2 Access Order (cont'd)
 - 5.2.3 Access Order Modifications (cont'd)

RECD DEC 20 2002

- C. Expedited Order Charge (cont'd)
 - (1) Analog (Metallic, Telegraph, and Wideband Analog), Voice Grade, DS0 (Wideband Data, MegaLink Data and DovLink) and DS1 Access Services (cont'd)

	Analog/Voice Grade/DS0 Access Services					
<u>USOC</u>	Expedited Service Intervals	Expedited Order Charge				
EODXN	9 days	\$375.00				
EODXL	8 days	425.00				
EODXJ	7 days	475.00				
EODXG	6 days	525.00				
EODXE	5 days	575.00				
EODXC	4 days	625.00				
EODXA	3 days	675.00				
	DS1 Acces	<u>s Services</u>				
	<u>DST Acces</u>	<u>s Services</u>				

<u>USOC</u>	Expedited Service Intervals	Expedited Order Charge
EODXR	4 days	\$625.00
EODXP	3 days	675.00

- In addition to expedited order charges, special construction charges may apply, if the a. Telephone Company determines that additional cost will be incurred.
- When the request for expediting occurs subsequent to the issuance of the Access Order, a b. Service Date Change Charge as specified in (B), preceding, also applies.
- If the Telephone Company is subsequently unable to meet an agreed upon expedited ¢. service date, no Expedited Order Charge will apply, unless the missed service date was caused by customer.

Missouri Public

Issued: December 20, 2002

EFEED: JA48-20299103

By CINDY BRINKLEY, President-Missouri

Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company Compa

St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 Original Sheet 7.02

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE (cont'd) Missouri Public Service Commission

- 5.2 Access Order (cont'd)
 - 5.2.3 Access Order Modifications (cont'd)

REC'D DEC 20 2002

- C. Expedited Order Charge (cont'd)
 - (2) For all Access Service, excluding Analog, Voice Grade, DS0 and DS1 Access Services

If the customer desires that service be provided on an earlier date than that which has been established for the Access Order or the provision of the Access Service, the customer may request that service be provided on an expedited basis. If the Telephone Company determines that service can be provided on the requested date and that additional labor cost or extraordinary costs are required to meet the requested service date, the customer will be notified and will be provided with an estimate of the additional charges involved. The total charge to the customer for the Additional Engineering may not exceed the estimate amount by more than 10%. If the customer instructs the Telephone Company to proceed, such additional charges will be determined and billed to the customer as follows:

To calculate the additional labor charges, the Telephone Company will, upon authorization from the customer to incur the additional labor charges, keep track of the additional labor hours used to meet the request of the customer and will bill the customer at the applicable Additional Labor charges as set forth in 13.2.6, following.

Missouri Public

FILED JAN 2 0 2003

Service Commission

Issued: December 20, 2002

Effective: January 20, 2003

By CINDY BRINKLEY, President-Missouri Southwestern Bell Telephone, L.P., d/b/a Southwestern Bell Telephone Company St. Louis, Missouri

(FC)

(MT)

(MT)

(RT)

(RT)

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.4 Cancellation of An Access Order

(CT)	A.	A customer may cancel an Access Order at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The cancellation date is the date the
(CT)		Telephone Company receives written or verbal notice from the customer that the order is to be cancelled. The verbal notice must be followed by written confirmation within 10 days.
(AT)	B.	When Cancellation Charges Apply
(CT) (RT) (CT) 		If a customer or the customer's end user is unable to accept Access Service and the new service date requested is beyond the allowable service date change time period described in Section 5.2.3, A., preceding, the Access Order will be cancelled. When the customer cancels an access order on or after the application date, a Cancellation Charge will apply as specified below in addition to any other applicable charges specified in Paragraph 5.2.3.
		1. For all Access Services, the Cancellation Charge equals:
		The number of business days from the access order application date through the access order cancellation date (i.e., the service interval)
		Multiplied by the average daily charge
		Plus the access order charge.
		The service interval is the number of business days from the access order application date through the access order cancellation date with the application date being day 1. Service installation costs incurred by the Telephone Company start on the application date.
		If the customer has requested a service date change beyond the original service date, the number of business days beyond the original service date are included in the service interval.
 (CT)		Average daily charge equals installation charges plus rearrangement charges divided by the number of business days in the service interval.
	Issued:	March 21, 1994 Effective: April 30, 1994

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 2nd Revised Sheet 8 Replacing 1st Revised Sheet 8

ACCESS SERVICES

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

Issued: AUG 0 9 1991

- 5.2.4 Cancellation of An Access Order
- (RT) An IC may cancel an Access Order at any time prior to the service date or Α. notification by the Telephone Company that service is available for the customer's use, whichever is later. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the order is to be cancelled. The verbal notice must be followed by written confirmation within 10 days.

If an IC or its End User is unable to accept Access Service and the new IC requested service date is beyond the allowable service date change time period described in Section 5.2.3, A., preceding, the Access Order will be cancelled and the charges set forth in Paragraph 5.2.4, B., following will apply.

- (AT) Β. When a customer cancels an Access Order for the installation or rearrangement of service, a Cancellation Charge will apply as follows:
 - 1. Costs incurred by the Telephone Company start on the Application Date. The Application Date is described in Paragraph 5.1, preceding. This date is also known as the order date.
 - 2. Where the customer cancels an Access Order prior to the Application Date, no charges shall apply.
 - 3. When the customer cancels an Access Order on or after the Application Date, a cancellation charge will be calculated as follows:

CANCELLED

APR 301994 BY 3nd R.S. #8 Public Service Commission MISSOURI

Effective: SEP 0 9 1991 FILED By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Public Service Commis

SEP 3 0 1991

RECEIVED

AUG 9 1991

MISSOURI

^Dublic Service Commission

(CT)

(FC)

(RT) (RT)

Access Services Tariff Section 5 1st Revised Sheet 8 Replacing Original_Sheet 8 Replacing GFIWED

MISSUUR!

Public Service Commission

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) JUN 27 1960

5.2 Access Order-(Continued)

5.2.3 Cancellation of an Access Order

A. An IC may cancel an Access Order for the installation of service at any time prior to the service date or notification by the Telephone Company that service is available for the customer's use, whichever is later. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the order is to be canceled. The verbal notice must be followed by written confirmation within 10 days.

If an IC or its End User is unable to accept Access Service and the new IC requested service date is beyond the allowable service date change time period described in Section 5.2.2, A., preceding, the Access Order will be cancelled and the charges set forth in Paragraph B., following will apply.

- B. When a customer cancels an Access Order for the installation of service, a Cancellation Charge will apply as follows:
 - Costs incurred by the Telephone Company in conjunction with the installation of Switched or Special Access Service, start on the Application Date. The Application Date is described in Paragraph 5.1, preceding. This date is also known as the order date.
 - (2) Where the customer cancels an Access Order prior to the Application Date, no charges shall apply.

(3) When the customer cancels an Access Order on or after the Application Date, a cancellation charge will be calculated as follows:



CANCELLED

SEP 3 0 1991 BY 2 ~ R.S.

Public Service Commission MISSOURI

Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 8

> EISSOURI Public Service Commission

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWIICHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.2 Access Order Modifications-(Continued)

A. Service Date Change Charge-(Continued)

	USOC	Charge
Service Date Change Charge, per order	OMC	\$29.59

If the service date is changed to an earlier date, an Expedited Order Charge as set forth in Paragraph 5.2.2, D., following, will apply in addition to the Service Date Change Charge.

B. Partial Cancellation Charge

Any decrease in the number of Special Access Services, Switched Access Service busy-hour minutes of capacity or Dedicated Access Lines will be treated as a partial cancellation.

An IC may cancel any number of Special Access Services. For Switched Access Service busy-hour minutes of capacity, the amount cancelled cannot cause the order to drop below the minimum requirements as set forth in Paragraph 5.1.1, preceding, and Paragraph 5.5, following. When the capacity cancelled brings the ordered capacity below these minimum requirements, the entire order will be cancelled.

A Cancellation Charge will apply on a per-order, per-occurrence basis for the quantity cancelled. The applicable Cancellation Charge is computed as set forth in Paragraph 5.2.3, following.

C. Design Change Charge

The IC may request a design change to the service ordered. A design change is any change to an Access Order which requires engineering review. A design change may include such things as the addition or deletion of optional features or functions or a change in the type of Transport Termination (Switched Alleess only), type of facility interface, type of facility or type of IC terminal location interface group. It may not include a change of IC terminal location, End User's premises, end office switch or Feature Group type. JAN - 1 1934

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

2 GR 8,3 - 2,53

Ey R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

BLIC SERVICE COMMISSION

Access Services Tariff Section 5 3rd Revised Sheet 9 Replacing 2nd Revised Sheet 9

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

- 5.2 Access Order-(Continued)
 - 5.2.4 Cancellation of An Access Order-(Continued)
- C. When Cancellation Charges Do Not Apply
 - 1. When a customer cancels an order for the discontinuance of service, no charges apply for the cancellation.
 - 2. When a customer cancels an access order prior to the application date, no charges apply for the cancellation.
 - 3. If the Telephone Company or the customer misses a service date by m ore than 30 days, due to circumstances over which it has no direct control (i.e., acts of God, governmental requirements, work stoppages and civil commotions), the customer may cancel the access order without incurring Cancellation Charges.

(CT)

(CT)

Issued: March 21, 1994

Effective:



)	No supplement to this tariff will be issued except for the purpose of canceling this tariff.			Replac		2nd	Revise	es Tariff Section 5 d Sheet 9 d Sheet 9	
		ł	ACCESS SERVICES				RECE	IVED	
(RT)	5. ORDERING FOR SWITCHED AND		CIAL ACCESS SERVIC	CE-(Con	ntinu	ed)	AUG 9	1991	
(FC)	5.2 Access Order-(Continued 5.2.4 Cancellation of an A B. (Continued)		s Order-(Continued	1)	[.] ?uł	olic	MISS(Service	ouri Commissia	Dn
	3. (Continued)								
(AT)	Installation Charge(s) and/or Rearrangement Charge(s)	÷	Number of Business Days in the Installa- tion Interval	=	Aver Dail Char	Ly			
(CP)	· · · · · · ·	x	Average Daily Charge	+	Acce Orde Chai	er	=	Cancella- tion Charge	
	C. When an IC cancels an	orde	r for the discont	inuanc	e of	ser	vice. r	no charges	

- When an IC cancels an order for the discontinuance of service, no charges apply for the cancellation.
- D. If the Telephone Company misses a service date by more than 30 days due to circumstances over which it has direct control (excluding, e.g., acts of God, governmental requirements, work stoppages and civil commotions), the customer may cancel the Access Order without incurring cancellation charges. In addition, if the customer misses a service date by more than 30 days due to circumstances over which it has no direct control (i.e., acts of God, governmental requirements, work stoppages and civil commotions), the customer may cancel the Access Order without incurring cancellation charges.

CANCELLED

APR 301994 3rd R.S. Public Service Commission MISSOUH

Issued: AUG 0 9 1991

Effective: grp.0-9-1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

SEP 3 0 1991

11

ILED

No supplement to this tariff will be issued except for the purpos of canceling this tar	l ;e	Access Services Tariff Section 5 1st Revised Sheet 9 Replacing Original Sheet 9
	(CP)ACCESS SERVICES	REGEIVED
5. ORDERING OPTIONS	FOR SWITCHED AND SPECIAL ACCES	SS SERVICE-(Continued)
5.2 Access Order-(C	Continued)	, JUN 2 7 1968
5.2.3 Cancellation	n of an Access Order-(Continued	a) MISSUURI Auctic Service Jormalission
B. (Continued)		
3. (Continued	1)	
Number o from Oro	on Charge : Number of Business in the Installati Interval Average Daily Charge of Business Days der Application x Average Dai rough Cancella- te Cancellation Charge	ion = CANCELLED SEP 3 0 1991
	ncels an order for the discont: cancellation.	inuance of service, no charges
to circumstand of God, govern the customer n charges. In a than 30 days o (i.e., acts of	one Company misses a service da ces over which it has direct co nmental requirements, work stop nay cancel the Access Order with addition, if the customer misse due to circumstances over which f God, governmental requirement the customer may cancel the Acce charges.	ontrol (excluding, e.g., acts ppages and civil commotions), thout incurring cancellation es a service date by more h it has no direct control ts, work stoppages and civil



Issued: JUN 27 1986

1

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 9

FIISSOURE

JAN - 1 1930

Public Service Commission

ACCESS SERVICES

- 5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) (SEC)
 - 5.2 Access Order-(Continued)

5.2.2 Access Order Modifications-(Continued)

C. Design Change Charge-(Continued)

Changes of this nature will require the issuance of a new order and the cancellation of the original order with appropriate charges applied. The Design Change Charge will apply to all Special Access Service, channels or Switched Access Service busy-hour minutes of capacity.

The Telephone Company will review the requested change, notify the IC whether the change can be accommodated and specify if a new service date is required. If the IC authorizes the Telephone Company to proceed with the design change, a Design Change Charge will apply. If the change involves the addition or deletion of optional features for which nonrecurring charges are stated, the Design Change Charge is equal to one-half the nonrecurring charge for each feature being added or deleted. When there is no nonrecurring charge associated with the change being made, the Design Change Charge is equal to one-half of the nonrecurring charges for the Access Service ordered. If multiple changes are made on a single order, some of which have separately stated nonrecurring charges and others which do not, both of the above charges will apply. The Design Change Charge will apply on a per-order, per-occurrence basis.

If a change of service date is required, the Service Date Change Charge as set forth in Paragraph 5.2.2, A., preceding, will also apply.

D. Expedited Order Charge

When placing an Access Order, an IC may request a service date that is prior to the standard interval service date. An IC may also request an earlier service date on a pending standard or negotiated interval Access Order. If the Telephone Company agrees to provide service on the requested date, an Expedited Order Charge will apply.



Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

- 5.2 Access Order-(Continued)
 - 5.2.5 Selection of Planned Facilities for Access Orders
 - A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, or when there are digital high-capacity facilities to an interconnecting Signal Transfer Point (STP) for CCS/SS7 Interconnection Service, the customer must specify a channel assignment for each service ordered.
 In addition, when a customer requests a Rollover, the Connecting Facility Assignments (CFA's) must be specified for the facilities involved in the Rollover for each service. The customer will provide this information to the Telephone Company during the order process.
 - B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
 - 5.2.6 Minimum Period

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

CT)

CT)

(RT)

(RT)

The minimum period for which Access Service is provided and for which charges are applicable is set forth in each section of this tariff.

When Access Service is disconnected prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory. This terminology does not refer to when billing is stopped, but rather distinguishes a disconnect from a service rearrangement.

Service arrangements may be made without a change in minimum period requirements where so specified in this tariff.

September 22, 1994 Effective:

October 1, 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

5.2 Access Order-(Continued)

Access Services Tariff Section 5 6th Revised Sheet 10 Replacing 5th Revised Sheet 10

ACCESS SERVICES

RECEIVED

ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5.

FF9 17 1994

MISSOURI

5.2.5 Selection of Planned Facilities for Access Orders Public Service Commission

- A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, or when there are digital high-capacity facilities to an interconnecting Signal Transfer Point (STP) for CCS/SS7 Interconnection Service, the customer must specify a channel assignment for each service ordered. In addition, when a customer requests a Rollover, the Connecting Facility Assignments (CFA's) must be specified for the facilities involved in the Rollover for each service. The customer will provide this information to the Telephone Company during the order process.
- For all other Access Orders, the option to request a specific transfedsion path or channel is not provided, except as provided for under Nobectal В. Facilities Routing as set forth in Section 11, following. OCT 011994

5.2.6 Minimum Period

(AT)

(AT)

- nmission Except as set forth in Paragraph 9.4, A., following, the minimum para Α. for which Access Service is provided and for which charges Bars and Confi is one month. For the application of minimum period charges to MSSQ ched Access Service FGB, FGC, FGD, BSA-B, BSA-C and BSA-D, it is assumed that the last identical capacity placed in service is the first one discontinued.
- Β. Service Rearrangements as set forth in 6.10.1, C.4, and 7.3.1, B.3. following, may be made without a change in minimum period requirements.
- 5.2.7 Minimum Period Charges

When Access Service is discontinued prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- For Switched Access Service, the charge for a month or fraction Α. thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.10.4, following
- For Special Access Service, the charge for a month or function Β. thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.4, following. MAR 201993

Effective: MAR 2 0 1991 Issued: FEB 1 8 1994 By M. H. SCHULTEIS, Division Manager-Regulatory & Industry, Relyappondial By M. H. SCHULTEIS, Division Manager-Regulatory & Industry, Relyappondial Dic Service Commission St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 6th Revised Sheet 10 Replacing 5th Revised Sheet 10

· · · ·

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

DEC 23 1993

RECEIVED

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

- 5.2.5 Selection of Planned Facilities for Access Orders
- A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, or when there are digital high-capacity facilities to an interconnecting Signal Transfer Point (STP) for CCS/SS7 Interconnection Service, the customer must specify a channel assignment for each service ordered. This information will be provided to the Telephone Company during the order process.
- B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
- 5.2.6 Minimum Period

(CT)

(CT)

(RT)

RT)

The minimum period for which Access Service is provided and for which charges are applicable is set forth in each section of this tariff.

When Access Service is disconnected prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory. This terminology does not refer to when billing is stopped, but rather distinguishes a disconnect from a service rearrangement.

Service rearrangements may be made without a change in minimum period requirements where so specified in this tariff.

CANCELLED

ommission BY Public Ser SSOUR

JAN 1 1994 93 - 224

MISSOURI Public Service Commission

Issued: DEC 2 3 1993

Effective: JAN 0 1 1994

By M. H. SCHULTEIS, Division Manager-Regulatory and Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 5th Revised Sheet 10 Replacing 4th Revised Sheet 10

ACCESS SERVICES

RECEIVES

- 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) MAR 2 - 1993
- 5.2 Access Order-(Continued)

MISSOU~

- 5.2.5 Selection of Planned Facilities for Access Orders Public Service Communation,
 - A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, or when there are digital high-capacity facilities to an interconnecting Signal Transfer Point (STP) for CCS/SS7 Interconnection Service, the customer must specify a channel assignment for each service ordered. This information will be provided to the Telephone Company during the order process.
 - B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
- 5.2.6 Minimum Period
- A. Except as set forth in Paragraph 9.4, A., following, the minimum period for which Access Service is provided and for which charges are applicable is one month. For the application of minimum period charges for Switched Access Service FGB, FGC, FGD, BSA-B, BSA-C and BSA-D, it is assumed that the last identical capacity placed in service is the first one CANCELLED discontinued.
- в. Service Rearrangements as set forth in 6.10.1, C.4, and 7.3.1, B.3, following, may be made without a change in minimum period requirenges. Th R.S. #10
 - 5.2.7 Minimum Period Charges

When Access Service is discontinued prior to the expiration US Commission minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- Α. For Switched Access Service, the charge for a month or fraction thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.10.4, following.
- For Special Access Service, the charge for a month or fraction Β. thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.4, following.

Issued:

Effective: APR 1 1 1998 -MAR 2 6 1993 By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri MO. PUBLIC SERVICE COMM

(AT)

(FC)

(FC)

Access Services Tariff Section 5 4th Revised Sheet 10 Replacing 3rd Revised Sheet 10

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MAY 11 1992

MO. PULLIC SERVICE COMM.

BY STAR.S. #

RECEIVED

5.2 Access Order-(Continued)

5.2.5 Selection of Planned Facilities for Access Orders

- A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, or when there are digital high-capacity facilities to an interconnecting Signal Transfer Point (STP) for CCS/SS7 Interconnection Service, the customer must specify a channel assignment for each service ordered.
- This information will be provided to the Telephone Company during the order process.
- B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
- 5.2.6 Minimum Period
- A. Except as set forth in Paragraph 9.4, A., following, the minimum period for which Access Service is provided and for which charges are applicable is one month. For the application of minimum period charges for Switched Access Service FGB, FGC, and FGD it is assumed that the last identical capacity placed in service is the first one discontinued.
- B. Service Rearrangements as set forth in 6.7.1, C.4, and GAN, CELLED following, may be made without a change in minimum period requirements.
- 5.2.7 Minimum Period Charges

When Access Service is discontinued prior to the **Peppic Babics** Commission minimum period, charges are applicable for the balance of the Babics of the balance of the bala

The Minimum Period Charge for monthly billed services will be determined as follows:

- A. For Switched Access Service, the charge for a month or fraction thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.7.4, following.
- B. For Special Access Service, the charge for a month or fraction thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.4, following.

Issued:	MAY 1	3	1992	Effective: DET 2 1992	19 2
	By A.	D.	ROBERTSON,	Assistant Vice President-External Affairs 16 1992	
			Southwe	St. Louis, Missouri MO. PUBLIC SERVICE COMM.	

(AT) (AT)

Access Services Tariff Section 5 3rd Revised Sheet 10 Replacing 2nd Revised Sheet 10

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 9 1991

5.2 Access Order-(Continued)

MISSOURI

NOV 16 1992

- (FC) 5.2.5 Selection of Planned Facilities for Access Orders ublic Service Commission
 - A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, the customer must specify a channel assignment for each service ordered. This information will be provided to the Telephone Company during the order process.
 - B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Facilities Routing as set forth in Section 11, following.

(FC) 5.2.6 Minimum Period

- A. Except as set forth in Paragraph 9.4, A., following, the minimum period for which Access Service is provided and for which charges are applicable mission is one month. For the application of minimum period charges for Switched Access Service Feature Groups B, C and D, it is assumed that the last identical capacity placed in service is the first one discontinued.
- (CT) B. Service Rearrangements as set forth in 6.7.1, C.4, and 7.3.1, B.3, following, may be made without a change in minimum period requirements.
- (FC) 5.2.7 Minimum Period Charges

When Access Service is discontinued prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- A. For Switched Access Service, the charge for a month or fraction thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.7.4, following.
- B. For Special Access Service, the charge for a month or fraction thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.4, following.

Issued: **XUG 0 9 1991**

Effective SEP 0 9 1991 LED SEP 3 0 1995 EP 3 0 1991

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commisting St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 2nd Revised Sheet 10 Replacing 1st Revised Sheet 10

ACCESS SERVICES

- 5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE (Gontinued)
- 5.2 Access Order-(Continued)

SEP 2 5 1989

5.2.4 Selection of Planned Facilities for Access Orders

- A. When there are analog or digital high-capacity facilities to the firstor point of switching for switched access or to a hub for special access, the customer must specify a channel assignment for each service ordered. This information will be provided to the Telephone Company during the order process.
- B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
- 5.2.5 Minimum Period
 - A. Except as set forth in Paragraph 9.4, A., following, the minimum period for which Access Service is provided and for which charges are applicable is one month. For the application of minimum period charges for Switched Access Service Feature Groups B, C and D, it is assumed that the last identical capacity placed in service is the first one discontinued.

CANCELLED B. Service Rearrangements as set forth in 6.7.1(C)(3) and 7.3.1(B)(3)(F()) following, may be made without a change in minimum period reputrements.

5.2.6 Minimum Period Charges

BY 3 R.S. #10

When Access Service is discontinued prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- A. For Switched Access Service, the charge for a month or fraction thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.7.4, following.
- B. For Special Access Service, the charge for a month or fraction thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.4, following.

OCT 1 1989 **89 - 14** Public Service Commission

Issued: SEP 2 5 1989;

(FC)

Effective: OCT 1 1989

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 Ist Revised Sheet 10 Replacing Original Sheet 10 常已历日的月间

(CP) ACCESS SERVICES

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE- (Continued) 5.

5.2 Access Order-(Continued)

MISSUER

5.2.4 Selection of Planned Facilities for Access Orders 200 C Service Commission

- A. When there are analog or digital high-capacity facilities to the first point of switching for switched access or to a hub for special access, the customer must specify a channel assignment for each service ordered. This information will be provided to the Telephone Company during the order process.
- B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Spectal Facilities Routing as set forth in Section 11, following. OCT 1 1989

5.2.5 Minimum Period

- BY JMBRS. #10 A. Except as set forth in Paragraph 9.4, A., following, the minimum period ission for which Access Service is provided and for which charges are applicable is one month. For the application of minimum period charges for Switched Access Service Feature Groups B, C and D, it is assumed that the last identical capacity placed in service is the first one discontinued.
- Service Rearrangements as set forth in 6.7.1(C)(2) and 7.4.1(B)(3) в. following, may be made without a change in minimum period requirements.
- 5.2.6 Minimum Period Charges

When Access Service is discontinued prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- For Switched Access Service, the charge for a month or fraction A. thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.7.4, following;
- For Special Access Service, the charge for a month or fraction в. thereof is the applicable monthly rates for the service as 1986 set forth in Section 7, Paragraph 7.5, following.

Issued: JUN 27 1986

JUL 1 1986 Effective:

86-84 Public Service Commission

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 Original Sheet 10 REGEIVED

EIISSOURI Public Service Commission

ACCESS SFRVICES

- 5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continged) 2.9 1983
 - 5.2 Access Order-(Continued)
 - 5.2.2 Access Order Modifications-(Continued)
 - D. Expedited Order Charge-(Continued)

To calculate the Expedited Order Charge, the total of all the nonrecurring charges associated with the order is divided by the number of days in the service date interval. The charge is then applied on a per day (i.e., calendar day) of improvement basis, per order. When the request for expediting occurs subsequent to the issuance of the Access Order, a service date change charge as set forth in Paragraph 5.2.2, A., preceding, also applies.

When expediting causes the Telephone Company to incur extraordinary costs that are not recovered by the Expedited Order Charge, the IC will be billed, on an individual-case basis, an amount equal to the costs incurred in lieu of an Fxpedited Order Charge.

E. Other Modification Charge

Modifications other than those described in Paragraphs 5.2.2, A., B., C. and D., preceding, may be requested by the IC. If the modification can be accommodated and implemented by the Telephone Company, an Other Modification Charge will apply on a per-order, per-occurrence basis.

Some of the types of modifications for which this charge will apply include, but are not limited to, the following:

- Change of billing address.
- Change of telephone number.
- Change in Hunt Group Arrangement.
- Change in Nonhunting Number of Hunt Group Arrangement.
- Change of Jurisdiction (i.e., breakdown between intrastate and interstate minutes of the Mori Supplied Access Services).

USOC Charge E10) 1986 JUL 1 Other Modification Charge \$29.59AN OMC per order . PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

1.

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 5th Revised Sheet 11 Replacing 4th Revised Sheet 11

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.8 Shared Use Facilities

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered. In addition, when a customer request s a Rollover, the Connecting Facility Assignments (CFA's) must be specified for the facilities involved in the Rollover for each service.

(RT)

(RT)

Issued:

September 22, 1994

Effective:

October 1, 1994



Access Services Tariff Section 5 4th Revised Sheet 11 Replacing 3rd Revised Sheet 11

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) FR 17 1994

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

RECEIVED

- 5.2.7 Minimum Period Charges-(Continued)
 - C. For CCS/SS7 Interconnection Service, the minimum period charge is set forth in Section 20, Paragraph 20.6.3, following.
- 5.2.8 Shared Use Facilities

(AT)

(AT)

Issued:

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered. In addition, when a customer requests a Rollover, the Connecting Facility Assignments (CFA's) must be specified for the facilities involved in the Rollover for each service.

CANCELLED

OCT 011994 Public Service Commission MISSOURI

MAR 201993

FEB 1 8 1994 By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

i

(RT)

(RT)

Access Services Tariff Section 5 4th Revised Sheet 11 Replacing 3rd Revised Sheet 11

ACCESS SERVICES

RECEIVED

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) DEC 23 1933

5.2 Access Order-(Continued)

MISSOURI Public Service Commission

5.2.8 Shared Use Facilities

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered.

CANCELLED

COT 1 1994 9507 193 BY TR SS-85 Public Service Commission MISSOURI

JAN 11994 93-224 MISSOURI

Public Service Commission

Issued: DEC 2 3 1993

Effective: JAN 0 1 1994

By M. H. SCHULTEIS, Division Manager-Regulatory and Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 3rd Revised Sheet 11 Replacing 2nd Revised Sheet 11

ACCESS SERVICES

5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

MAY 11 1992

RECEIVED

- 5.2 Access Order-(Continued)
- 5.2.7 Minimum Period Charges-(Continued)

MO. PUBLIC SERVICE COMP.

- (AT) (AT)
- C. For CCS/SS7 Interconnection Service, the minimum period charge is set forth in Section 20, Paragraph 20.6.3, following.
- 5.2.8 Shared Use Facilities

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered.

CANCELLED

MAR 201994 Th R.S#11 Public Service Commission MISSOURI

FILED

NOV 10 1992

MO. PUBLIC SERVICE COMM.

Issued: MAY 1 3 1992

Effective: OCT 2 1987 NOV 1 6 1992

By A. D. ROBERTSON, Assistant Vice President-External Affairs Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 2nd Revised Sheet 11 Replacing 1st Revised Sheet 11

ACCESS SERVICES

RECEIVED

(RT) 5. ORDERING FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) AUG 9 1991

5.2 Access Order-(Continued)

(FC) 5.2.8 Shared Use Facilities

MISSOURI

atic Service Commission

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered.

CANCELLED

NOV 16 1992 BY <u>3 ~ R.S.</u> # I(Public Service Commission MISSOURI

Issued: AUG 0 9 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri Public Service Commissic

Access Services Tariff Section 5 1st Revised Sheet 11 Replacing Original Sheet 11

JUN 2 7 1986

MISSUAR

ħ

(CP)ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.7 Shared Use Facilities

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowed. Shared use facilities to a Hub will be ordered and provided as Special Access Service. While shared use is allowed, individual services utilizing these facilities must be ordered either as Switched Access Service or Special Access Service. When placing the order for the individual service(s), the customer must specify a channel assignment for each service ordered.

CANCELLED

SEP 3 0 1991 BY <u>2⁻⁻⁻ R.S.⁺</u> Public Service Commission MISSOURI



Issued: JUN 27 1986

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 Original Sheet 11

BECEIVED

I. IISSOURI Public Service Commission

ACCESS SERVICES

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 2 9 1993 5.

5.2 Access Order-(Continued)

5.2.3 Cancellation of an Access Order

A. An IC may cancel an Access Order for the installation of service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the order is to be cancelled. The verbal notice must be followed by written confirmation within 10 days.

If an IC or an End User is unable to accept Access Service within 30 calendar days of the original service date, the IC has the choice of the following options:

- The Access Order shall be cancelled, and charges set forth in Paragraph 5.2.6, C., following, will apply, or
- Billing for the service will commence.

In such instances, the cancellation date or the billing date, depending on which option is selected by the IC, shall be the 31st day beyond the original service date of the Access Order.

- B. Access Order costs are considered to have started when the Telephone Company incurs any cost in connection therewith or in preparation thereof which would not otherwise have been incurred. These costs include but are not limited to preliminary engineering, orders to suppliers and other similar items of cost. For purposes of determining Cancellation Charges, the costs are considered to have started on the order date (i.e., the day the Telephone Company gives a firm order confirmation to the IC).
- C. When an IC cancels an Access Order for the installation of service, a Cancellation Charge will apply on a per-order basis. The Cancellation Charge is calculated by multiplying the total of the nonrecurring charges associated with all items on the Access Order by the number of calendar days elapsed since the order date and dividing that figure by the number-ofdays in the service interval (i.e.n raennumber of days between the order date and the service dane). Fire resubring amount is the Cancellation Charge. When determining the charge for a partial cancellation as set forth in Paragraph 5.2.2, B., preceding, only those nonrecurring^C charges associated with the portion of the order being cancelled are used 5r8 computing the charge. Public Service Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

PUBLIC SERVICE COMMISSION

MISSOURI

BY.

Access Services Tariff Section 5 1st Revised Sheet 12 Replacing Original Sheet 12 through Original Sheet 30

ACCESS SERVICES

(RT)

Issued: June 27, 1986

Effective:

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 12

Elissod M Public Service Commission

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE- (coll in ged) (223

- 5.2 Access Order-(Continued)
- 5.2.3 Cancellation of an Access Order-(Continued)
- D. When an IC cancels an order for the discontinuance of service, no charges apply for the cancellation.
- 5.2.4 Selection Of Planned Facilities For Access Orders
 - A. When there are facilities in the activated state for its use under a Planned Facilities Order or analog or digital high capacity services or a Hub in the Access Service state, an IC may request a specific channel or transmission path be used to provide the Switched or Special Access Service requested in an Access Order. The Telephone Company will make a reasonable effort to accommodate the IC's request. If the Telephone Company determines that the request cannot be met with reasonable effort, and another facility is activated for the IC's use, the Telephone Company will specify the use of that facility and notify the IC accordingly.
- B. For all other Access Orders, the option to request a specific transmission path or channel is not provided, except as provided for under Special Facilities Routing as set forth in Section 11, following.
- 5.2.5 Minimum Period
- A. Except as set forth in Paragraph 5.2.5, B. and C., following, the minimum period for which Access Service is provided and for which charges are applicable is six months. For the application of minimum period charges for Switched Access Service Feature Groups B, C and D, it is assumed that the last identical capacity placed in service is the first one discontinued.
- B. The minimum period for which Access Service ordered under a short Notice Interval is provided and for which charges are applicable is one month.
- C. The minimum period for analog and digital high capacity services (facilities) to a Hub (service date and activation date are always the same) is 24 months and applies as set forth in Paragraph 5.4.9, A., following.

months and applies as set forth in Paragraph 5.4.9, A., following. 阿正尼回 JUL 1 1986 JAN - 1 1934 83.-253 NR S. #12 PUBLIC SERVICE COMMISSION Public Fer Int Comprising OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri
No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Orfginal Sheet 13

ACCESS SERVICES

DEC 2 3 1003

Public Service Commission

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

- 5.2 Access Order-(Continued)
 - 5.2.6 Minimum Period Charges

When Access Service is discontinued prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. A disconnect constitutes facilities being returned to available inventory.

The Minimum Period Charge for monthly billed services will be determined as follows:

- A. For Switched Access Service, the charge for each remaining month and/or fraction thereof is equal to the applicable minimum monthly charge for the capacity as set forth in Section 6, Paragraph 6.7.4, following.
- B. For Special Access Service, the charge for each remaining month and/or fraction thereof is the applicable monthly rates for the service as set forth in Section 7, Paragraph 7.5, following.

GANGELLED BALE . JAN - 1 15" 1986 .101-1 Punii8r3 BY PUBLIC SERVICE COMMISSION OF MISSOURI Issued: DEC 2 9 1983 Effective: JAN 0 1 1934

ACCESS SERVICES

DEC 2 0 1203

Public Service Commission

Section 5

Access Services Tariff

Original Sheet

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.2 Access Order-(Continued)

5.2.7 Shared Use Facilities

Shared Use (i.e., Switched and Special Access Services provided over the same analog or digital high capacity facilities) is allowable only for existing services and to fill out existing systems. Only Switched Access Service busy-hour minutes of capacity may be added to fill out existing systems. When Switched or Special Access Services provided over a shared use facility are disconnected, only Switched Access Service busy-hour minutes of capacity can be ordered by the IC to backfill the system.

5.3 Available Inventory

Available inventory is limited and does not include facilities previously ordered or reserved. Available inventory also does not include analog or digital high capacity facilities. All Access Orders for analog or digital high capacity services or interface groups (i.e., Group, Supergroup, Mastergroup, DS1, DS1C, DS2, DS3 and DS4) require the use of Planned Facilities.

To the extent they can be provided, the Telephone Company will provide other than high capacity Access Services from an available inventory, except when the IC has Planned Facilities activated for its use and requests the Telephone Company to use such Planned Facilities in providing the requested services.

GANGELLED JAN - 1 1981 .111 1 1986 ŘΥ PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff NE ASection 5 Original Sheet

ACCESS SERVICES

DEC 2 0 1203

ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 3000 , Public Service Commission

5.4 Planned Facilities Order

5.4.1 General

- A. An IC may place a Planned Facilities Order for the provision of Planned Facilities. Planned Facilities provided for Special Access Service include Access Connections, Special Transport, Features and Functions and Special Access Lines as described in Section 7, Paragraph 7.1, following. Planned Facilities provided for Switched Access Service include Access Connections, Local Transport and End Office functions as described in Section 6, Paragraph 6.1, following. Planned Facilities will be made available by the Telephone Company for the IC's use on an established activation date. The combinations which may be ordered and made available for the IC's use on the activation date are set forth in Paragraph 5.4.2, following.
- Planned Facilities Orders must be placed separately for each activation в. date for Special Access Service and Switched Access Service. For Switched Access Service, a separate order must be placed for each IC terminal location to end office (by Feature Group type) ordered.

For Special Access Service, separate orders must be placed for each IC terminal location to End User's premises, IC terminal location to IC terminal location and IC terminal location to Hub serving wire center, End User's premises to End User's premises or End User's premises to WATS or WATS-type serving office.

Metallic facilities between the IC terminal location and the IC serving wire center may be ordered for use with Switched Access Service or Special Access Service. A separate order must be placed for each request for such facilities.

C. To the extent that Planned Facilities can be made available, and unless Special Construction Charges apply as set forth in Paragraph 5.4.1, D., following, the Planned Facilities ordered will be available to the IC at rates and charges as set forth in Paragraph 5.4.11, following, on the activation date. Except as set forth following, the activation date may be set for any working day at least 24 months but not more than 36 months 后近国间 BELLED after the order date.

JUL 1 1986

OF MISSOURI

Effective:

Issued: DEC 2 9 1983

JAN 0 1 1984

JAN - 1 1934

Dublin Service Commission

83-253

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

DUBLIC

Access Services Tariff

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Section 5 Original Sheet 16 REGEIVED ACCESS SERVICES ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5. LISTORY 5.4 Planned Facilities Order-(Continued) Public Service Commission 5.4.1 General-(Continued)

C. (Continued)

With the agreement of the Telephone Company, the Planned Facilities activation date may be scheduled for a date earlier than 24 months from the order date. If an IC requests an activation date more than 36 months after the order date, the Telephone Company will determine if such requests can be accommodated. When they can be, orders shall be accepted.

D. Planned Facilities for which the Telephone Company has no foreseeable reuse, may be subject to Special Construction Charges as set forth in Section 14, following, in addition to the rates and charges set forth in Paragraph 5.4.11, following.

The Telephone Company shall determine whether Special Construction Charges apply and will so notify the IC prior to establishing a firm order.

5.4.2 Planned Facilities Provided

The IC may order Planned Facilities as follows:

- For Switched Access Service Α.
 - Two-wire and four-wire metallic facilities between the IC terminal 1. location and the IC serving wire center. These must be ordered in 100 pair units.
 - Busy-hour minutes of capacity (BHMC) between an IC terminal location 2. and an end office (by Feature Group and type of BHMC). Optional features may also be specified by the IC.

GANGELLED FILED JUL 1 1986 JAN - 1 1984 83-253 PUBLIC SERVICE COMMISSION Service Commission OF MISSOUN

DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

Ρ	.s	.C.	Mo	No	•	36

DEC 28 1083

1.1220041

JAN - 1 1934 83 - 253

Public Service Communs

Public Service Commission

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.4 Planned Facilities Order-(Continued)

5.4.2 Planned Facilities Provided-(Continued)

A. For Switched Access Service-(Continued)

Only those rate elements included on the Planned Facilities Order will be made available to the IC on the activation date.

- B. For Special Access Service
 - 1. Two-wire and four-wire metallic facilities between the IC terminal location and the IC serving wire center. These must be ordered in 100 pair units.
 - 2. Analog and digital high capacity (i.e., Group, Supergroup, Mastergroup, DS1, DS1C, DS2, DS3 and DS4) facilities between an IC terminal location and a Hub serving wire center. Hub functions may also be specified by the IC.
 - 3. Analog and digital high capacity (i.e., Group, Supergroup, Mastergroup, DS1, DS1C, DS2, DS3 and DS4) facilities between two IC terminal locations. One of the IC terminal locations will be treated as an End User's premises. Facility interface combinations and optional features may also be specified by the IC.
 - 4. Narrowband, Voice Grade, Program Audio, Wideband Analog, Wideband Digital, Digital Data Access and High Capacity 1 (HCl) facilities between an IC terminal location and an End User's premises. Features and functions (e.g., facility interface combinations and optional features) may also be specified by the IC.
 - 5. Dedicated Access Line Service two-wire or four-wire facilities between an End User's premises and a WATS or WATS-like serving office. Optional features may also be specified by the IC.

Only those rate elements included on the Planned Facilities Order will be made 문비나리비 available to the IC on the activation date

JUL 1 1986

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

SERVICE COMMISSION

Access Services Tariff Section 5 Original Sheet 18

ACCESS SERVICES

I SEIVED

DEC 20 1083

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

5.4 Planned Facilities Order-(Continued)

5.4.3 Advance Payments

Planned Facilities Orders are subject to advance payments by the IC. Three advance payments are required during the Order Interval, i.e., interval between the order date and the original activation date. These are set forth in Paragraphs 5.4.3, A., B. and C., following. Each advance payment is based on the monthly charge as described in Paragraph 5.4.11, following.

- A. The first advance payment is due to the Telephone Company 30 days after the order date. The payment shall be an amount equal to three times the monthly charge.
- B. The second advance payment is due to the Telephone Company on the last day of the month that is specified in the following table:

Order Interval In Months	Month After Order Date Second Payment Is Due		
up to 19	lst		
over 19 to 20	2nd		
over 20 to 21	· 3rd		
over 21 to 22	4th		
over 22 to 23	5th		
over 23 to 24	6th		
over 24	7th		

The payment shall be an amount equal to six times the monthly charge.

GANGELLED 后江臣的 JUL 1 1986 JAN - 1 1984 BYLSA PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 19 NEOEUVEU

1.1.3360.1

Public Service Commission

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Contanued)

5.4 Planned Facilities Order-(Continued)

- 5.4.3 Advance Payments-(Continued)
- C. The third advance payment is due to the Telephone Company on the last day of the month that is specified in the following table:

Order Interval In Months	Month After Order Date Third Payment Is Due
up to 12	lst
over 12 to 13	2nd
over 13 to 14	3rd
over 14 to 15	4th
over 15 to 16	5th
over 16 to 17	6th
over 17 to 18	7th
over 18 to 19	8th
over 19 to 20	9th
over 20 to 21	lOth
over 21 to 22	11th
over 22 to 23	12th
over 23 to 24	13th
over 24	14th

The payment shall be an amount equal to three times the monthly charge.

If any advance payment is not received by the Telephone Company by the end of the first working day of the next month, the Telephone Company will cease all work on the Planned Facilities Order involved and Cancellation Charges as set forth in Paragraph 5.4.6, following, will apply.

In those cases where the activation date is less than 24 months, the second and possibly the third advance payment may be due at the same time as the first advance payment.

The advance payments will be credited against money owed by the IC for activated Planned Facilities of Access Setvice of the same type, i.e., Switched Access Feature Group accounts or Special Access accounts, on the activation date for the facility for which the advance payment was taken.

PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

J/.N ~ 1 1934 8 3 - 2 5 3

Public Service Commission:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 Original Sheet 20 REGEIVED

MASSOURE

Public Service Commission

ACCESS SERVICES

									DEC on	1000
5.	ORDERING	OPT IONS	FOR	SWITCHED	AND	SPECIAL	ACCESS	SERVICE-	(continue2)9	んしじ

5.4 Planned Facilities Order-(Continued)

5.4.3 Advance Payments-(Continued)

C. (Continued)

Should the Planned Facilities Order be cancelled by the IC, the amount of the advance payments will be credited to the IC's account, and any remaining credit balance will be refunded. The fact that advance payments have been made in no way relieves the IC from complying with the Telephone Company's regulations as to other payments or the prompt payment of bills.

5.4.4 Hub Serving Wire Centers

Planned Facilities Hub serving wire centers for Special Access may be established by the Telephone Company at designated locations in a LATA. A Hub is a Telephone Company office where facilities will be available to reach a specified set of serving wire centers and End User's premises. The location of Hubs, if any, in a LATA will be specified by the Telephone Company. Different serving wire centers may be designated as Bridging Hubs and Multiplexing Hubs. The wire centers served by the Hub serving wire center will also be specified by the Telephone Company.

5.4.5 Modifications of Planned Facilities Orders

Modifications of Planned Facilities Orders may be made by the IC without charge prior to acceptance by the Telephone Company of the Planned Facilities Order. The date when the order is accepted is known as the order date. After the order date, modifications may be made as set forth following:

A. Change in Number of Facilities

After the order date, any increase or decrease in the number or type of facilities ordered will be treated as follows: JAN = 1.4934

- 1. An increase in the number of family files ordered will be tressed as 2353 new Planned Facilities Order for the increased amount only files Commission

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 5 Original Sheet 21 DECENTED

ELISEUL II

JAN - 1 19

83 -

ACCESS SERVICES

- ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued) 5. DEC 2.0 (000
- 5.4 Planned Facilities Order-(Continued)
 - 5.4.5 Modifications of Planned Facilities Orders-(Continued) **Public Service** Commission
 - B. Activation Date Changes

With the agreement of the Telephone Company, the IC may change the activation date of a Planned Facilities Order subject to the following conditions:

- No activation date may be established that is more than 12 months beyond 1. the activation date specified on the original order. If the IC is unable to accept the facilities within 12 months of the original activation date, the IC has the choice of the following options:
 - The Planned Facilities Order shall be cancelled and charges set forth à. in 5.4.6, B., following, will apply or
 - Billing for the facilities will commence. Ъ.
- 2. If the requested date is prior to the original activation date, Additional Engineering, Additional Labor and/or special construction charges may apply in order to meet the date. The Telephone Company will determine whether such charges apply and will notify the IC accordingly.
- 3. A Planned Facilities Order Modification Charge as set forth in Paragraph, 5.4.5, D., following, will apply for each activation date change.
- С. Other Modifications
 - With the agreement of the Telephone company, or modifications be made prior to the activation date. The types of modifications $|\vec{r}| = 1$ are not limited to: 1.
 - Change in facility interface
 - Addition or deletion of optional features
 - Change in type of Transport Termination (Switched
 - Access only) Change from two-wire facilities or from four-wire facilities to two-wire facilities

1986 A Planned Facilities Order Modification Charge as set forth in Paragraph 5.4.5, D., following, will apply for these types of modification. BY

BY BY CE COMMISSION
 The following types of Public Service COMMISSION may not be made, but instead require the cancellation and reissuance of the order:

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 22

1

į

	ACCESS SERVICES		
5. OR	DERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SET	RVICF-(C	ontinued): 1000
5.4	Planned Facilities Order-(Continued)	l I Pul	1. ISSUUM blic Service Commission
5.4.	5 Modifications of Planned Facilities Orders-(Con	ntinued)	
С.	Other Modifications-(Continued)		
	2. (Continued)		
	 Change in Feature Group type Change in location, i.e., the IC terminal location, the end office or the End User's Change in type of facilities other than from to four-wire facilities or four-wire facilities 	premise om two-w	s ire facilities
	Cancellation Charges as set forth in Paragra will apply in these cases.	ph 5.4.6	, following,
D.	Planned Facilities Order Modification Charge		
	The Planned Facilities Order Modification Charge per-occurrence basis. The applicable charge is:	is appl	ied on a per-order,
		USOC	Charge
	Planned Facilities Modification Charge, per order	OMC	\$29.59
	If more than one change is being made at the sam applies.	e tíme,	only one charge
5.4.	.6 Cancellation of a Planned Facilities Order		
А.	A Planned Facilities Order may be cancelled afte Cancellation Charges will apply as set forth in and C following. The cancellation date is the Company receives written or verbal notice from t or channels ordered are to be cancelled. The ve followed by written confirmation within 10 days.	Paragrap date th he IC#th	hs 5,4.6 [[B5 []] e Telephone at the capacity
B.	The applicable Cancellarion What a fisl based on to cancellation date and the activation date or the cancellation date and 24 months after, the order is shorter. This interval is known as the Cancel Cancellation Charge applicable to each Cancellat the following table:	interva date, wh llation	ichever interval Interval. The
Issued	d: DEC 2 9 1983 Effective:	IAN 0 1 1	1984

· -

•

No supplement to this tariff will be issued evcept for the purpose of canceling this tariff.	Access Services Tariff Section 5 Original-Sheet 23
ACCESS SERVI	
5. ORDERING OPTIONS FOR SWITCHED AND SPECIA	L ACCESS SERVICE-(Continued) 1883
5.4 Planned Facilities Order-(Continued)	FUISSOUN Public Service Commission
5.4.6 Cancellation of a Planned Facilitie	s Order-(Continued)
B. (Continued)	
Cancellation Interval In Months Prior To Activation Date	Cancellation Charge Per Capacity Or Channel Cancelled
Less than 1 1 to 2 Over 2 to 3 Over 3 to 4 Over 4 to 5 Over 5 to 6 Over 6 to 7 Over 7 to 8 Over 8 to 9 Over 9 to 10 Over 10 to 11 Over 11 to 12 Over 12 to 13 Over 13 to 14 Over 14 to 15 Over 15 to 16 Over 18 to 19 Over 19 to 20 Over 20 to 21 Over 21 to 22 Over 22 to 23 Over 23	24 times Monthly Charge(1) 23 times Monthly Charge(1) 22 times Monthly Charge(1) 21 times Monthly Charge(1) 20 times Monthly Charge(1) 19 times Monthly Charge(1) 18 times Monthly Charge(1) 17 times Monthly Charge(1) 16 times Monthly Charge(1) 15 times Monthly Charge(1) 13 times Monthly Charge(1) 12 times Monthly Charge(1) 11 times Monthly Charge(1) 10 times Monthly Charge(1) 9 times Monthly Charge(1) 6 times Monthly Charge(1) 7 times Monthly Charge(1) 6 times Monthly Charge(1) 5 times Monthly Charge(1) 6 times Monthly Charge(1) 5 times Monthly Charge(1) 4 times Monthly Charge(1) 5 times Monthly Charge(1) 13 times Monthly Charge(1) 14 times Monthly Charge(1) 15 times Monthly Charge(1) 15 times Monthly Charge(1) 16 times Monthly Charge(1) 17 times Monthly Charge(1) 17 times Monthly Charge(1) 10 times Monthly Charge(1) 10 times Monthly Charge(1) 11 times Monthly Charge(1) 12 times Monthly Charge(1) 12 times Monthly Charge(1) 13 times Monthly Charge(1) 11 times Monthly Charge(1)
 C. Where special construction is involved. Termination Liability charges as set will apply in addition to charges set preceding. (1) Monthly Charge is the charge as the charge of Monthly Charge is the charge as the charge of Monthly Charge is the charge as the charge of Monthly Charge is the charge as the charge of Monthly Charge is the charge as the charge of Monthly Charge is the charge of Monthly Charge of Monthly Charge is the charge of Monthly Charge of Monthly Charge is the charge of Monthly Charge of	forth in Section 14, following,
Issued: DEC 2 9 1983 Effect	tive: JAN 0 1 1904
By R. D. BARRON, Vice H Southwestern Bell Te St. Louis, N	elephone Company

____

ACCESS SERVICES

DEC 2.9 (003)

Original Sheet 24

Access Services Tariff

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Continued)

Public Service Commission

5.4 Planned Facilities Order-(Continued)

5.4.7 Activation of Planned Facilities

The Planned Facilities ordered are made available for the IC's future service needs on the activation date. Special or Switched Access Service using the Planned Facilities will be provided pursuant to standard, negotiated or short-notice service date intervals, under an Access Order as set forth in Paragraph 5.2.1, preceding. The service date on such orders may be the same as, or later than, the Planned Facilities Order activation date. For analog and digital high capacity planned facilities order to a Hub, the service date and activation date are always the same. When the service and activation dates are the same, the Access Order charges apply and the Planned Facilities charges do not. If the service is disconnected prior to the expiration of the minimumperiod requirements for the Planned Facilities Order, the facilities will be returned to the activated status and Planned Facilities Order minimum-period charges will commence and will apply until such time as the minimum-period requirements are fulfilled.

5.4.8 Delayed Activation Credit

The Telephone Company will make every effort to provide the Planned Facilities to the IC on the established activation date. If the date is missed, unless for reasons beyond the control of the Telephone Company as set forth in Section 2, Paragraph 2.1.3, preceding, and Paragraphs 5.4.12, following, a Delayed Activation Credit will apply and will be credited to the IC's account.

The Delayed Activation Credit will be 1/30 of the monthly charge as specified in Paragraph 5.4.11, following, for each day the date is missed.

5.4.9 Minimum Period

A. The minimum period for which Planned Facilities are activated for the IC's use and for which charges are applicable is 24 months. Beginning on the activation date, the appropriate charges as specified in Paragraph 5.4.11, following, with applitude ELLE

JUL 1

et R.S

1986

BLIC SERVICE COMMISSION

DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

·J7.\! = 1 1934 83 - 253

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 5 Original Sheet 25 RECEIVED

> LIISSOURI Public Service Commission

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Chi fined)

5.4 Planned Facilities Order-(Continued)

5.4.9 Minimum Period-(Continued)

- B. Facilities in the activated status for which the minimum period has ended will be treated as follows:
 - 1. Subject to Paragraph 5.4.9, C., following, the IC may retain the facilities in the activated status, and the charges as set forth in Paragraph 5.4.11, following, will apply.
 - 2. Subject to the provisions of Paragraph 5.5, following, the IC may return the facilities to the Telephone Company upon 30-days' notice. All availability for use by the IC will cease as of that date.
- C. Facilities in the activated status for which the minimum period has ended, and for which there are no currently pending Access Orders, except those where special construction applies, may be taken by the Telephone Company for its use upon written 90-days' notice to the ordering IC. If the IC does not submit Access Orders for the use of such Planned Facilities prior to the end of the 90-day notice period, all availability for use by the IC will cease as of that date or other date mutually agreeable to the Telephone Company and the IC, and charges as set forth in Paragraph 5.4.11, following, will cease.
- D. Planned Facilities provided by special construction will be subject to liabilities and charges as set forth in Section 14, following, in addition to Paragraphs 5.4.9, A, through C., preceding.

		·	BANBELLED JULI 1986 BY 24 25, # 12 PUBLIC SERVICE COMMISSION OF MISSOUR	[-]][E-]] JAN - 1 1931 Public Service Com-
	Issued:	DEC 2 9 1983	Effective: JAN 0 1	1984
		Ву	R. D. BARRON, Vice President-Missou Southwestern Bell Telephone Company St. Louis, Missouri	

Access Services Tariff Section 5 Original, Sheet 26 医试验检疗

> <u>. 1 193 - 1 193)</u> 83-258

ACCESS SERVICES						
	DEC 2,5 1223					
5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS	SERVICE-(Continued)					
5.4 Planned Facilities Order-(Continued)	EdiSoutial Public Service Commission :					
5.4.10 Minimum Period Charges						

When Planned Facilities are returned to the Telephone Company prior to the expiration of the minimum period, charges are applicable for the remainder of the minimum period. The charge for the quantity returned will be an amount equal to the applicable monthly charge times the number of months and/or fraction thereof between the return date and the end of the minimum period. The monthly charge is set forth in Paragraph 5.4.11, following.

5.4.11 Monthly Charge

Monthly charges are applicable for activated Planned Facilities. The monthly charge will apply from the activation date until such time as the facilities are used to provide Access Service via an Access Order or until the minimum period requirements have been satisfied. If the minimum period requirement has been satisfied and the IC chooses to retain the facilities in the activated status, the monthly charge will continue to apply. If the IC requests the Planned Facilities be used to provide Access Service and they are so used, the Planned Facilities monthly charge will cease, and the appropriate Switched or Special Access Service rates and charges set forth in Section 6, Paragraph 6.8 and Section 7, Paragraph 7.5, respectively, will apply.

The monthly charge will serve as the basis for advance payments and Cancellation Charges, as well as for the monthly charge to be billed for activated Planned Facilities.

The monthly charges are as follows:

A. For Switched Access

Activated Switched Access Planned Facilities between the IC terminal location and the end office will be billed at the rates established for Minimum Monthly Usage charge (1900) [Sellset forth in Section 6, Paragraph 6.7.5, following A. R. J. S. L. [2011] FILLER

JUL 1 1986 S.#12 SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the of

5.

5

riff will be issued cept for the purpose canceling this tariff.	Section 5 []Original, Sheet 27 []]」」」」」」」」」)」
ACCESS SERVICES	DEC 2 9 1000
ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS	SERVICE-(Continued)
ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS .4 Planned Facilities Order-(Continued)	LUSSUURI Public Service Commission
5.4.11 Monthly Charge-(Continued)	

Access Services Tariff

B. For Special Access

Activated Special Access Planned Facilities will be billed at the monthly rates in Section 7, Paragraphs 7.5.1, 7.5.2 and 7.5.4, following, for the Access Connections, Special Transport and Special Access Lines, respectively.

Bridging functions, if specified on the Planned Facilities Order, will be billed at the monthly rates set forth in Section 7, Paragraph 7.5.3, C., following. Billing for all other Features and Functions, including Facility Interface Combinations, even if specified on the order, will not commence until placed in service via an Access Order. The monthly rate is the total of all the items included on the order for which charges are to apply while in the activated status.

C. For Metallic Facilities

Two-wire and four-wire metallic facilities between an IC terminal location and the IC serving wire center, whether ordered as Switched Access or Special Access, will be billed at the monthly rates set forth in Section 7, Paragraph 7.5.1, following, for two-wire and four-wire Special Access Service Access Connections.

5.4.12 Planned Facilities Test and Line Up

Prior to the activation date, the Telephone Company will test and line up A. the facilities provided, as appropriate. Test and line up will be made by the Telephone Company using the IC facilities which are installed at the IC terminal location to terminate the capacity or services provided. The IC shall provide personnel and equipment and shall have its facilities available to complete the testing and line up during the period specified. If the personnel, equipment and facilities are not provided by the IC, and the activation date is missed, no Delayed Activation Credit applies.

	GANGELLED FALED
	JUL 1 1986 JAN - 1 1981 JUL 1 1986 83 - 253 JUL
	BY AR.S. # 12-
Issued:	DEC 2 9 1983 Effective: JAN 0 1 1984
	By R. D. BARRON, Vice President-Missouri

Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 5 Original Sheet 28

USSOURI

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Condigued)03

5.4 Planned Facilities Order-(Continued)

5.4.12 Planned Facilities Test and Line Up-(Continued) Public Service Commission

B. Testing and line up as set forth in Paragraph 5.4.12, A., preceding, may begin up to 30 days prior to the scheduled activation date. The Telephone Company will notify the IC of the scheduled start date at least ten working days prior to such date. If this date is mutually agreeable, the Telephone Company will begin testing on the scheduled date. If this date is not agreeable to the IC and another date is agreed upon, no Delayed Activation Credit will apply if the facilities are not available on the activation date.

5.5 Switched Access Service Minimum Capacity Requirements

- 5.5.1 When Switched Access Service Access Connections are ordered under Planned Facilities Orders or Access Orders, they will be provided subject to the minimum capacity provisions set forth in Paragraph 5.1.1, preceding, and Paragraphs 5.5.2 through 5.5.7, following.
- 5.5.2 There is no minimum capacity for Interface Groups 1 and 2, except as set forth in Paragraphs 5.4.2, A. 1., preceding. The minimum capacity provided for Interface Groups 3 through 10 and for which charges are applicable is set forth in Paragraph 5.5.6, following.
- 5.5.3 When additional busy-hour minutes of capacity are ordered, they may be provided using activated or in-service Access Connection Interface Groups. Such busy-hour minutes of capacity, up to and including the number that brings the Interface Group to its designated capacity, will be provided without being subject to the minimum capacity provisions as set forth in Paragraph 5.5.6, following. Any additional busy-hour minutes of capacity ordered will be subject to the minimum capacity provisions, unless they are provided under an Access Order using Planned Facilities.

	·
GANGELLED	Fail (210)
	JAN - 1 1931
JUL 1 1986	JAN - 1 1931 88 - 258 Public Service Co
BY LAL R.S. H. 12- PUBLIC SERVICE COMMISSION	Public Service Community
QF MISSOUR	

Issued: DEC 2 9 1983

Effective: JAN 0 j 1984

Access Services Tariff Section 5 Original Sheet 29 UEU 20 ICOD

LUSSUERI

- 11- Sar

ACCESS SERVICES

- (ContinueaSommission 5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE
 - 5.5 Switched Access Service Minimum Capacity Requirements-(Continued)
 - 5.5.4 When an IC requests high frequency analog or Digital interface groups, the IC is required to order at a minimum, sufficient capacity to utilize 70 percent of the channels.
 - 5.5.5 For the purpose of administering the minimum capacity provisions, Access Orders for Access Connection Interface Groups for different Feature Groups may be grouped together if the facilities provided for all the connections are the same and terminate in the same facilities terminal in the same end office.
 - 5.5.6 The following table provides the total capacity of the interface and the threshold for minimum order requirements:

Interface Type	Interface Name	Total Capacity (Channels)	Minimum Capacity (Channels)
Analog	Group	12	9.
Analog	Supergroup	60	42
Analog	Mastergroup	600	420
Digital	DS1	24	17
Digital	DS1C	48	34
Digital	DS 2	96	68
Digital `	DS 3	672	471
Digital	DS4	4032	2823

The Telephone Company will not provide these Interface Groups when less than 70 percent of the capacity is ordered. For purposes of grouping as set forth in Paragraph 5.5.5, preceding, it shall be assumed that Feature Groups A, B, C or D minutes may be combined.

- 5.5.7 When Switched Access Service provided from available inventory is disconnected and the disconnect causes the in-service capacity to fall below the minimum requirements, the Telephone Company will at the option of the IC:
- Disconnect all the service GANCE To the minimum capacity requirements. A. . and all appropriate charges will apply, or JUL

PUBLIC SERVICE COMMISSION

OF MISSOURI

DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

Access Services Tariff Section 5 Original Sheet 30

ACCESS SERVICES

5. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE-(Contended)3

5.5 Switched Access Service Minimum Capacity Requirements-(Continued)?

5.5.7 (Continued)

B. Establish a Planned Facilities Order for the minimum capacity as of the disconnect date, and all facilities not in service required to meet the minimum requirements will be established as activated Planned Facilities, with appropriate charges applying.

When a return of Planned Facilities to the Telephone Company causes the activated capacity to fall below the minimum, the Telephone Company will take back all Planned Facilities involved and all appropriate charges will apply.

5.6 Record Modification and Charges

An IC may request a record modification for existing Access Service or activated Planned Facilities. Record modifications include the following types of changes:

- Billing name
- Billing address
- Contact telephone number
- Rearrange account structure
- Change designation of IC serving wire center (applies only to the application of the five-mile rule as set forth in Paragraph 6.7.13, B., following).

The charges to be applied for record modifications are as follows:

- For changes to existing Access Service, a charge equal to the amount of the Other Modification Charge set forth in Paragraph 5.2.2, E., preceding, will apply.
- For changes to activated Planned Facilities, a charge equal to the amount of the Planned Facilities Order Modification Charge set forth in Paragraph 5.4.5, D., preceding, will apply.

GANGELLED JUL 1 1986 SERVICE COMMISSION OF MISSOUR Issued: DEC 2 9 1983 Effective: JAN 0 1 1984